## Extremal and random hyperbolic geometry

Bram Petri

### À soutenir le 21 février 2024 devant le jury composé de :

Nalini Anantharaman	Collège de France	Examinatrice
Grégory MIERMONT	École Normale Supérieure de Lyon	Rapporteur
Frédéric NAUD	Sorbonne Université	Examinateur
Alan Reid	Rice University	Rapporteur
Peter Sarnak	Princeton University, et	Rapporteur
	Institute of Advanced Study	

#### **Rapporteurs** :

Grégory Miermont	École Normale Supérieure de Lyon
Alan Reid	Rice University
Peter SARNAK	Princeton University, et
	Institute of Advanced Study



## Acknowledgements

First of all, I am very grateful to my PhD advisor Hugo Parlier and my post-doc mentor Ursula Hamenstädt. Both of them have had an immense impact on my mathematical life, through the mathematics I've learned from them and also because they have supported all other aspects of my mathematical career.

I also consider it a great honor that Grégory Miermont, Alan Reid and Peter Sarnak agreed to write a report on this thesis and that Nalini Anantharaman and Frédéric Naud are participating in the jury. I warmly thank all of them for the time and effort they put into this.

My co-authors, all of whom I have learned a lot from, deserve thanks as well. The work presented in this text would obviously never have been possible without them. I also thank Mingkun Liu and Anna Roig-Sanchis for trusting me as their (co-)advisor and for the many mathematical discussions that we've had and that I hope we will have in the future.

I consider myself very lucky for being part of the Institut de Mathématiques de Jussieu – Paris Rive Gauche. Not only is this a very stimulating mathematical environment, it's also a lot of fun. I've made many friends, both in the Analyse Complexe et Géométrie group, that I have the honor to be part of, and the other groups at the department. So thanks go to all these people.

Finally, thanks to all my friends and family for their support and of course to Federica and Maarten, for many reasons.

# Contents

1	Intr	oducti	on	9
	1.1	Hyper	bolic manifolds	10
<b>2</b>	Flat	tori a	and graphs	11
	2.1	Packin	ngs, coverings and flat tori	11
		2.1.1	Three invariants	12
		2.1.2	Lattices and flat tori	12
		2.1.3	A bit of history	15
		2.1.4	Linear programming	16
	2.2	Regula	ar graphs	17
		2.2.1	The girth	18
		2.2.2	The kissing number	19
		2.2.3	The spectral gap	20
		2.2.4	Random graphs	22
3	Ext	remal	problems I	25
	3.1	The sy	vstole	25
		3.1.1	The maximal systole	27
		3.1.2	Arithmetic surfaces with large systoles	31
		3.1.3	A combinatorial construction	33
		3.1.4	Long systolic pants decompositions	36

	3.2	The ki	issing number	38
	3.3	The fit	rst eigenvalue	40
	3.4	The m	ultiplicity of the first eigenvalue	42
	3.5	The n	umber of small eigenvalues	46
4	Cou	Inting		49
	4.1	Subgro	oup growth	49
		4.1.1	Invariant random subgroups	50
		4.1.2	Estimating the subgroup growth of a group	51
		4.1.3	Right-angled Artin groups	53
		4.1.4	Right-angled Coxeter groups	55
	4.2	Covers	s by distinct manifolds	60
	4.3	Count	ing curves	63
<b>5</b>	Rar	ndom n	nanifolds	65
	5.1	Rando	m surfaces	65
		5.1.1	Random Belyĭ surfaces	66
		5.1.2	Random covers of finite degree	73
		5.1.3	The Weil–Petersson measure	74
	5.2	Rando	om 3-manifolds	81
		5.2.1	Random Heegaard splittings	81
		5.2.2	Random covers of torus knot complements	86
		5.2.3	Random 3-manifolds with boundary	90
6	$\mathbf{Ext}$	remal	problems II	97
	6.1	The B	ers constant and related problems	98
	6.2	The di	iameter	99
		6.2.1	A model for random surfaces	100
		6.2.2	Ingredients of the proof	101

7	Fut	ure dir	rections	111
		6.3.3	Further remarks and questions	109
		6.3.2	Method of proof	107
		6.3.1	Results	106
	6.3	The C	Cheeger constant	104
		6.2.3	Further remarks and questions	103

## 1 Introduction

In this thesis, we will mainly discuss two types of related questions. The first of these concerns extremal problems in hyperbolic geometry. An example of such a question is: *what is the closed hyperbolic surface of given genus of minimal diameter?* On the one hand, these questions are often the simplest questions that we can ask about the geometric invariants of interest, but on the other hand there are only very few cases in which we are able to answer them.

The second type of questions is probabilistic in nature and often concerns growth and asymptotics in hyperbolic geometry. For example, we will study the geometry of random hyperbolic manifolds. An example of such a question is: *how many short closed geodesics does a random hyperbolic surface of large area have?* Besides making the notion of a "typical" manifold rigorous, random manifolds can also be useful for extremal problems: sometimes it's easier to prove that a random manifold has extremal properties with high probability than providing an explicit construction. This idea is called the probabilistic method and we will see several examples of it.

We have tried to write this text as one coherent story. We will draw connections to work that has been done by others, in hyperbolic geometry, but also in Euclidean geometry and graph theory. Of course, the space is too limited to give proper overviews of all related work, so a selection has been made. We also mention open questions and conjectures throughout the text. Some of these are questions that naturally came up during some of the work we will discuss, others are well known conjectures in the field. In the last chapter, we discuss directions for future research.

Before all this, we will use the rest of this introduction to recall, very briefly, some definitions from hyperbolic geometry.

### 1.1 Hyperbolic manifolds

To ensure that the readers and the author are thinking of the same thing, we start with a definition:

**Definition 1.1.** A hyperbolic manifold is a complete Riemannian manifold with constant sectional curvature equal to -1.

We can replace the condition on the curvature by the condition that the manifold is locally isometric to the hyperbolic space  $\mathbb{H}^d$  of the right dimension d. Every such manifold is isometric to a quotient  $\Gamma \setminus \mathbb{H}^d$ , where  $\Gamma < \text{Isom}(\mathbb{H}^d)$  is a discrete, torsion-free subgroup of the isometry group  $\text{Isom}(\mathbb{H}^d)$  of  $\mathbb{H}^d$ . This isometry group can be identified with  $O(d, 1)^\circ$ and the subgroup of isometries that preserve the orientation  $\text{Isom}^+(\mathbb{H}^d) \simeq SO(d, 1)^\circ$ . In dimension 2 and 3, we have two accidental isomorphisms:

 $\operatorname{Isom}^+(\mathbb{H}^2) \simeq \operatorname{PSL}(2,\mathbb{R}) \quad \text{and} \quad \operatorname{Isom}^+(\mathbb{H}^3) \simeq \operatorname{PSL}(2,\mathbb{C}).$ 

In this thesis we will mainly consider orientable hyperbolic manifolds of finite volume. Most of the time, our hyperbolic manifolds will be 2- or 3-dimensional, but hyperbolic manifolds exist in all dimensions  $d \ge 2$ . In dimension 2, we can even deform a hyperbolic metric on a given surface and we will write  $\mathcal{M}_{g,n}$  for the moduli space of hyperbolic surfaces of genus g and with n cusps and  $\mathcal{M}_g = \mathcal{M}_{g,0}$ . The Mostow–Prasad rigidity theorem [Mos68, Pra73] implies that in dimension greater than two there are no non-trivial deformations. In dimension 3, we still know many constructions of hyperbolic manifolds of finite volume. We can even say, without causing too much controversy, that a "typical" closed 3-dimensional manifold admits a hyperbolic metric. In dimensions above 3, hyperbolic manifolds of finite volume are much rarer. One way to construct them is using arithmetic subgroups of Isom<sup>+</sup>( $\mathbb{H}^d$ ). The question whether there are also non-arithmetic manifolds was open for a long time and was resolved by Gromov–Piatetski-Shapiro [GPS88]. However, we still know very few constructions of such manifolds.

For more details on hyperbolic geometry, we refer the reader for instance to: [Bus10, BP92, MR03, Mar22].

## 2 Flat tori and graphs

In this chapter, we discuss the problems that we will study in this thesis, but in two different contexts, namely flat tori and regular graphs. We do this because there are many analogies with hyperbolic geometry and a lot of ideas that we will use. In the section on graphs (Section 2.2) we will also briefly discuss our work with Maxime Fortier Bourque on kissing numbers of regular graphs [FBP22].

### 2.1 Packings, coverings and flat tori

Several problems that this thesis will discuss are hyperbolic analogues of very classical (and in general very open) problems in Euclidean geometry. For some motivation and also because we will be using ideas from this area in hyperbolic geometry later on, we will discuss Euclidean packings and coverings in this section. The selection we make is surely very biased and moreover the space is too small to give a serious overview of the theory. We refer to the specialized literature for more details. The book [CS93] is a classic reference, even if there has been a lot of progress since its appearance, the survey articles [PZ04, Oes19] provide overviews of more recent progress.

For us, a **packing in**  $\mathbb{R}^n$  is a collection  $\mathcal{A} = \{B(a_i, r)\}_{i \in \mathcal{I}}$  of disjoint open balls  $B(a_i, r) \subset \mathbb{R}^n$  with center  $a_i$  and fixed radius r > 0. A **covering of**  $\mathbb{R}^n$  is a collection  $\mathcal{C} = \{B_c(c_i, r)\}_{i \in \mathcal{I}}$  of closed balls  $B_f(c_i, r) \subset \mathbb{R}^n$  with center  $c_i$  and radius r > 0 such that

$$\bigcup_{i\in\mathcal{I}}B_c(c_i,r)=\mathbb{R}^n.$$

All the questions we will be asking are invariant under homothety, so as soon as the radius of all the balls is the same, we are free to choose it as we like. The most interesting case for us is the case where the centers form a lattice in  $\mathbb{R}^n$ .

#### 2.1.1 Three invariants

We are going to define three invariants: the **density** and the **kissing number** of a packing and the **thickness**<sup>1</sup> of a covering. We will start with the density: the density of a packing  $\mathcal{A} = \{B(a_i, r)\}_{i \in \mathcal{I}}$  is the proportion of the volume that is taken up by the balls in  $\mathcal{A}$ . Formally:

$$\delta(\mathcal{A}) := \limsup_{R \to \infty} \frac{\operatorname{vol}\left(\left(\bigcup_{i \in \mathcal{I}} B(a_i, r)\right) \cap B(0, R)\right)}{\operatorname{vol}(B(0, R))}$$

The kissing number is the maximal number of balls that is tangent to a given one in  $\mathcal{A}$ :

$$\operatorname{Kiss}(\mathcal{A}) := \max_{i \in \mathcal{I}} \left| \{ j \in \mathcal{I} \setminus \{i\}; B_c(a_i, r) \cap B_c(a_j, r) \neq \emptyset \} \right|.$$

The thickness of a covering  $\mathcal{C}\{B_c(c_i, r)\}_{i \in \mathcal{I}}$  is the average (with respect to  $x \in \mathbb{R}^n$ ) of the number of balls containing x:

$$\theta(\mathcal{C}) := \liminf_{R \to \infty} \frac{\sum_{i \in \mathcal{I}: B_c(c_i, r) \subset B(0, R)} \operatorname{vol}(B_c(c_i, r))}{\operatorname{vol}(B(0, R))}.$$

There is a vast body of literature on these three invariants. The central questions are:

#### Question 1

Given  $n \in \mathbb{N}$ , what are:

$$\sup_{\mathcal{A}} \delta(A), \quad \max_{\mathcal{A}} \operatorname{Kiss}(\mathcal{A}) \quad \text{and} \quad \inf_{\mathcal{C}} \theta(\mathcal{C})$$

where  $\mathcal{A}$  varies among packings in  $\mathbb{R}^n$  and  $\mathcal{C}$  varies among coverings of  $\mathbb{R}^n$ ?

#### 2.1.2 Lattices and flat tori

We have already alluded to the fact that there are a packing and a covering associated to each lattice  $\Lambda < \mathbb{R}^n$ . For the packing, we use the **injectivity radius**:

$$r(\Lambda) = \max\{s > 0; B(v, s) \cap B(w, s) = \emptyset \text{ for all } v \neq w \in \Lambda\}$$

<sup>&</sup>lt;sup>1</sup>This is sometimes also called the density of a covering.

and for a covering we use the **covering radius** of  $\Lambda$ :

$$R(\Lambda) := \min\left\{s > 0; \bigcup_{v \in \Lambda} B_f(v, s) = \mathbb{R}^n\right\}.$$

Given a lattice  $\Lambda < \mathbb{R}^n$ , the quotient  $\mathbb{R}^n/\Lambda$  with the metric induced by the Euclidean metric of  $\mathbb{R}^n$  is a flat torus. Moreover, the Killing–Hopf theorem implies that every flat torus is of this form.

In the case of lattices, the three invariants defined above can also be understood in terms of the geometry of the associated flat torus. Indeed, we have

$$\delta(\Lambda) = \frac{\omega_n}{2^n} \frac{\operatorname{sys}(\mathbb{R}^n / \Lambda)^n}{\operatorname{vol}(\mathbb{R}^n / \Lambda)}$$

where  $\omega_n = \frac{\pi^{n/2}}{\Gamma(n/2+1)}$  is the volume of a ball of radius 1 in  $\mathbb{R}^n$  and  $\operatorname{sys}(\mathbb{R}^n/\Lambda)$  is the **systole** of  $\mathbb{R}^n/\Lambda$ : the minimum length of a closed geodesic in  $\mathbb{R}^n/\Lambda$ . Equivalently,  $\operatorname{sys}(\mathbb{R}^n/\Lambda)$  is the minimal Euclidean norm of a non-zero vector in  $\Lambda$ . The (scale invariant) quantity  $\frac{\operatorname{sys}(\mathbb{R}^n/\Lambda)^n}{\operatorname{vol}(\mathbb{R}^n/\Lambda)}$  is called the **systolic ratio** of  $\mathbb{R}^n/\Lambda$ .

We also observe that  $\operatorname{Kiss}(\Lambda)$  is the number of non-zero vectors in  $\Lambda$  of minimal norm, or equivalently: the number of free homotopy classes of oriented geodesics that realize systole of  $\mathbb{R}^n/\Lambda$ .

The systole and kissing number also admit a spectral interpretation. To this end, we define the Laplacian  $\Delta : C^{\infty}(\mathbb{R}^n) \to C^{\infty}(\mathbb{R}^n)$  by

$$\Big(\Delta f\Big)(x) \coloneqq -\Big(\operatorname{div} \circ \operatorname{grad} f\Big)(x) = -\Big(\sum_{i=1}^n \frac{\partial^2}{\partial x_i^2} f\Big)(x), \quad f \in C^\infty(\mathbb{R}^n), \ x \in \mathbb{R}^n$$

The Laplacian commutes with Euclidean isometries of  $\mathbb{R}^n$  and thus descends to a differential operator  $\Delta : C^{\infty}(\mathbb{R}^n/\Lambda) \to C^{\infty}(\mathbb{R}^n/\Lambda)$ . The eigenfunctions  $(L^2$ -normalized with respect to the Lebesgue measure) for this operator are the functions  $\varphi_w : \mathbb{R}^n/\Lambda \to \mathbb{R}$  defined by

$$\varphi_w(x) = \frac{1}{\sqrt{\operatorname{vol}(\mathbb{R}^n/\Lambda)}} e^{2\pi i \cdot \langle x, w \rangle}, \quad x \in \Lambda, w \in \Lambda^\star,$$

where  $\langle \cdot, \cdot \rangle : \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R}$  is the standard inner product and  $\Lambda^*$  is the **dual lattice** of  $\Lambda$ , defined by

$$\Lambda^{\star} := \{ w \in \mathbb{R}^n; \langle w, v \rangle \in \mathbb{Z}, \ \forall v \in \Lambda \}.$$

The eigenfunctions  $(\varphi_w)_{w \in \Lambda^*}$  form an orthonormal basis of  $L^2(\mathbb{R}^n/\Lambda)$  and the eigenvalue associated to  $w \in \Lambda^*$  is

$$4\pi ||w||^2$$
.

So, the smallest non-zero eigenvalue of the Laplacian on  $\mathbb{R}^n/\Lambda$  is  $4\pi \operatorname{sys}(\mathbb{R}^n/\Lambda^*)^2$  and  $\operatorname{Kiss}(\Lambda^*)$  is its multiplicity: the dimension of the associated eigenspace.

Finally, the thickness of  $\Lambda$  can be written as

$$\theta(\Lambda) = \omega_n \; \frac{R(\Lambda)^n}{\operatorname{vol}(\mathbb{R}^n/\Lambda)}$$

Moreover, the covering radius  $R(\Lambda)$  coincides with the diameter

diam
$$(\mathbb{R}^n/\Lambda) := \max \{ d(x, y); x, y \in \mathbb{R}^n/\Lambda \},\$$

where d(x, y) denotes the distance between x and y. Indeed,  $\mathbb{R}^n / \Lambda$  is a homogeneous space, so its diameter is realized at every point. That is:

diam
$$(\mathbb{R}^n/\Lambda)$$
 = max {d([0], y);  $y \in \mathbb{R}^n/\Lambda$ } =  $R(\Lambda)$ .

The conclusion of all of this is that, if we restrict to lattices, Question 1 is equivalent to:

Question 2
Let $n \in \mathbb{N}$ , what are:
$\max_{\Lambda} \operatorname{sys}(\mathbb{R}^n / \Lambda),  \max_{\Lambda} \operatorname{Kiss}(\Lambda)  \text{and}  \min_{\Lambda} \operatorname{diam}(\mathbb{R}^n / \Lambda),$
where $\Lambda$ varies among lattices of covolume 1 in $\mathbb{R}^n$ ?

The moduli space of lattices of covolume 1 in  $\mathbb{R}^n$  can be identified with  $SL(n, \mathbb{Z}) \setminus SL(n, \mathbb{R})$ . Moreover, the fact that we can replace the supremum and the infimum of Question 1 with a maximum and a minimum is a consequence of Mahler's compactness theorem [Mah46]. Using the fact that duality induces an involution on  $SL(n, \mathbb{Z}) \setminus SL(n, \mathbb{R})$ , the maximum systole and the maximum kissing number can be replaced by the maximum of the first eigenvalue and the maximum multiplicity respectively.

#### 2.1.3 A bit of history

As mentioned before, the three questions discussed above have a long history, even if there are still very few dimensions in which their answers are known.

Among all packings, the maximal density is known in dimension 1–3, in dimension 8 and in dimension 24. In dimension 1, the question is trivial and from the classical results of Thue [Thu10] and Fejes Tóth [Fej43] we know that the densest packing in dimension 2 is the hexagonal lattice. The results in dimensions 3, 8 and 24 are consequences of the spectacular progress of the last 30 years. In 1998 Hales–Ferguson [Hal05] demonstrated that the densest 3-dimensional packing is the face-centered cubic lattice, as claimed by Kepler in 1610. Viazovska [Via17] proved that the densest 8-dimensional packing is the  $E_8$  lattice and in dimension 24 it is the Leech lattice, as proved by Cohn–Kumar–Miller– Radchenko–Viazovska [CKM<sup>+</sup>17]. The last two results use the linear programming method of Cohn–Elkies [CE03], which we will briefly describe in the next section. The idea behind this bound has its analogue in hyperbolic geometry, which we will use in Chapter 3. Among lattices, the maximum density is known in dimension 1–8 [CS93, Table 1.1, p.12] and in dimension 24 [CK09]. The best bound when the dimension  $n \to \infty$  is that of Kabatjanskiĭ– Levenšteĭn [KL78] and can be reproduced in using linear programming [CZ14].

Another point of view on density is that of systolic geometry (see [Kat07] for an introduction). There, we try to maximize the systolic ratio among all the Riemannian metrics on our torus. Apart from the trivial case of dimension 1, the answer to this question is only known in dimension 2, due to Loewner who never published his result: the hexagonal torus maximizes the systolic ratio among all Riemannian tori two -dimensional. The hexagonal torus also maximizes the first non-zero eigenvalue of the Laplacian among all Riemannian metrics of the same area [Nad96]. The spectral version of the problem has no meaning in higher dimensions: on any closed smooth manifold of dimension at least 3 there exist Riemannian metrics whose volume is 1, but whose first eigenvalue is arbitrarily large [CD94].

The kissing number has been studied a lot as well (see for example [CS93, PZ04]). The global maximizers are known in dimension 1–4, 8 and 24. The problem is trivial in dimension 1 and an exercise in dimension 2. The 3-dimensional version was the subject of a famous discussion between Isaac Newton and David Gregory: Newton thought it was 12, the kissing number of the face-centered cubic lattice, and Gregory thought there was space for a thirteenth ball. One reason that this is not a simple question is that the configuration of the face-centered cubic lattice is not rigid: we can move the balls while keeping the kissing

number equal to 12. In the end, Newton was right and this was proved by Schütte–van der Waerden [SvdW53] after many attempts by many mathematicians. The maximal kissing numbers in dimension 8 and 24 are those of the  $E_8$  lattice and the Leech lattice respectively, as proved by Odlyzko–Sloane and independently Levenšteĭn [OS79, Lev79]. Their proof is based on the linear programming method of Delsarte-Goethals-Seidel [DGS77]. Much more recently, the maximal kissing number in dimension 4 was determined by Musin [Mus08].

Finally, the maximal thickness of a covering of  $\mathbb{R}^n$  is only known in dimensions 1 and 2 and is realized by the hexagonal lattice in dimension 2 [Ker39]. Among lattices, the best coverings are known in dimensions 1–5 [Bam54, DR63, RB75, RB78] The best asymptotic bound known, as  $n \to \infty$ , is due to Rogers [Rog59] and states that there exists a constant c > 0 such that

 $\min \{\theta(\Lambda); \Lambda < \mathbb{R}^n \text{ a lattice of covolume } 1\} \leq n \cdot \log(n+1)^c$ 

for all  $n \ge 1$ .

#### 2.1.4 Linear programming

The results by Cohn–Kumar, Viazovska and Cohn–Kumar–Miller–Radchenko–Viazovska on the densest packings in dimensions 8 and 24 all start with a bound proved by Cohn–Elkies [CE03] and Gorbachev [Gor00] that we will briefly discuss now, mainly because we will treat a hyperbolic version of it in the next chapter.

We will state the Cohn–Elkies–Gorbachev theorem for lattices, because that is the case that corresponds best to hyperbolic manifolds. However, we emphasize that the bound can be adapted to general general sphere packings and the proof of that version is only slightly longer. In the theorem,  $\hat{f} : \mathbb{R}^n \to \mathbb{R}$  will denote the Fourier transform of  $f : \mathbb{R}^n \to \mathbb{R}$ , given by

$$\widehat{f}(\xi) = \int_{\mathbb{R}^n} f(x) \exp(-i\langle x,\xi\rangle) dx, \quad \xi \in \mathbb{R}^n$$

where  $\langle \cdot, \cdot \rangle : \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R}$  denotes the standard inner product and integration is performed with respect to the Lebesgue measure. We will call  $f : \mathbb{R}^n \to \mathbb{R}$  admissible if there exist constants  $C, \delta > 0$  such that

$$|f(x)| \leq \frac{C}{(1+||x||)^{n+\delta}}$$
 and  $\left|\widehat{f}(\xi)\right| \leq \frac{C}{(1+||\xi||)^{n+\delta}}$ 

for all  $x, \xi \in \mathbb{R}^n$ . Cohn-Elkies and Gorbachev proved:

**Theorem 2.1** (Cohn–Elkies–Gorbachev). Suppose there exists a non-zero admissible function  $f : \mathbb{R}^n \to \mathbb{R}$  such that

(a)  $f(x) \leq 0$  when  $||x|| \geq R$ ,

(b) 
$$f(\xi) \ge 0$$
 for all  $\xi \in \mathbb{R}^n$  and

(c) 
$$f(0) < \hat{f}(0)$$
.

Then sys $(\mathbb{R}^n/\Lambda) \leq R$  for all lattices  $\lambda < \mathbb{R}^n$  of covolume 1.

*Proof.* Suppose  $\Lambda < \mathbb{R}^n$  is a lattice of covolume 1 and whose shortest non-zero lattice vector has norm more than R. Then by the Poisson summation formula and properties (a) and (b) of f:

$$f(0) \ge \sum_{v \in \Lambda} f(v) = \sum_{w \in \Lambda^{\star}} \widehat{f}(w) \ge \widehat{f}(0).$$

This contradicts property (c) of f.

The first time one sees this proof, it seems amazing that anything can come out of this bound: in the displayed string of inequalities, we throw away infinitely many terms in two steps. Nonetheless Cohn and Elkies, by making a computer search for good test functions, already observed that their bound came very close to the systole of the  $E_8$  lattice and the Leech lattice in dimensions 8 and 24 respectively. This started the search for two "magical functions" that saturate the bounds in these dimensions, which were found by Viazovska and Cohn–Kumar–Miller–Radchenko–Viazovska. We refer to [Via17, CKM<sup>+</sup>17, Oes19] for more on this.

### 2.2 Regular graphs

The second parallel we would like to draw is with graph theory. Indeed, just like for flat tori, almost all the invariants that we are going to study for hyperbolic manifolds have their analogues in graph theory. Moreover, even if in graph theory the state of the art is more advanced, most of the analogous questions are also open and form the basis of a very active area of research. Like in the last section, we do not at all claim to provide an overview of the state of the art in graph theory, but will make a selection based on analogies with hyperbolic geometry.

The graphs that we will discuss are all regular: for  $k \in \mathbb{N}$ , a k-regular graph is a graph whose vertices all have degree k. Because 0-, 1- and 2-regular graphs are somewhat special, we will assume that  $k \ge 3$  in what follows.

One reason for the analogy between regular graphs and hyperbolic surfaces is that the (p+1)-regular tree can be identified with the coset space  $\mathrm{PGL}(2, \mathbb{Q}_p)/\mathrm{PGL}(2, \mathbb{Z}_p)$  (see for example [Ser77, Chapter 2]). Given a torsion-free lattice  $\Gamma < \mathrm{PGL}(2, \mathbb{Q}_p)$ , the quotient  $\Gamma \setminus \mathrm{PGL}(2, \mathbb{Q}_p)/\mathrm{PGL}(2, \mathbb{Z}_p)$  is a finite (p+1)-regular graph. So, like hyperbolic surfaces, we can describe graphs using  $2 \times 2$  matrices. This is not just a superficial connection, we will see that certain techniques (especially those related to arithmetic lattices) carry over from one side to the other.

#### 2.2.1 The girth

Just like for flat tori and manifolds, we can speak about the systole of a graph. In graph theory this is usually called the **girth** of the graph: the minimal length of a cycle in the graph. If G is a graph, we write  $\mu(G)$  for its girth. The simple question we are going to ask ourselves is: given  $k, n \in \mathbb{N}$ , what is the maximum possible girth of a k-regular graph on n vertices?

We start with an elementary upper bound:

**Lemma 2.2** (Moore bound). Let G be a k-regular graph of girth  $\mu$ , then the number of vertices of G is at least

$$1 + k \sum_{j=0}^{(\mu-3)/2} (k-1)^j \quad or \quad 2 \sum_{j=0}^{(\mu-2)/2} (k-1)^j$$

depending on whether  $\mu$  is odd or even respectively.

This bound implies that the girth of a k-regular graph on n vertices is bounded by  $2\log_{k-1}(n) + O(1)$  as  $n \to \infty$ . The proof is based on the volume growth of balls in the k-regular tree. Moreover, it is the same as that of the analogous bound in hyperbolic geometry (Lemma 3.1 below).

A graph that saturates the Moore bound is called a **Moore graph**. It turns out there are very few of them. For instance, Moore graphs of odd girth have almost been classified. There are two infinite families of Moore graphs of odd girth (cycles and complete graphs) and two sporadic examples (the Petersson graph and the Hoffman–Singleton graph) [BI73, Dam73] (see also [EJ08]). The classification is not complete because it is not known whether there exists a Moore graph of degree 57 and girth 5 or not. By definition, if this graph exists, it would have 3250 vertices. More generally, we know that the girth of a Moore graph of degree at least 3 cannot exceed 12.

The natural question that remains, and that also corresponds best to the questions that we are going to ask in hyperbolic geometry, is what happens when we fix the degree k and we make the number of vertices tend towards infinity. Even the question of whether the limit

$$\lim_{n \to \infty} \frac{\max \{\mu(G); G \text{ a } k \text{-regular graph on } n \text{ vertices}\}}{\log_{k-1}(n)}$$

exists and if so, what its value is, is open. The first constructions of regular graphs whose girth grows logarithmically as a function of the number of vertices is due to Erdős–Sachs [ES63]. They provided an algorithm that produces infinite sequences  $(G_n)_n$  of k-regular graphs on 2n vertices with girth  $\mu(G_n) \ge (1 + o(1)) \cdot \log(2n)$ , as  $n \to \infty$ . Improving on this construction is a well known and difficult problem that has plenty of history [Mar82, Imr84, Wei84b, LPS88, Mar88, Mor94, Big98, MWW04, Dah14, LS21]. The best upper bound on the asymptotic girth ratio above is still the bound given by the Moore bound. The best lower bound comes from the Ramanujan graphs of Margulis [Mar88] and Lubotzky– Phillips–Sarnak [LPS88] and their generalizations by Morgenstern [Mor94] and states that when the degree equals  $k = p^m + 1$  with p an odd prime, then there exists a sequence of k-regular graphs  $(G_{n_j})_j$  on  $n_j$  vertices such that  $n_j \to \infty$  and

$$\lim_{j \to \infty} \frac{\mu(G_{n_j})}{\log_{k-1}(n_j)} = \frac{4}{3}$$

The upper bound in the above is due to Biggs-Boshier [BB90]. The graphs involved are in fact Cayley graphs of  $PGL(2, \mathbb{Z}/q\mathbb{Z})$  of degree  $p^m + 1$ .

#### 2.2.2 The kissing number

In analogy with flat tori, we can also define a **kissing number** of a finite graph: the number of distinct oriented cycles in the graph that realize its girth. Together with Maxime Fortier Bourque [FBP22], we proved that Moore graphs can be characterized by their kissing number:

**Theorem 2.3.** If G is a connected graph of maximal degree k and girth g on n vertices, then their kissing number satisfies

$$\operatorname{Kiss}(G) \leqslant \frac{nk(k-1)^{\lfloor g/2 \rfloor}}{g}$$

with equality if and only if G is a Moore graph.

Our proof is elementary and is heavily inspired by Parlier's proof of a similar inequality in hyperbolic geometry [Par13]. For regular graphs, our bound improves on earlier inequalities due to Teo–Koh [TK92], Azarija–Klavžar [AK15].

Using the Moore bound, we can remove the dependence on girth from this theorem and obtain for instance that, a k-regular graph on n vertices has kissing number  $\operatorname{Kiss}(G) \ll n^2/\log(n)$ . Like the Moore bound, it is unclear whether this is sharp. It is easy, for instance using cyclic covers, to construct sequences of regular graphs of any degree  $k \ge 3$  whose kissing number grows linearly as a function of their number of vertices.

Together with Maxime Fortier Bourque [FBP22], we showed that the Ramanujan graphs of Lubotzky–Phillips–Sarnak can be used to obtain examples of super-linear growth:

**Theorem 2.4.** For every prime number  $p \equiv 1 \mod 4$  there is a subsequence  $(X^{p,q_k})_k$  of the (p+1)-regular graphs  $X^{p,q}$  of Lubotzky–Phillips–Sarnak such that

$$\lim_{k \to \infty} \frac{\log(\operatorname{Kiss}(X^{p,q_k}))}{\log(n_k)} = \frac{4}{3}$$

where  $n_k$  is the number of vertices of  $X^{p,q_k}$ .

Like Lubotzky–Phillips–Sarnak, we use bounds on the number of integral solutions to certain quadratic equations to analyze their geodesics, but in a different regime.

#### 2.2.3 The spectral gap

For graphs, unlike for tori but just like for hyperbolic manifolds, the spectral questions are distinct from the geometric questions. The question on the spectral gap that we will discuss here can be thought of as a question on connectivity. Indeed, the main reason Lubotzky–Phillips–Sarnak [LPS88] and Margulis [Mar88] introduced their graphs, was to provide examples of explicit optimal expander graphs. An expander sequence is a sequence of graphs that are both sparse (their degree is bounded) and highly connected. For an overview of the subject, we refer to [HLW06]. Here we will just briefly discuss some of what is known about Ramanujan graphs.

There are multiple ways to measure the connectivity of a graph G. In this section we will use the spectral gap of its adjacency matrix  $A_G$  (or equivalently, of its graph Laplacian). If G is a connected k-regular graph on n vertices then its adjacency matrix has real eigenvalues:

$$\lambda_1(G) = k > \lambda_2(G) \ge \lambda_3(G) \ge \ldots \ge \lambda_n(G) \ge -k$$

and G is **bipartite** if and only if  $\lambda_n(G) = -k$ . Recall that G = (V, E) is bipartite if we can write  $V = V_1 \sqcup V_2$  and there are no edges interior to  $V_1$ , nor to  $V_2$ .

We set

$$\lambda(G) = \max\{|\lambda_i(G)|; \lambda_i(G) \neq \pm k\}.$$

This number can be seen as a measure of connectivity of G for instance because it shows up in the mixing rate of a random walk on G and also because it relates to the Cheeger constant of G (that we will briefly discuss in Section 6.3). The general rule is that the smaller  $\lambda(G)$  is, the more connected G is.

So, the natural extremal question is how small  $\lambda(G)$  can be. The Alon-Boppana bound [Alo86] states that for a k-regular graph G,

$$\lambda(G) \ge 2\sqrt{k-1} - \varepsilon(G)$$

where the error term  $\varepsilon(G) \to 0$  as the number of vertices of G tends to infinity. The significance of the number  $2\sqrt{k-1}$  is that the spectrum of the adjacency matrix of the infinite k-regular tree is contained in  $[-2\sqrt{k-1}, 2\sqrt{k-1}]$ . A sequence  $(G_n)_n$  of k-regular graphs on n vertices is called an **expander sequence** if  $\lambda(G_n)$  is uniformly bounded from above. A k-regular graph G is called **Ramanujan** if  $\lambda(G) \leq 2\sqrt{k-1}$ .

The construction of Margulis and Lubotzky–Phillips–Sarnak yields Ramanujan graphs of degree  $p^m+1$  for any odd prime p. They can be made bipartite, in which case they're Cayley graphs of PGL(2,  $\mathbb{Z}/q\mathbb{Z}$ ), or not, in which case they're Cayley graphs of PSL(2,  $\mathbb{Z}/q\mathbb{Z}$ ).

Besides having very large girth and an asymptotically optimal spectral gap, they also have very small but not optimal diameter [LPS88, Sar19].

More recently, using a completely different argument, Marcus–Spielman–Srivastava [MSS15] proved that there exist infinite sequences of bipartite Ramanujan graphs of every fixed degree. The non-bipartite case is currently still open.

#### 2.2.4 Random graphs

The final subject we discuss before we get to hyperbolic geometry is that of random regular graphs. Also here, we will just highlight some aspects that have analogues in hyperbolic geometry and refer to the specialized literature (for example [Bol01, JŁR00, Wor99]) for a more complete overview.

The configuration model. An important model for a random k-regular graph is the configuration model: we take n (an even number if k is odd) vertices with k half-edges emanating from each of them and uniformly randomly pair up the half-edges. Every pairing is allowed, so in particular, there is a chance the result is disconnected. However, this probability is of the order  $O(n^{-1})$  as  $n \to \infty$  [Bol81, Wor81a]. We will denote the resulting random regular graph by  $G_{n,k}$ .

One result of which we will see many analogues in hyperbolic geometry is a classical result by Bollobás [Bol80] and Wormald [Wor81b], who proved that the number of short cycles in  $G_{n,k}$  converges to a Poisson distributed random variable as  $n \to \infty$ . For sharper versions of this result, see [MWW04, Joh15].

The graphs  $G_{n,k}$  are also highly connected. The first example of this is their diameter. Before making this precise, we note that the Moore bound (Lemma 2.2) in the case of odd girth is equivalent to:

**Lemma 2.5** (Moore bound for the diameter). Let G be a k-regular graph of diameter d, then G has at most

$$1 + k \sum_{j=0}^{d-1} (k-1)^j$$

vertices.

In particular, if  $(G_n)_n$  is a sequence of k-regular graphs on n vertices and  $k \ge 3$ , then

diam
$$(G_n) \ge (1 + o(1)) \cdot \log_{k-1}(n), \text{ as } n \to \infty.$$

We will provide a proof of a hyperbolic version of this bound, that uses the exact same idea, in Section 6.2. Bollobás–Fernandez-de-la-Vega [BFdlV82] proved that random regular

graphs saturate this bound. That is,

$$\frac{\operatorname{diam}(G_{n,k})}{\log_{k-1}(n)} \longrightarrow 1$$

in probability as  $n \to \infty$ .

Finally,  $G_{n,k}$  is also nearly Ramanujan, this was a conjecture by Alon [Alo86] and was proved by Friedman [Fri08]. Concretely, for all  $\varepsilon > 0$ ,

$$\mathbb{P}\left(\lambda(G_{n,k}) \leqslant 2\sqrt{k-1} + \varepsilon\right) \longrightarrow 1.$$

as  $n \to \infty$ . A shorter proof and generalizations of this theorem to random finite degree covers of graphs were found in [Bor20, BC19, BC23].

**Random Cayley graphs.** Another model of random regular graphs that has been studied is that of random Cayley graphs. That is, we take a sequence of finite groups  $(Q_n)_n$  and let  $S_n \subset Q_n$  be a random subset. This yields a random graph

$$G_n = \operatorname{Cay}(Q_n, S_n \cup S_n^{-1}),$$

the Cayley graph for  $Q_n$  with edges determined by  $S_n \cup S_n^{-1}$ , where  $S_n^{-1} = \{s^{-1}; s \in S\}$ .  $G_n$  is connected if and only if  $S_n$  is a generating set for  $Q_n$ , which is typically the case for the sequences of groups that are considered.

Random Cayley graphs were first studied for sequences of random generating sets whose size grows with the size of the group [AR93, Pak99, LR04, LS04b]. We will consider the case in which  $S_n \subset Q_n$  is a uniformly random subset of size k. The case that is best understood is when the sequence of groups is  $(\mathbb{G}(\mathbb{F}_q))_q$ , where  $\mathbb{G}$  is a simple group of fixed Lie type and fixed rank over  $\mathbb{F}_q$ .

In [GHS<sup>+</sup>09], Gamburd–Hoory–Shahshahani–Shalev–Virág proved that, with high probability as  $q \to \infty$ , the girth of a random 2k-regular Cayley graph  $G_q$  of  $\mathbb{G}(\mathbb{F}_q)$  satifies

$$\mu(G_q) \ge \left(\frac{1}{\dim(\mathbb{G})} + o(1)\right) \cdot \log_{2k-1}(|\mathbb{G}(\mathbb{F}_q)|).$$

Combined with work by Bourgain–Gamburd [BG08], this also implies these graphs are expanders in the case of  $SL(2, \mathbb{Z}/p\mathbb{Z})$ . This was generalized to all finite simple groups of Lie type by Breuillard–Green–Guralnick–Tao [BGGT15].

For bounds in the case of random Cayley graphs of symmetric groups and finite simple groups of Lie type of growing rank over a fixed finite field, we refer to [Ebe17, LS19, EJ22].

## 3 Extremal problems I

The goal of this chapter is to present a first selection of results on extremal problems in hyperbolic geometry. A big role in this chapter will be played by our joint project with Maxime Fortier Bourque [BP22, BP23b, BP23c]. In this project we use the Selberg trace formula to prove bounds on spectral and geometric invariants of hyperbolic manifolds, in a way that is analogous to that of Cohn–Elkies–Gorbachev [Gor00, CE03] (see Section 2.1.4). With this method we were able to prove bounds on various invariants with a uniform proof strategy, whereas the best bounds before used very varied methods.

This chapter will also discuss constructions of hyperbolic surfaces of large systole, based on joint work with Alexander Walker [PW18] and on [Pet18].

### 3.1 The systole

Perhaps the simplest geometric invariant of a hyperbolic manifold M is its **systole** sys(M): the minimal length of a closed geodesic on M. Equivalently, this is the minimum length of a closed essential curve<sup>1</sup>. Figure 3.1 shows an example.



Figure 3.1: A curve that realizes the systole.

It is not difficult to see that there are hyperbolic surfaces of finite (and bounded) area

<sup>&</sup>lt;sup>1</sup>Essential means that the curve is not homotopic to a point and that there does not exist a family of curves, freely homotopic to the given curve, which leave any compact subset of M.

and arbitrarily small systole. In dimension 3, Dehn surgery techniques – in particular Thurston's convergence theorem (see for example [BP92, Chapter E] or [Mar22, Chapter 15]) – imply the analogous result. In higher dimension, the fact that there exists only a finite number of isometry classes of hyperbolic manifolds of volume  $\leq v$  and dimension d for all v > 0 and  $d \geq 4$  [Wan72] implies that the volume of a sequence of manifolds whose systole tends towards 0, necessarily tends towards infinity. However, the existence of hyperbolic manifolds of finite volume and arbitrarily small systole is known for any dimension from the work of Agol and Belolipetsky–Thomson [Ago06, BT11].

The simple question that remains is:

Question 3
Given $d \ge 2$ and $v > 0$ , what is
$\max \left\{ sys(M); \begin{array}{c} M \text{ a hyperbolic } d\text{-manifold} \\ \text{of volume } \leqslant v \end{array} \right\}  ?$
And which hyperbolic manifold realizes it?

The fact that this quantity is indeed a maximum and not just a supremum is not trivial. In dimension 2 it is a theorem of Mumford [Mum71], in dimension 3 it is a consequence of the work of Jørgensen–Thurston [Thu02, Theorem 5.12.1] and in dimension at least four Wang's theorem mentioned above implies it. In dimension 2, the function sys :  $\mathcal{M}_g \rightarrow$  $(0, \infty)$  is a topological Morse function [Akr03], so in theory it could be used to understand the topology of  $\mathcal{M}_g$ . However, this currently seems out of reach, even counting the number of local maxima is a very hard problem [FBR22].

Like for all the geometric invariants that we will consider, there is a classical upper bound on systole and the real question is whether this bound is close to being optimal. This classical bound is the analogue of the Moore bound (see Section 2.2.1) and states:

**Lemma 3.1.** For every  $d \ge 2$  there exists a constant  $c_d > 0$  such that

$$\operatorname{sys}(M) \leq \frac{2}{d-1} \log \left( \operatorname{vol}(M) \right) + c_d,$$

for every closed hyperbolic d-manifold M.

*Proof.* Take any point  $p \in M$ . The open ball B(p, sys(M)/2) of radius sys(M)/2 and center p is isometric to an open ball in  $\mathbb{H}^n$  of the same radius.

#### 3.1. THE SYSTOLE

Indeed, if not, there would be two distinct geodesic segments between p and another point  $q \in B(p, \operatorname{sys}(M)/2)$ , both shorter than  $\operatorname{sys}(M)/2$ . The loop that these segments form is strictly less than  $\operatorname{sys}(M)$  and can't be contractible (if it were contractible, we could lift it to a geodesic bigon in  $\mathbb{H}^d$  and such bigons don't exists). This contradicts the fact that the systole is realized by the shortest closed loop in M.

Now take an open ball  $B(\hat{p}, \text{sys}(M)/2)$  of radius sys(M)/2 around  $\hat{p} \in \mathbb{H}^d$ . The observation above implies that

$$\operatorname{vol}(\mathbb{S}^{d-1}) \cdot \int_0^{\operatorname{sys}(M)/2} \sinh^{d-1}(t) dt = \operatorname{vol}(B(\hat{p}, \operatorname{sys}(M)/2)) \leqslant \operatorname{vol}(M).$$

This implies the lemma.

If we look closely at the proof above, we observe that in reality, the lemma above is a bound *maximal* injectivity radius. The fact that such a bound implies a bound on the systole is due to the fact that on a negatively curved manifold, the systole is twice the *minimal* injectivity radius.

#### 3.1.1 The maximal systole

For closed hyperbolic manifolds of dimension more than two, Lemma 3.1 is the best known upper bound. In dimension 2, better bounds are known. In [Bav96], Bavard proved that the maximum injectivity radius injrad(M) of a closed hyperbolic surface M of genus g satisfies:

$$\operatorname{injrad}(M) \leq \operatorname{arccosh}\left(\frac{1}{2\sin(\pi/(12g-6))}\right) \stackrel{g \to \infty}{=} 2\log(g) + 2.680353\ldots + o(1).$$

Moreover, as a bound on the radius of maximal injectivity radius, this bound is sharp: there exist hyperbolic surfaces of all genera  $g \ge 2$  for which the inequality is an equality. On the other hand, Bavard already noted in his paper that the bound that we derive on the systole cannot be sharp, because of the fact that there exists only a finite number of points in a closed hyperbolic surface at which the injectivity radius is maximal. However, Bavard's bound is better than that of Lemma 3.1, even if it is still based on the maximal injectivity radius.

Recently, with Maxime Fortier Bourque [BP23c], we found the first bound that is better than the bound coming from the maximal injectivity radius. We proved:

**Theorem 3.2.** There exists a genus  $g_0 \ge 2$  such that

$$sys(X) < 2\log(g) + 2.409.$$

for every oriented closed hyperbolic surface X of genus  $g \ge g_0$ .

Our proof uses a hyperbolic version of the Cohn–Elkies–Gorbachev method from Section 2.1.4. The proof of the Cohn–Elkies–Gorbachev bound is based on the Poisson summation formula. The analogue of this formula in hyperbolic geometry is the **Selberg trace formula**:

#### The Selberg trace formula

Let X be a closed, oriented and connected hyperbolic surface of genus g. Let  $0 = \lambda_0(X) < \lambda_1(X) \leq \ldots$  denote the eigenvalues of the Laplacian on X and  $\mathcal{C}(X)$  the set of closed oriented geodesics on X. Then

$$\sum_{j=0}^{\infty} \widehat{f}\left(\sqrt{\lambda_j(X) - \frac{1}{4}}\right) = 2(g-1) \int_0^{\infty} \widehat{f}(x) x \tanh(\pi x) \, dx + \frac{1}{\sqrt{2\pi}} \sum_{\gamma \in \mathcal{C}(X)} \frac{\Lambda(\gamma) f(\ell(\gamma))}{2\sinh(\ell(\gamma)/2)} dx + \frac{1}{\sqrt{2\pi}} \sum_{\gamma \in \mathcal{C}(X)} \frac{\Lambda(\gamma) f(\ell(\gamma))}{2\hbar(\ell(\gamma)/2)} dx + \frac{1}{\sqrt{2\pi}}$$

for every even function  $f : \mathbb{R} \to \mathbb{C}$  for which there exists an  $\varepsilon > 0$  such that

- the Fourier transform  $\widehat{f}(\xi) = \frac{1}{\sqrt{\pi}} \int_{-\infty}^{\infty} f(x) e^{-ix \cdot \xi} dx$  is holomorphic on the strip  $\{z \in \mathbb{C}; |\operatorname{Im}(z)| < \frac{1}{2} + \varepsilon\}$  and
- $\left| \widehat{f}(\xi) \right| = O\left( (1 + |\xi|)^{-2-\varepsilon} \right)$  uniformly on  $\left\{ z \in \mathbb{C}; |\operatorname{Im}(z)| < \frac{1}{2} + \varepsilon \right\}.$

In this formula,  $\ell(\gamma)$  denotes the length of  $\gamma \in \mathcal{C}(X)$  and  $\Lambda(\gamma)$  the primitive length of  $\gamma$ : the length of the unique primitive geodesic  $\gamma_0$  for which there exists a  $k \ge 1$  with  $\gamma_0^k = \gamma$ .

A function  $f : \mathbb{R} \to \mathbb{C}$  that satisfies the conditions above is called **admissible**. We also observe that, if f is admissible,  $\hat{f}$  is even and the expression on the right in the trace formula is therefore well defined: we can choose the square root we want. Selberg proved this formula in [Sel56]. Expositions can be found in [Bus10, Iwa02, Ber11].

Theorem 3.2 is based on the following bound (a hyperbolic analogue of Theorem 2.1):

**Theorem 3.3.** Let  $g \ge 2$ . Let f be a non-constant admissible function for which there exists a constant R > 0 such that:

•  $f(x) \leq 0$  if  $x \geq R$ ;

• 
$$\widehat{f}(\xi) \ge 0$$
 for all  $\xi \in \mathbb{R} \cup i\left[-\frac{1}{2}, \frac{1}{2}\right];$ 

•  $\widehat{f}(i/2) \ge 2(g-1)\int_0^\infty \widehat{f}(x)x \tanh(\pi x) dx.$ 

Then  $sys(X) \leq R$  for every  $X \in \mathcal{M}_g$ .

*Proof.* Suppose there exists a hyperbolic surface  $X \in \mathcal{M}_g$  such that  $\operatorname{sys}(X) > R$ . Then  $\operatorname{sys}(Y) > R$  for every  $Y \in \mathcal{M}_g$  sufficiently close to X. This in particular implies that  $f(\ell(\gamma)) \leq 0$  for every  $\gamma \in \mathcal{C}(Y)$ . Moreover,  $\widehat{f}\left(\sqrt{\lambda_j(Y) - \frac{1}{4}}\right) \geq 0$  for every  $j \geq 0$ . So the trace formula gives us that

$$\begin{aligned} \widehat{f}(i/2) &\leqslant \sum_{j=0}^{\infty} \widehat{f}\left(\sqrt{\lambda_j(Y) - \frac{1}{4}}\right) \\ &= 2(g-1) \int_0^{\infty} \widehat{f}(x) x \tanh(\pi x) \, dx + \frac{1}{\sqrt{2\pi}} \sum_{\gamma \in \mathcal{C}(Y)} \frac{\Lambda(\gamma) f(\ell(\gamma))}{2 \sinh(\ell(\gamma)/2)} \\ &\leqslant 2(g-1) \int_0^{\infty} \widehat{f}(x) x \tanh(\pi x) \, dx \\ &\leqslant \widehat{f}(i/2) \end{aligned}$$

for every Y sufficiently close to X. Then,

$$\widehat{f}\left(\sqrt{\lambda_j(Y) - \frac{1}{4}}\right) = 0 \text{ pour tout } j \ge 1.$$

Seeing how  $\hat{f}$  is holomorphic on a strip (and not constantly equal to zero), the roots of  $\hat{f}$  are isolated. This, combined with the fact that the eigenvalues depend continuously on the metric (see for example [BU83]), implies that all  $\lambda_j(Y)$  are constant on an open neighborhood of  $X \in \mathcal{M}_g$ . This is a contradiction because there are no continuous isospectral deformations of a hyperbolic surface [Gel63].

Another way of formulating Theorem 3.3 is to say that if we define, for R > 0,

$$g_{\min}(R) := \min \left\{ g \ge 2; \begin{array}{c} \text{there exists } X \in \mathcal{M}_g \text{ such that} \\ \operatorname{sys}(X) > R \end{array} \right\}$$

and

$$G(R) := \sup \left\{ \begin{array}{ll} 1 + \frac{f(i/2)}{2\int_0^\infty \hat{f}(x)x \tanh(\pi x)dx}; & f \text{ admissible, non-constant,} \\ \hat{f}(\xi) \ge 0 \text{ for all } \xi \in \mathbb{R} \cup i\left[-\frac{1}{2}, \frac{1}{2}\right] \end{array} \right\},$$

then

$$g_{\min}(R) \ge \mathcal{G}(R)$$

for every R > 0.

To prove the Theorem 3.2, we found a series of explicit functions that give rise to very good bounds. We don't have a proof that the functions we found are (near) optimal (asymptotically, as  $g \to \infty$ ). Nonetheless, for example based on the numerical results presented below, we think that they are not far from optimal for Theorem 3.3. We observe that the multiplicative constant in front of the logarithm is equal to 2 in all three limits above (Lemma 3.1, Bavard's bound and Theorem 3.2). The best known examples are of systole  $\sim \frac{4}{3} \log(\text{genus})$  (see Section 3.1.2 below), so it is not even clear that the multiplicative constant is optimal. On the other hand, we think that our method cannot improve this multiplicative constant:

#### Question 4

Let R(g) the optimal bound we can prove on the systole of a closed hyperbolic surface of genus g with Theorem 3.3. that is,  $R(g) := \inf \{R > 0; G(R) > g\}$ . Is it true that

$$R(g) = 2\log(g) + O(1)$$
 as  $g \to \infty$ ?

Moreover, what are the optimal functions?

Theorem 3.3 also lends itself to numerical optimization. With Maxime Fortier Bourque, we had our computers search, with SageMath, for optimal functions for Theorem 3.3. A posteriori, we used the functions found to prove rigorous numerical bounds on systole in genus 2–20. The result is displayed in Figure 3.2. For the exact numbers we found, we refer to [BP23c]. Our method improves the bound in all genera, except in genus 2, where the maximizer is known: the Bolza surface [Jen81].



Figure 3.2: The bound on systole in genus 2–20.

#### 3.1.2 Arithmetic surfaces with large systoles

Up until now, we have only considered upper bounds on the maximal possible systole. To find lower bounds, we need examples of manifolds of large systole. In dimension 2, the best known examples are those found by Brooks [Bro88] and Buser–Sarnak [BS94]. Their proof uses an arithmetic construction and was generalized by Katz–Schaps–Vishne [KSV07] in dimensions 2 and 3 and by Murillo [Mur19] in higher dimensions. Brooks and Buser–Sarnak prove the existence of sequences  $(X_k)_k$  of closed hyperbolic surfaces of genus  $g_k$ , such that  $g_k \to \infty$  and

$$\operatorname{sys}(X_k) \ge \frac{4}{3}\log(g_k) + O(1) \quad \text{as } k \to \infty.$$

To illustrate the idea, we will give the proof by Brooks and Buser–Sarnak in the noncompact case. The proof is the same as in the compact case, except that the compact case uses congruence coverings of a closed hyperbolic surface coming from a quaternion algebra instead of  $SL(2,\mathbb{Z})\setminus\mathbb{H}^2$ . The graph-theoretic analogue of this proof is also behind the fact that the girth of the Ramanujan graphs of Lubotzky–Phillips–Sarnak (see Section 2.2.3) is logarithmic. Let

$$\Gamma(N) = \left\{ \begin{bmatrix} a & b \\ c & d \end{bmatrix} \in \operatorname{PSL}(2, \mathbb{Z}); \begin{array}{c} a \equiv d \equiv 1 \mod N, \\ b \equiv c \equiv 0 \mod N \end{array} \right\}$$

denote the principal congruence subgroup of level N of  $PSL(2, \mathbb{Z})$ .

**Lemma 3.4** (Brooks, Buser–Sarnak). The systole of  $X(N) = \Gamma(N) \setminus \mathbb{H}^2$  satisfies

$$\operatorname{sys}(X(N)) \ge 2 \operatorname{arccosh}\left(\frac{N^2 - 2}{2}\right).$$

*Proof.* Let  $\begin{bmatrix} a & b \\ c & d \end{bmatrix} \in \Gamma(N)$  be a hyperbolic element. Then there exist  $x_0, x_1, x_2, x_3 \in \mathbb{N}$  such that

 $a = 1 + x_0 N$ ,  $b = x_2 N$ ,  $c = x_3 N$  and  $d = 1 + x_1 N$ .

Moreover, using the determinant, we get

$$0 = ad - bc - 1 = (x_0 + x_1)N + (x_0x_1 - x_2x_3)N^2.$$

So, N divides  $x_0 + x_1$ . Given that the element isn't parabolic, it's not possible that  $x_0 + x_1 = 0$ , hence

$$\left| \operatorname{tr} \left( \left( \begin{array}{c} a & b \\ c & d \end{array} \right) \right) \right| \ge N^2 - 2$$

which implies the lemma.

To compare the systole with the area of X(N), we note that

$$\operatorname{area}(X(N)) = \frac{\pi}{6} N^3 \prod_{p|N \text{ prime}} 1 - \frac{1}{p^2} \quad \text{for } N \ge 3$$

This quantity depends on the number of divisors of N, but the quantity

$$\operatorname{area}(X(N))/N^3$$

is uniformly bounded from both sides. In particular,

$$2\operatorname{arccosh}\left(\frac{N^2-2}{2}\right) = \frac{4}{3}\log(\operatorname{area}(X(N))) + O(1) \quad \text{as } N \to \infty.$$

One way to obtain closed surfaces of large systole from this construction is to compactify

them using the method of Brooks [Bro99]. This gives the same multiplicative constant  $(\frac{4}{3})$  in front of the logarithm. It is known that the multiplicative constant  $\frac{4}{3}$  is optimal for principal congruence coverings of arithmetic surfaces [Mak13]. Moreover, it is still the state of the art, that is to say, at the time of writing, we know no more than:

$$\limsup_{g \to \infty} \frac{\max\left\{ \operatorname{sys}(X); \ X \in \mathcal{M}_g \right\}}{\log(g)} \in \left[\frac{4}{3}, 2\right].$$

Schmutz-Schaller has conjectured that  $\frac{4}{3}$  is the actual value of this limit supremum [SS98].

#### 3.1.3 A combinatorial construction

In addition to arithmetic constructions, we currently know two other constructions of closed hyperbolic surfaces of logarithmic systole. The aim of this section and the next is to present these two constructions.

Both constructions are very strongly inspired by ideas from graph theory. The first is the result of joint work with Alexander Walker [PW18] and is a hyperbolic version of the construction of Erdős–Sachs [ES63] mentioned in Section 2.2.1. The construction is based on gluing together ideal hyperbolic triangles along their sides. This is similar to the model of random surfaces of Brooks–Makover [BM04], except that we glue carefully instead of randomly.

Since the side lengths of ideal triangles are infinite, there is an infinite number of isometries that can be used for the gluing. We will use **shear** 0 gluings. One way to measure this shear is to draw the two orthogonal projections onto the side we glue of the two vertices which are *not* incident to that side. The signed hyperbolic distance (using the orientation of the surface) between these two points is the shear. So, in our gluings we require the two orthogonal projections to coincide.

The goal is now to find a gluing such that there are no short geodesics on the resulting surface. With Alexander Walker, we found sequences of gluings  $(S_N)_N$  of 2N triangles, such that their systoles are at least

$$\log(N) - \log\log(N) - O(1) = \log(\operatorname{area}(S_N)) - \log\log(\operatorname{area}(S_N)) - O(1)$$
 (3.1.1)

as  $N \to \infty$ . Our construction also comes with some flexibility. We can for instance create such a sequence in which the systole has multiplicity K for any fixed  $K \ge 1$ .

Before we describe the construction, we begin with some observations about the geometry and topology of a triangulated surface. Since it is a surface with cusps, its fundamental group is free. In fact, it's isomorphic to the fundamental group of the dual graph – the graph we obtain if we add a vertex to each triangle and we draw edges between the vertices for each side the corresponding triangles share. The orientation of the surface also induces an orientation of the dual graph: a cyclic order on the set of edges incident to each vertex. Given an essential closed curve<sup>2</sup>  $\gamma$  in the surface, the length of the unique geodesic in the free homotopy class of  $\gamma$  – which we denote  $\ell(\gamma)$  – can be calculated using the orientation of the graph. Indeed, we trace the reduced circuit homotopic to  $\gamma$  and each time we cross a vertex we write down an L or R, depending on whether the circuit turns left or right with respect to the orientation. This gives us a word w in L and R. If we now replace the letters L and R by the matrices

$$L = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix} \quad \text{and} \quad R = \begin{pmatrix} 1 & 0 \\ 1 & 1 \end{pmatrix},$$

then w becomes a matrix and

$$\ell(\gamma) = 2 \operatorname{arccosh}\left(\frac{\operatorname{tr}(w(\gamma))}{2}\right)$$

There is some ambiguity in the definition of  $w(\gamma)$ : if we start at a different vertex or read in the opposite direction, then  $w(\gamma)$  changes. However, its trace does not and hence the expression on the right above is well-defined.

To construct our surfaces with large systoles, we need to find a triangulated surfaces whose dual graph has no circuits that carry words of small trace. If we want to obtain surfaces which satisfy the bound of the equation (3.1.1), we need to construct surfaces which do not carry closed curves of trace  $\langle k \rangle$  and which contain  $\approx k^2 \log(k)$  triangles.

To do this, we start with a gluing of  $\approx k^2 \log(k)$  triangles such that each triangle is incident to exactly two other triangles, as in Figure 3.3. We choose this initial gluing in such a way that the surface does not carry curves of trace  $\langle k \rangle$ . One way to ensure this is to start with a single cycle of  $2N \approx k^2 \log(k)$  triangles that carries a word of the form  $(LR)^N$ .

Now, we need to find a gluing of the boundary of this surface that does not create any circuits of trace  $\langle k$ . It is in this part of the proof that the ideas of Erdős et Sachs

<sup>&</sup>lt;sup>2</sup>Not homotopic to a point nor to a cusp.

#### 3.1. THE SYSTOLE



Figure 3.3: A cycle of triangles.

intervene. The idea is to perform an iterative construction. During every step, we perform one of the following two operations:

- If there is still a pair of sides that can be glued together without creating short circuits, we do so (if there are several, we make an arbitrary choice),
- if not, but the surface still has boundary, there must be two sides of two triangles (say  $c_1$  and  $c_2$ ) on the boundary that are too "close" to glue together. On the other hand, the fact that our surface contains enough triangles ( $\approx k^2 \log(k)$ ) guarantees that there exists a pair of triangles sufficiently "far" from  $c_1$  and  $c_2$ , whose sides  $c_3$  and  $c_4$  (which were on the boundary of the initial surface) were glued together earlier in the process. We open up this gluing and we glue  $c_3$  to  $c_1$  and  $c_2$  to  $c_4$ .

Since the number of glued sides increases in each step, there is no boundary left at the end and we obtain an surface  $S_N$  without boundary whose systole satisfies the equation (3.1.1). The approximation that makes the second operation above work is [PW18]:

**Proposition 3.5.** The positive semigroup generated by L and R is

$$\langle L, R \rangle_{+} = \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}(2, \mathbb{Z}); \ a, b, c, d \ge 0 \right\} =: \mathrm{SL}(2, \mathbb{N}).$$

Moreover,

$$\{A \in \mathrm{SL}(2,\mathbb{N}); \ 3 \leqslant \mathrm{tr}(A) \leqslant k\} = O\left(k^2 \log(k)\right)$$

as  $k \to \infty$ .

For more details, we refer to [PW18]. Like in the previous section, the results of Brooks [Bro99] can be used to compactify the surfaces we obtain, which gives rise to sequences of closed surfaces  $(X_k)$  whose genus tends to infinity and whose systole is at least  $(1 + o(1)) \cdot \log(\text{genus}(X_k))$ .

#### 3.1.4 Long systolic pants decompositions

The second construction produces a hyperbolic surface of logarithmic systole such that the geodesics that realize the systole form a pants decomposition of the surface. Here, a **pants decomposition** is a collection of simple closed curves such that the complement of these curves is a collection of **pairs of pants**: three-holed spheres. A hyperbolic metric with a totally geodesic boundary on a pair of pants is determined entirely by the boundary lengths. So one way to specify a hyperbolic metric on a closed surface, is to first choose a pants decomposition and then specify two parameters per curve in the decomposition: a length and a **twist** which describes the gluing of the two pants along the curve (see Figure **3.4** for an example).

A surface such that the geodesics that realize systole form a pants decomposition is not difficult to obtain: we take our favorite pants decomposition and pinch the curves to a sufficiently small length. The collar lemma [Kee74] (see also [Bus10, Chapter 4]) implies that the curves in the pants decomposition are the shortest in the surface. What is more difficult is to do this without creating surfaces with very small systoles. The first such construction is due to Buser [Bus78]. He finds a sequence of surfaces  $(X_g)_g$  of genus gwhose systole is  $\approx \sqrt{\log(g)}$ . In [Pet18], this result was improved as follows:

**Theorem 3.6.** There exists a sequence of closed orientable hyperbolic surfaces  $(X_g)_{g \ge 2}$ such that  $X_g$  has genus g and

$$\operatorname{sys}(X_g) \ge \frac{4}{7}\log(g) + O(1) \quad as \ g \to \infty.$$

The idea of the proof is to consider the following set:

 $S_a = \begin{cases} Closed hyperbolic surfaces of systole a, that admit a pants decomposition \\ - such that all the lengths in the decomposition are a \\ - and all the twists are <math>a/4 \end{cases}$ 

Afterwards, the proof has two steps. The first step is to show that  $\mathcal{S}_a$  is not empty.


Figure 3.4: A gluing of two hyperbolic pairs of pants of twist a/4. Here the twist is the hyperbolic distance between the two points on the curve where the orthogonal geodesics connecting the curve along which we glue with the other boundary components intersect the curve.

The idea is that twisting by a/4 forces curves that are not in the pants decomposition to be long. In particular, the only way to have a short curve in a surface with a pants decomposition as in the definition of  $S_a$  is to have a short cycle in the graph dual to the pants decomposition. So if we have a surface with a pants decomposition all of whose lengths are a and all of whose twists are a/4 and whose dual graph is of a sufficiently large girth, then this surface is in  $S_a$ .

The second step is to define

$$g_{\min}(a) = \min\left\{\operatorname{genus}(X); X \in \mathcal{S}_a\right\}$$

Given that  $S_a$  isn't empty,  $g_{\min}(a)$  is finite. The goal now is to find an upper bound on  $g_{\min}(a)$ . To do so we use the diameter of the surface  $X_{\min}(a)$  realizing  $g_{\min}(a)$ . If this diameter is very large compared to a, then  $X_{\min}(a)$  contains two pairs of pants  $P_1$  and  $P_2$  that are very far away from each other. Because of this, we can remove  $P_1$  and  $P_2$  and glue the resulting boundary components together without creating any curves of length below a. So we obtain a surface whose systole is still a, but whose genus is below that of  $X_{\min}(a)$ , thus contradicting minimality.

In conclusion, we obtain a bound on the diameter of  $X_{\min}(a)$  in terms of its systole. The diameter of a hyperbolic surface is at least logarithmic as a function of its genus (see Section 6.2 for a detailed discussion), so we obtain a lower bound on the systole in terms of the genus, which is the bound in the theorem above (see [Pet18] for more details).

## 3.2 The kissing number

There are variants of the method behind our bounds on the systole with Maxime Fortier Bourque (see Section 3.1.1) that can be used to prove bounds on other geometric and spectral invariants of hyperbolic manifolds as well. The idea is always the same: we use the Selberg trace formula with a function that satisfies certain linear conditions to find a bound on the quantity that interests us. Afterwards we optimize the bound on the set of functions which satisfy the conditions.

The second invariant for which we found a new bound is the **kissing number** of a hyperbolic surface: the number of oriented geodesics that realize the systole of the surface. This terminology was invented by Schmutz-Schaller [Sch96a, Sch96b, SS97] in analogy with the number of contacts of a Euclidean lattice. The basic question is:

Question 5

Given 
$$d \ge 2$$
 and  $v > 0$ , what is  

$$\max \left\{ \begin{aligned} \operatorname{Kiss}(M); & M \text{ a hyperbolic } d\text{-manifold} \\ & \text{of volume } \leqslant v \end{aligned} \right\} ?$$
And which hyperbolic manifold realizes it?

The bound that we find is [BP23c]:

**Theorem 3.7.** There exists a  $g_0 \ge 2$  such that

$$Kiss(X) < \frac{4.873 \cdot g^2}{\log(g) + 1.2045}$$

for every closed oriented hyperbolic surface X of genus  $g \ge g_0$ .

This improves a bound of Parlier [Par13], which is also of the form  $g^2/\log(g)$ , but with a multiplicative constant of 200. Our method is completely different: Parlier's proof is a geometric argument and does not use the Selberg trace formula.

The best known examples are again given by principal congruence coverings of arithmetic surfaces: Schmutz-Schaller [SS97] proved that there exist sequences of congruence coverings  $(X_k)_k$  of genus  $g_k$  of an arithmetic surface compact such that  $g_k \to \infty$  and for all



Figure 3.5: The bound on the kissing number in genus 2-20.

 $\varepsilon > 0$ , the kissing number satisfies

$$\operatorname{Kiss}(X_k) \gg g_k^{4/3-\varepsilon} \quad \text{as } k \to \infty$$

Schmutz-Schaller [SS97, SS98] conjectured that the exponent  $\frac{4}{3}$  is optimal. In higher dimension, constructions of hyperbolic manifolds whose kissing number grows faster than linearly as a function of their volume were found by Dória-Murillo and Dória-Freire-Murillo [DM21, DFM23]. With Maxime Fortier Bourque, we have also proved an analogous bound to Theorem 3.7 for higher dimensional closed hyperbolic manifolds [BP22].

Like for the systole, our method can also be used to prove bounds for surfaces of small area. The result can be found in Figure 3.5. The bound that we obtain is the best known bound, except in the case of genus 2, where the maximum possible kissing number is known: it is 24 [Sch94] and it is achieved by the Bolza surface [Jen81].

## 3.3 The first eigenvalue

The other three invariants for which we found bounds with Maxime Fortier Bourque are spectral invariants. We consider the Laplacian

$$\Delta_X := -\operatorname{div} \circ \operatorname{grad} : C^{\infty}(X) \to C^{\infty}(X)$$

acting on functions on a closed orientable hyperbolic surface X. The eigenvalues of this operator (the same eigenvalues that appear in the Selberg trace formula above) form a discrete spectrum in  $[0, \infty)$ .

These eigenvalues contain a lot of geometric information about the underlying surface. For example, a classical theorem due to Huber [Hub59, Hub60, Hub61] states that the spectra of the Laplacians of two closed hyperbolic surfaces are the same if and only their length spectra – the multisets containing the lengths of all closed geodesics – are. moreover, the first eigenvalue  $\lambda_1(X)$  can be seen as a measure of connectivity of the surface, for instance because of the Cheeger–Buser inequality [Che70, Bus82], that we will discuss in detail in Section 6.3, and also because  $\lambda_1(X)$  influences the mixing rate of the geodesic flow of the surface: the larger it is, the faster the flow mixes (see for instance [Mat13] for an effective version).

It has been known for a long time, due to results by Huber and Cheng [Hub74, Che75b] that, if the genus of a hyperbolic surface X is large, then  $\lambda_1(X)$  can't be much larger than  $\frac{1}{4}$ . The significance of  $\frac{1}{4}$  here is that it's the spectral gap of the hyperbolic plane. This is the hyperbolic analogue of the Alon-Boppana bound mentioned in Section 2.2.3. The best upper bound that was known before our work with Maxime Fortier Bourque combines the result of Cheng with work of Gage [Gag80] and Bavard [Bav96] and states:

$$\lambda_1(X) \leq \frac{1}{4} + \left(\frac{2\pi}{\log(g-1)}\right)^2.$$

With Maxime Fortier Bourque, we proved:

**Theorem 3.8.** There exists  $g_0 \ge 2$  such that

$$\lambda_1(X) < \frac{1}{4} + \left(\frac{\pi}{\log(g) + 0.7436}\right)^2.$$

for every hyperbolic surface X of genus  $g \ge g_0$ .



Figure 3.6: The bound on the first eigenvalue in genus 2–20.

Our method improves the error term by a factor of 4. This method is different from that of Cheng and moreover, thanks to the existence of hyperbolic surfaces of small diameter (see Section 6.2), it is not possible to obtain our bound with Cheng's method.

The bounds that we obtain in small genus can be found in Figure 3.6. There has been a lot of recent progress the maximum of the first eigenvalue. The best known upper bounds in genus 2 and 3 are due to work by Bonifacio [Bon22] and Kravchuk–Mazac–Pal [KMP21] and were found using a method called *conformal bootstrap*. These bounds are very close to the first eigenvalues of the Bolza surface and the Klein quartic respectively, so it's natural to conjecture that these two surfaces maximize the first eigenvalue in their moduli spaces. Somewhat curiously, the conformal bootstrap yields a decidedly sharper bound than ours in genus 2 and 3, yet our bound is sharper in higher genus. In neither case, this seems to be due to not running the numerical optimizer long enough. Finally, in genus 4 and 6, a classical bound by Yang–Yau [YY80] is still the best known bound.

The question of whether there exist sequences of closed surfaces whose genus tends to infinity and whose first eigenvalue tends to  $\frac{1}{4}$  has been open for a long time. Recently, there has been a spectacular breakthrough due to Hide–Magee [HM23b, Hid23]: such a sequence exists. Hide–Magee construct the sequence as compactifications of random covers of a sphere with three cusps. Their proof makes essential use of the work of Bordenave– Collins [BC19, BC23] on the asymptotic freedom of random permutations and the related question of spectral gaps in random coverings of graphs. However, the following question remains open:

#### Question 6

Does there exist a hyperbolic surface  $X \in \mathcal{M}_g$  such that

$$\lambda_1(X) \ge \frac{1}{4}$$
 or even  $\lambda_1(X) > \frac{1}{4}$ 

for every  $g \ge 2$ ? Such surfaces are sometimes called Ramanujan (or strict Ramanujan) surfaces, in analogy with Ramanujan graphs.

Selberg's conjecture [Sel56] and its generalizations [BLS92] assert that congruence lattices give examples of Ramanujan surfaces (and higher-dimensional Ramanujan manifolds). The best known bounds towards these conjectures are due to Kim–Sarnak [Kim03] in dimension 2 and to Bergeron–Clozel [BC13] for higher dimensional hyperbolic manifolds.

# 3.4 The multiplicity of the first eigenvalue

Another invariant that we can access with our methods is the multiplicity  $m_1(X)$  of the first eigenvalue: the dimension of the eigenspace associated to  $\lambda_1(X)$ .

The study of this invariant has a long history as well. For the case of surfaces, Colinde-Verdière (see [CdV86, p.269] and [CdV87, p.601]) has conjectured that, if  $\Sigma$  is a closed surface,

 $\max \{m_1(\Sigma, g); g \text{ a Riemannian metric on } \Sigma\} = \operatorname{chr}(\Sigma) - 1,$ 

where  $chr(\Sigma)$  is the chromatic number of  $\Sigma$ : the maximal possible number of vertices of a complete graph that embeds in  $\Sigma$ . A theorem by Ringel–Youngs [RY68] states that (if  $\Sigma$  is not the Klein bottle):

$$\operatorname{chr}(\Sigma) = \left\lfloor \frac{1}{2} \left( 7 + \sqrt{49 - 24\chi(\Sigma)} \right) \right\rfloor,\,$$

where  $\chi(\Sigma)$  denotes the Euler characteristic of  $\Sigma$ . Colin-de-Verdière also proved that in higher dimension, the situation is very different: on any closed manifold of dimension at least 3 and for any  $N \ge 1$ , one can find a Riemannian metric such that the multiplicity of its first eigenvalue is N [CdV87]. Going back to the case of surfaces, Colin-de-Verdière's conjecture has been proven for the sphere by Cheng [Che76], for the torus and the projective plane by Besson [Bes80] and for the Klein bottle by Colin-de-Verdière and Nadirashvili [CdV87, Nad87]. Colbois– Colin-de-Verdière [CCdV88] have found examples of sequences of closed hyperbolic surfaces whose first eigenvalue multiplicity grows like the square root of their genus, but at a slower rate than predicted by Colin-de-Verdière's conjecture. The best known upper bound on the multiplicity of the first eigenvalue among all Riemannian metrics is due to Sévennec [Sév02], who uses a topological argument based on Courant's nodal domain theorem [Cou23] and topological methods "à la Borsuk–Ulam" to prove that the supremum above can be bounded by  $5 - \chi(\Sigma)$ , which comes down to 2g + 3 for a closed orientable surface of genus g.

With our methods we obtain a slight improvement of Sévennec's result in the case of hyperbolic surfaces:

**Theorem 3.9.** There exists a  $g_0 \ge 2$  such that

$$m_1(X) \leqslant 2g - 1.$$

for every closed hyperbolic surface X of genus  $g \ge g_0$ .

In fact, our method works less well for multiplicity than for the other invariants. The problem is that with the Selberg trace formula, we are only able to prove bounds on the number of eigenvalues in certain intervals. On the other hand, it is known that in any moduli space  $\mathcal{M}_g$  and for any  $\varepsilon > 0$ , we can find hyperbolic surfaces with as many eigenvalues in the interval  $(\frac{1}{4}, \frac{1}{4} + \varepsilon)$  as we want [Bus10, p.219]. In other words, our bound on the multiplicity necessarily depends on  $\lambda_1(X)$  and it explodes when  $\lambda_1(X) \to \frac{1}{4}$ . Instead of the trace formula, we use a bound by Otal [Ota08] for surfaces X such  $\lambda_1(X) \leq \frac{1}{4}$ , which states that in this case,  $m_1(X) \leq 2g - 3$ . Afterwards, we use the topological methods of Otal and Sévennec to prove that there exists an explicit  $a_g > 0$  such that,  $m_1(X) \leq 2g - 1$ , if  $\lambda_1(X) \in (\frac{1}{4}, \frac{1}{4} + a_g]$ . For the remaining interval (between  $\frac{1}{4} + a_g$  and the upper bound in Theorem 3.8), we use the trace formula. On this small interval, the bound we find is in fact sublinear as a function of g.

A general sublinear bound, under the condition that the systole of the surface is not too small, has recently been found by Letrouit–Machado [LM23]. Their proof is strongly inspired by that of the analogous result for regular graphs  $[JTY^+21]$ .

The bounds that we found in genus 2–20 can be found in Figure 3.7. Our bound is sharp in genus 3. In [BP23b] we proved:



Figure 3.7: The bound on the multiplicity of the first eigenvalue in genus 2–20.

Theorem 3.10. We have:

$$\max\left\{m_1(X); \ X \in \mathcal{M}_3\right\} = 8$$

and this is realized by the Klein quartic.

The quickest way to define the Klein quartic is to say that it's the unique closed orientable hyperbolic surface of genus 3 whose group of orientation preserving isometries has order 168 (which is maximal in genus 3, by the Hurwitz bound). For more information on the Klein quartic, we refer the reader to [Lev99], a whole book dedicated to this surface.

To prove the fact that the first eigenvalue multiplicity of the Klein quartic is 8, we use the fact that the Klein quartic has so many symmetries. This is a combination of our methods based on the trace formula and the approach that Jenni [Jen81] and Cook [Coo18] used to prove that the multiplicity of the first eigenvalue multiplicity of the Bolza surface and the Klein quartic are 3 and at least 6 respectively.

Concretely, the group of isometries of the Klein quartic is  $PGL(2, \mathbb{Z}/7\mathbb{Z})$  (and the subgroup of orientation preserving isometries is  $PSL(2, \mathbb{Z}/7\mathbb{Z})$ ) and the action of this group commutes with the Laplacian. This implies that the eigenspaces of the Laplacian decompose into representations of  $PGL(2, \mathbb{Z}/7\mathbb{Z})$ . This group has two irreducible representations of dimension 1, three of dimension 6, two of dimension 7 and two of dimension 8. The goal is therefore on the one hand to rule out representations of dimensions lower than 8 and on the other hand to prove that the multiplicity is at most 8. We first prove, using the spectrum of the quotient of the Klein quartic by  $PGL(2, \mathbb{Z}/7\mathbb{Z})$ , that representations of dimension 1 do not appear in the first eigenspace. Then we prove, with the trace formula, that in a certain interval which contains the first eigenvalue of the Klein quartic, there are more than 7 eigenvalues and less than 12. The only positive integer combination of 6, 7 and 8 between 7 and 12 being 8, the multiplicity of the first eigenvalue of the Klein quartic has to be 8.

What is striking is that our method works for genus 3, but not for genus 2. The problem is that the upper bound we prove (6) is larger than the largest known multiplicity (3, for the Bolza surface). Nor do we know whether the Klein quartic is the unique surface of genus 3 of multiplicity 8.

### Question 7

• Is it true that

 $\max\left\{m_1(X); X \in \mathcal{M}_2\right\} = 3?$ 

And if so, is the Bolza surface the unique closed orientable surface of genus 2 and first eigenvalue multiplicity 3?

• Is the Klein quartic the unique closed orientable surface of genus 3 and first eigenvalue multiplicity 8?

In upcoming work with Maxime Fortier Bourque, Émile Gruda-Médiavilla and Mathieu Pineauilt, we will prove that Colin-de-Verdière's conjecture is false: there are two counterexamples, one of genus 10 and one of genus 17. The graph in Figure 3.7 will soon get an update as well.

## 3.5 The number of small eigenvalues

The last invariant that we studied with Maxime Fortier Bourque is the number of small eigenvalues:

$$N_{\text{small}}(X) = \max\left\{k; \, \lambda_k(X) \leq \frac{1}{4}\right\} + 1,$$

where we've added 1 because we're calling the trivial eigenvalue  $\lambda_0(X)$ . This invariant is of interest for example because small eigenvalues give rise to exponential error terms in the asymptotic count of the number of short geodesics on X (see for example [Bus10, Section 9.6]).

Buser [Bus77] and Otal–Rosas [OR09] proved that:

$$\max \{ N_{\text{small}}(X); X \in \mathcal{M}_g \} = 2g - 2 \quad \text{pour tout } g \ge 2.$$

The lower bound proved by Buser uses explicit examples of hyperbolic surfaces. The systoles of these examples are very small, so it is a natural question to ask whether this is necessary. The first result which implies that this is indeed the case is due to Huber [Hub76] and says that

$$N_{\text{small}}(X) < \min\left\{\frac{3\pi^2(g-1)}{8 \;(\log(\cosh(\text{sys}(X)/4)))^3}, \frac{g-1}{2 \;\log(\cosh(\text{sys}(X)/4))}\right\}.$$

Huber's method is similar to ours, except that he chooses another type of special functions: Huber uses Legendre functions and we use functions constructed with Bessel functions. With our functions, we obtain a small improvement on Huber's result:

**Theorem 3.11.** If X is a closed oriented hyperbolic surface of genus  $g \ge 2$ , then

$$N_{small}(X) < \min\left(\frac{24\pi^2(g-1)}{sys(X)^3}, \frac{16(g-1)}{sys(X)^2}\right).$$

For the proof that our bound is really strictly smaller than (but still asymptotic to) Huber's bound, we refer to [BP23c, Section 10.2].

Like for the other invariants, in small genus, numerical optimization gives sharper results than the test functions we constructed by hand (see Figure 3.8). For example, we prove that  $N_{\text{small}}(X) < 2g - 2$  as soon as

$$\operatorname{sys}(X) \ge 2.317.$$



Figure 3.8: The bound on  $\mathrm{N}_\mathrm{small}/(g-1)$  when the systole is small.

This improves on Huber's earlier mentioned results that imply that  $N_{\text{small}}(X) < 2g - 2$ when  $\text{sys}(X) > 4 \operatorname{arccosh}(e^{1/4}) = 2.9476...$  (and Jammes's 3.46 that was found using different methods [Jam21]).

# 4 Counting

A large part of this thesis will be taken up by random manifolds. Before we get to these, the very first question – at least for discrete models of random manifolds – is that of counting: in order to calculate probabilities with respect to the uniform measure on a finite set, one needs to know the number of elements in the set (or at least have a good approximation). Often, this question is already very hard.

A classical example is counting triangulations of manifolds. In the 1960s, Tutte [Tut63] developed a way to enumerate triangulations of 2-dimensional disks and since then, there has been an immense amount of work on counting triangulations of surfaces (we will discuss a very small portion of it in the next chapter). Even if a lot is known, many very basic questions remain open. In dimension 3 and above, we know much less, even the asymptotic number of triangulations of the sphere is out of reach of current methods: the best known bounds are very far apart (see for instance [BZ11, DJ95, AB20, CP21]).

In this chapter we will discuss our work on subgroup growth with Hyungryul Baik and Jean Raimbault [BPR20, BPR19] and with Stefan Friedl, JungHwan Park, Jean Raimbault and Arunima Ray [FPP+21] and our paper with Patricia Cahn and Federica Fanoni on counting mapping class group orbits of curves on surfaces [CFP18],

# 4.1 Subgroup growth

In this section, we will discuss subgroup growth: the study of the number of finite index subgroups of a given group. Concretely, given a finitely generated group  $\Gamma$ , we define

$$s_n(\Gamma) = |\{H < \Gamma; [\Gamma : H] = n\}|.$$

For a finitely generated group, this quantity is always finite. The basic question now is how  $s_n(\Gamma)$  depends on n and which algebraic and geometric properties of  $\Gamma$  one can deduce from the information contained in this sequence. Before we'll get to our joint work with Hyungryul Baik and Jean Raimbault on the subgroup growth of Artin and Coxeter groups, we will start with some context. We will only give a quick introduction to the subject of subgroup growth and we make a selection which is very biased by applications to random manifolds. For a more complete introduction we refer to the book by Lubotzky–Segal [LS03].

Subgroup growth is relevant for the study of random manifolds, because a random finite index subgroup of the fundamental group  $\pi_1(M)$  of a manifold M gives rise to a random finite degree cover of M. This model has recently been very successful in dimension 2 [MP23, MNP22, HM23b, Nau22] (see Section 5.1.2) and we'll also treat a 3-dimensional example in Section 5.2.2.

### 4.1.1 Invariant random subgroups

One way to look at these random finite index subgroups is through the lens of invariant random subgroups. For a finitely generated group  $\Gamma$ ,  $\operatorname{Sub}(\Gamma)$  will denote the **Chabauty** space of subgroups of  $\Gamma$  (see for instance [Gel18] for an introduction).

We will be interested in random index n subgroups of such a group  $\Gamma$ . Given  $n \in \mathbb{N}$ , we will write

$$\mathcal{S}_n(\Gamma) = \{H < \Gamma; [\Gamma : H] = n\} \subset \operatorname{Sub}(\Gamma),$$

so that  $s_n(\Gamma) = |\mathcal{S}_n(\Gamma)|$ . Studying a random index *n* subgroup of  $\Gamma$  comes down to understanding the measure  $\mu_n$  on Sub( $\Gamma$ ), defined by

$$\mu_n = \frac{1}{s_n(\Gamma)} \sum_{H \in \mathcal{S}_n(\Gamma)} \delta_H$$

where  $\delta_H$  denotes the Dirac mass on  $H \in \text{Sub}(\Gamma)$ .

 $\mu_n$  is an example of what is called an **invariant random subgroup** (IRS) of  $\Gamma$  – i.e. a Borel probability measure on Sub( $\Gamma$ ) that is invariant under conjugation by  $\Gamma$ . We will write IRS( $\Gamma$ ) for the space of IRS's of  $\Gamma$  endowed with the weak-\* topology. This space has been first studied under this name in [AGV14] and [Bow14] and under a different name in [Ver12].

### 4.1.2 Estimating the subgroup growth of a group

Most estimates  $s_n(\Gamma)$  start with the following lemma. In this lemma,  $\mathfrak{S}_n$  denotes the symmetric group on n letters.

**Lemma 4.1.** Let  $\Gamma$  be a group. Then

$$s_n(\Gamma) = \frac{t_n(\Gamma)}{(n-1)!},$$

where

 $t_n(\Gamma) = |\{\varphi \in \operatorname{Hom}(\Gamma, \mathfrak{S}_n); \varphi(\Gamma) \text{ acts transitively on } \{1, \ldots, n\}\}|$ 

*Proof.* If  $H < \Gamma$  is such that  $[\Gamma : H] = n$ , then  $\Gamma$  acts transitively on the set of cosets

$$\Gamma/H = \{H, g_2H, \dots, g_nH\}$$

The induced permutation action on the indices of the  $g_i$  is an action on a set of n elements. In doing this, we have made an arbitrary identification of the last n - 1 cosets. So, in reality, H gives rise to (n - 1)! distinct actions.

In the other direction, given a transitive action of  $\Gamma$  on  $\{1, \ldots, n\}$ , the stabilizer  $\operatorname{Stab}_{\Gamma}(\{1\}) < \Gamma$  is a subgroup of index n. Moreover, if we conjugate the action by an element of  $\mathfrak{S}_{n-1}$  acting on  $\{2, \ldots, n\}$  this doesn't affect the stabilizer.  $\Box$ 

For an infinite group  $\Gamma$ , it is rare that one can calculate an explicit formula for  $s_n(\Gamma)$ , and even if this is possible, the formula is often so complicated that it is difficult to extract information from it. So typically, we rather look for the asymptotic behavior of  $s_n(\Gamma)$  as  $n \to \infty$ . The first example of a class of infinite finitely generated groups for which such an asymptotic expression has been calculated is the class of free groups. First, we observe that the question is trivial for  $\mathbb{Z}$ :  $s_n(\mathbb{Z}) = 1$  for all  $n \in \mathbb{N}$ . For non-abelian free groups, a result by Dixon [Dix69] implies that, if  $r \ge 2$  the probability that a homomorphism  $\varphi \in \operatorname{Hom}(F_r, \mathfrak{S}_n)$  is transitive tends towards 1 when  $n \to \infty$ . Here  $F_r$  denotes the free group of rank r. Combined with Lemma 4.1, this implies that

$$s_n(F_r) \sim n \cdot (n!)^{r-1}$$
 as  $n \to \infty$ 

for all  $r \ge 2$ . This result was generalized to lattices in PSL(2,  $\mathbb{R}$ ) by Müller–Schlage-Puchta [MP02, MSP07] and Liebeck–Shalev [LS04a]. Like in the case of free groups, the factorial

growth of  $s_n(\Gamma)$  is linked to the area of the associated hyperbolic surface, i.e.

$$\lim_{n \to \infty} \frac{\log(s_n(\Gamma))}{n \log(n)} = \frac{\operatorname{area}(\Gamma \setminus \mathbb{H}^2)}{2\pi} = -\chi(\Gamma)$$

where  $\chi(\Gamma)$  denotes the Euler characteristic of  $\Gamma$ . These results form the starting point for earlier mentioned results on the geometry and spectra of random covers of hyperbolic surfaces.

For lattices in  $\operatorname{Isom}(\mathbb{H}^d)$  for  $d \ge 3$ , a case that is of obvious interest to us, much less is known. There is no such lattice for which we know an asymptotic equivalent for the sequence  $(s_n(\Gamma))_n$ . The fact that the lattices in  $\operatorname{Isom}(\mathbb{H}^d)$  are large – they have finite index subgroups which surject onto  $F_2$  –, which was proven by Agol [Ago13], using work by Kahn–Markovic [KM12], Wise [Wis21], Bergeron–Wise [BW12] and many others, does imply that regularized counting function

$$s_{\leq n}(\Gamma) = \sum_{k=1}^{n} s_k(\Gamma)$$

grows factorially fast as a function of n. On the other hand, Agol's theorem is not effective, so we currently have no good estimates on  $\lim_{n\to\infty} \frac{\log(s_n(\Gamma))}{n\log(n)}$  in general. There is one example of a lattice in  $\operatorname{Isom}(\mathbb{H}^3)$  for which we can determine this factorial growth rate. We will discuss this in Section 4.1.4.

Also the following question is open:

Question 8
Let $d \ge 3$ and let $\Gamma < \text{Isom}(\mathbb{H}^d)$ be a lattice. Can one deduce the volume $\text{vol}(\Gamma \setminus \mathbb{H}^d)$
from the sequence $(s_n(\Gamma))_n$ ?

If the answer to this question is "yes", then the relation is more complicated than in dimension 2: using the fact that there exist closed hyperbolic manifolds of dimension 3 of arbitrarily large volume but whose fundamental group can be generated with 5 elements (i.e. there is an epimorphism of  $F_5$  onto these fundamental groups), we can prove that there exist hyperbolic manifolds of arbitrarily large volume whose factorial subgroup growth rate is bounded by 4. Indeed, if we have an epimorphism  $\Lambda \to \Gamma$ , then  $s_n(\Gamma) \leq s_n(\Lambda)$  for all n. To find examples of hyperbolic manifolds with the desired property, we can for example take a sequence of cyclic coverings of a hyperbolic manifold which fibers on the circle.

A second observation is that the sequence  $(s_n(\Gamma))_n$  is a profinite invariant of  $\Gamma$ . In other

words, if two groups  $\Gamma$  and  $\Lambda$  have the same profinite completion (see for example [RZ10]), then  $s_n(\Gamma) = s_n(\Lambda)$  for all  $n \in \mathbb{N}$ . In particular, if the answer is "yes", this also gives a positive answer to the open question of whether the volume of a hyperbolic manifold is a profinite invariant. This question has for instance been raised in [BF20, Question 3.18] and [Liu23] and also connects to more general questions on profinite rigidity of 3-manifold groups, see [Rei15, Rei18] for an introduction and [BMRS20] for examples of profinitely rigid Kleinian groups.

Finally, we note that the situation for lattices in Lie groups of higher rank, in which case arithmetic methods can be used, is wildly different from that in rank one (see for instance [Lub95, LN04, GLP04]).

### 4.1.3 Right-angled Artin groups

Together with Hyungryul Baik and Jean Raimbault [BPR20, BPR19], we've studied the subgroup growth of right-angled Artin and Coxeter groups. We'll start with the former. Given a finite graph G = (V, E), the associated right-angled Artin group is

$$\Gamma_{\mathcal{A}}(G) = \langle v \in V | [v, w] \text{ for all } \{v, w\} \in E \rangle.$$

These groups interpolate between free groups and free abelian groups. Indeed, if G is a complete graph, then  $\Gamma_A(G)$  is a free abelian group and if G has no edges, then  $\Gamma_A(G)$  is a free group. They show up in hyperbolic geometry for instance because Agol proved that every cocompact lattice in PSL(2,  $\mathbb{C}$ ) virtually embeds in one of them [Ago13].

For us, they're a good testing ground because they are large groups, but are given by relatively manageable relations. Moreover, they are also a first step towards right-angled Coxeter groups, that appear as lattices in the isometry groups of some hyperbolic spaces.

What we could access with Hyungryul Baik and Jean Raimbault is the factorial subgroup growth rate of these groups:

**Theorem 4.2.** let G = (V, E) be a finite graph, then

$$\lim_{n \to \infty} \frac{\log(s_n(\Gamma_{\mathcal{A}}(G)))}{n \log(n)} = \alpha(G) - 1,$$

where  $\alpha(G)$  denotes the independence number of G: the maximal size of a subset  $W \subset V$  such that there are no edges between any of the pairs of vertices in W.

The independence number  $\alpha(G)$  also has a geometric interpretation. Indeed,  $\Gamma_A(G)$  is the fundamental group of a cube complex called the Salvetti complex. The number  $\alpha(G)$  counts the maximal number of disjoint hyperplanes in this complex (see for instance [Vog15, Example 5.2]).

The proofs of the upper and lower bounds in the theorem above are distinct. In fact, the lower bound follows from the results on free groups mentioned above. Indeed, if G = (V, E) is a finite graph and  $W \subset V$  and independent set – a set of vertices that have no edges between them – then we obtain a surjection

$$\Gamma_{\mathcal{A}}(G) \longrightarrow F_{|W|},$$

by mapping the generators of  $\Gamma_{\mathcal{A}}(G)$  corresponding to the elements of W to distinct generators of the free group  $F_{|W|}$ . This implies that  $s_n(\Gamma_{\mathcal{A}}(G)) \ge s_n(F_{|W|})$  for all  $n \in \mathbb{N}$  and thus proves the lower bound.

The bulk of the proof of the theorem goes into showing that, at the factorial level, this simple lower bound is sharp. Morally, this comes from the fact that typically the centralizers of elements in a symmetric group are very small compared to the size of the symmetric group itself. Our proof uses an induction on the number of vertices of G and yields a bound of the form

$$s_n(\Gamma_{\mathcal{A}}(G)) \leq C^{n\log\log(n)}(n!)^{\alpha(G)-1}$$
 for all  $n \in \mathbb{N}$ ,

with a constant C > 0 that depends on the graph G. For details, we refer to [BPR20]. Here, we will include the proof in a special case, which is much quicker, leads to a sharper upper bound, and uses a very nice classical theorem from graph theory, due to Kőnig. It is however not very representative of the general case.

Proof of Theorem 4.2 in the case of bipartite graphs. We assume that G = (V, E) is bipartite. That is, we assume that we can write  $V = V_1 \sqcup V_2$  and there are no edges interior to  $V_1$  or  $V_2$ .

We will need two more notions from graph theory. First of all, a **matching** of G is a set  $M \subset E$  of edges such that no pair of edges in M shares an endpoint. A matching is called **maximal** if it has the maximal number of edges among all matchings in the graph. We will write  $\mu(G)$  for the number of edges in such a maximal matching.

Secondly, a **vertex cover** of G is a set of vertices  $W \subset V$  such that W contains at least

one of the endpoints of every edge in E. A vertex cover is called **minimal** if it contains the minimum number of vertices among all vertex covers of the graph. We will write  $\nu(G)$ for the number of vertices in a minimal vertex cover of G.

Kőnig's theorem states that, if G is bipartite:

$$\mu(G) = \nu(G).$$

Moreover, observe that for any graph G,

$$\nu(G) + \alpha(G) = |V|,$$

because every vertex cover is complementary to an independent set and vice versa. So, Kőnig's theorem implies that, when G is bipartite, we can find a matching  $M \subset E$  that contains

$$|V| - \alpha(G)$$

edges. Such a matching M is incident to  $2(|V| - \alpha(G))$  vertices and hence there are  $2\alpha(G) - |V|$  vertices not incident to M. This means that there is a surjection

$$(\mathbb{Z}^2)^{*(|V|-\alpha(G))} * F_{2\alpha(G)-|V|} \longrightarrow \Gamma_{\mathcal{A}}(G),$$

where "\*" denotes the free product. Writing  $h_n(\Gamma) = |\text{Hom}(\Gamma, \mathfrak{S}_n)|$  for any group  $\Gamma$ , the surjection above implies that

$$h_n(\Gamma_{\mathcal{A}}(G)) \leq h_n\left((\mathbb{Z}^2)^{*(|V|-\alpha(G))} * F_{2\alpha(G)-|V|}\right) = p(n)^{|V|-\alpha(G)} \cdot (n!)^{\alpha(G)},$$

where p(n) denotes the number of partitions of the natural number n. Together with Lemma 4.1, this implies the theorem in the case of bipartite graphs.

### 4.1.4 Right-angled Coxeter groups

The case of right-angled Coxeter groups is already a lot more subtle than that of rightangled Artin groups. First, we remind the reader that the only thing that changes is that we now assume the generators to be involutions. That is, given a finite graph G = (V, E), the associated right-angled Coxeter group is

$$\Gamma_{\mathcal{C}}(G) = \langle v \in V | v^2 \text{ for all } v \in V \text{ and } [v, w] \text{ for all } \{v, w\} \in E \rangle.$$

An example from hyperbolic geometry of such a group is the group generated by the reflections in the faces of a regular ideal right-angled octahedron<sup>1</sup> in  $\mathbb{H}^3$ . The graph defining this group is the 1-skeleton of the cube. In general, it is known that right-angled Coxeter groups can appear as lattices in Isom( $\mathbb{H}^d$ ) only if  $d \leq 14$ , examples of such lattices are known in dimensions 2–8 [PV05].

Just like in the case of right-angled Artin groups, right-angled Coxeter groups admit surjections on groups that are easier to understand. We could take independent sets of vertices of G again and obtain surjections on free products of multiple copies of  $\mathbb{Z}/2\mathbb{Z}$ . It however turns out we can do better. To this end, we remind the reader that a **clique** in G = (V, E) is a complete subgraph of G. Given a clique C in G, we define its weight as

$$w(C) = 1 - 2^{|C|},$$

where |C| denotes the number of vertices in C. A system C of independent cliques in G is a collection of disjoint cliques in G such that G contains no edges between any pair of these cliques. The total weight of such a system is

$$w(\mathcal{C}) = \sum_{C \in \mathcal{C}} w(C).$$

We now define the following graph invariant

 $\gamma(G) = \max \{ w(\mathcal{C}); \mathcal{C} \text{ a system of independent cliques in } G \}.$ 

This invariant plays the same role as the independence number did for right-angled Artin groups. Indeed, given a system  $\mathcal{C}$  of independent cliques in G, we obtain a surjection

$$\Gamma_{\mathcal{C}}(G) \longrightarrow \underset{C \in \mathcal{C}}{*} (\mathbb{Z}/2\mathbb{Z})^{|C|}$$

It follows from work by Müller [Mül97, Mül96] that

$$\liminf_{n \to \infty} \frac{\log \left( s_n(\Gamma_{\mathcal{C}}(G)) \right)}{n \log(n)} \ge \gamma(G) - 1.$$

The crux here is that the exponents of the copies of  $\mathbb{Z}/2\mathbb{Z}$  in the free product above have an influence on the factorial growth rate. This is why cliques, instead of just isolated vertices, show up. From a combinatorial point of view, this also makes the problem a lot

<sup>&</sup>lt;sup>1</sup>It turns out that this list of adjectives is long enough to determine a unique polytope up to isometry.

less tractable. Nonetheless, with Hyungryul Baik and Jean Raimbault, we conjecture that, at the factorial level, the lower bound we obtain is still sharp:

Conjecture				
Let $G = (V, E)$ be a finite non-complete graph, then				
$\lim_{n \to \infty} \frac{\log \left( s_n(\Gamma_{\mathcal{C}}(G)) \right)}{n \log(n)} = \gamma(G) - 1.$				

The reason we exclude complete graphs is that the corresponding groups are finite.

One of our reasons for conjecturing the above is that we proved it holds for a large class of groups:

**Theorem 4.3.** Let G = (V, E) be a finite graph for which there exists a (possibly empty) set of vertices  $\{v_1, \ldots, v_k\} \subset V$  such that

- The 1-neighborhood of  $v_i$  is a tree for all i = 1, ..., k,
- these 1-neighborhoods are disjoint
- and the graph

$$G - \{v_1, \ldots, v_k\}$$

is a tree.

Then the conjecture above holds for  $\Gamma_{\rm C}(G)$ .

We don't have a heuristic reason for the conditions above, they are the conditions that come out of our proof. We can actually push the methods slightly further and obtain a larger class of graphs. This class is defined properly in [BPR20]. Since we're not expecting that class to be optimal either, we've chosen to slightly weaken the theorem here for ease of exposition. Furthermore, there are again classes of graphs, trees for instance, for which the proof simplifies considerably. We include one example here, the only lattice in  $Isom(\mathbb{H}^3)$ for which we know the factorial subgroup growth rate

**Proposition 4.4.** Let G be the 1-skeleton of the cube, so that  $\Gamma_{O} := \Gamma_{C}(G)$  is the reflection group of the regular right-angled octahedron in  $\mathbb{H}^{3}$ . Then the conjecture above is true for  $\Gamma_{O}$ , that is

$$\lim_{n \to \infty} \frac{\log(s_n(\Gamma_O))}{n \log(n)} = \gamma(G) - 1 = 1.$$

*Proof.* Given our observations before, we only need to prove the upper bound. Writing S for the 1-skeleton of the square, we have a surjection

$$\Gamma_{\mathcal{C}}(S) * \Gamma_{\mathcal{C}}(S) \longrightarrow \Gamma_{\mathcal{C}}(G).$$

Indeed, we can decompose the vertex set V of G into two sets of 4 vertices, each corresponding to a face of the cube. Since the right-angled Coxeter group corresponding to a disjoint union of two graphs is the free product of the two corresponding right-angled Coxeter groups, the group corresponding to the union of the two squares on these two sets 4 of vertices is  $\Gamma_{\rm C}(S) * \Gamma_{\rm C}(S)$ . The map  $\Gamma_{\rm C}(S) * \Gamma_{\rm C}(S) \longrightarrow \Gamma_{\rm C}(G)$  induced by sending the generators corresponding to the vertices to themselves is a well-defined surjection.

Now,  $\Gamma_{\rm C}(S) = D_{\infty} \times D_{\infty}$ , where  $D_{\infty} \simeq \mathbb{Z}/2\mathbb{Z} * \mathbb{Z}/2\mathbb{Z}$  is the infinite dihedral group. In particular,  $\Gamma_{\rm C}(S)$  is virtually abelian. As such, it satisfies  $h_n(\Gamma_{\rm C}(S)) \leq C^n n!$  for some C > 0 (see for instance [LS03, Chapter 1]. This, combined with Lemma 4.1, implies that

$$s_n(\Gamma_{\mathcal{O}}) \leqslant \frac{h_n(\Gamma_{\mathcal{O}})}{(n-1)!} \leqslant \frac{h_n(\Gamma_{\mathcal{C}}(S) \ast \Gamma_{\mathcal{C}}(S))}{(n-1)!} = \frac{h_n(\Gamma_{\mathcal{C}}(S)) \cdot h_n(\Gamma_{\mathcal{C}}(S))}{(n-1)!} \leqslant n \cdot C^{2n} \cdot n!,$$

which proves our claim.

For more details on Theorem 4.3 and its proof, which in part relies on the results on Fuchsian groups we mentioned in the previous section, we refer to [BPR20].

Finally, there is one very specific class of right-angled Coxeter groups for which we obtain much better asymptotic results [BPR19]. These are virtually cyclic right-angled Coxeter groups and their free products. It turns out that every virtually cyclic right-angled Coxeter group is of the form

$$\Gamma_r := (\mathbb{Z}/2\mathbb{Z} * \mathbb{Z}/2\mathbb{Z}) \times (\mathbb{Z}/2\mathbb{Z})^r$$

for some  $r \ge 0$ .

The main advantage of these groups is that we can write down explicit generating functions for the sequences that interest us. Like in some of the proofs above, we start with the sequence

$$h_n(\Gamma_r) = |\operatorname{Hom}(\Gamma_r, \mathfrak{S}_n)|, \quad n \in \mathbb{N},$$

which is in fact equivalent information to the sequence  $s_n(\Gamma_r)$  (see [LS03, Corollary 1.1.4]).

#### 4.1. SUBGROUP GROWTH

Let us write

$$G_r(x) = \sum_{n=0}^{\infty} \frac{h_n(\Gamma_r)}{n!} x^n$$

for the exponential generating function of the sequence  $(h_n(\Gamma_r))_n$ . We have:

**Theorem 4.5.** Let  $r \in \mathbb{N}$ . Then

$$G_r(x) = \prod_{j=0}^r \left( \left( 1 - x^{2^{j+1}} \right)^{-s_{2^j}/2} \exp\left( -2^j s_{2^j} + \frac{2^j s_{2^j}}{1 - x^{2^j}} \right) \right)$$

where

$$s_{2^{j}} = s_{2^{j}}((\mathbb{Z}/2\mathbb{Z})^{r}) = \frac{\prod_{l=0}^{j-1}(2^{r}-2^{l})}{\prod_{l=0}^{j-1}(2^{j}-2^{l})}.$$

The fact that this function is so explicit allows us to apply methods from analytic combinatorics (see [FS09] for a standard reference) in order to prove asymptotic results for its coefficients. The particular methods we use are due to Hayman [Hay56]. Similar methods have a long history in counting homomorphisms of finite groups into symmetric groups (see for instance [CHM51, MW56, MW57, Vol86, Wil86, Mül97]).

Examples of results we obtain are:

**Theorem 4.6.** (a) Let  $r \in \mathbb{N}$ . Then

$$s_n(\Gamma_r) = n \left( 1 + \sum_{\substack{0 < j \le r \\ s.t. \ 2^j \mid n}} 2^j \ s_{2^j} \right) + \sum_{\substack{0 \le j \le r \\ s.t. \ 2^{j+1} \mid n}} 2^j \ s_{2^j},$$

for all  $n \in \mathbb{N}$ .

(b) Let  $m \in \mathbb{N}_{\geq 2}$ ,  $r_1, \ldots, r_m \in \mathbb{N}$  and

$$\Gamma = *_{l=1}^m \Gamma_{r_l}.$$

Then there exist explicit constants  $A_{\Gamma}, B_{\Gamma} > 0$  and  $C_{\Gamma} \in \mathbb{Q}$  such that

$$s_n(\Gamma) \sim A_\Gamma n^{1+C_\Gamma} \exp(B_\Gamma \sqrt{n}) n!^{m-1}.$$

as  $n \to \infty$ .

The constants in the theorem above are explicit, but their definitions (especially that of  $A_{\Gamma}$ ) are lengthy. So for these, we refer to [BPR19]. We do note that they behave well

with respect to free products, that is:

$$A_{\Gamma} = \prod_{l=1}^{m} A_{r_l}, \quad B_{\Gamma} = \sum_{l=1}^{m} B_{r_l} \text{ and } C_{\Gamma} = \sum_{l=1}^{m} C_{r_l},$$

where  $A_r = A_{\Gamma_r}$ ,  $B_r = B_{\Gamma_r}$  and  $C_r = C_{\Gamma_r}$ . As to their values, we will content ourselves with a table with some values:

	Exact values			Numerical values	
r	$A_r$	$B_r$	$C_r$	$A_r$	$B_r$
0	$\frac{1}{\sqrt{8\pi}}\cdot\exp\left(-\frac{1}{2}\right)$	2	$-\frac{1}{2}$	0.1210	2
1	$\frac{1}{\sqrt[4]{2048\pi^2}} \cdot \exp\left(-\frac{7}{4}\right)$	$2\sqrt{2}$	$-\frac{1}{4}$	0.01457	2.8284
2	$\frac{1}{320\sqrt{\pi}} \cdot \exp\left(-\frac{63}{8}\right)$	$2\sqrt{5}$	$\frac{1}{2}$	$6.7020 \cdot 10^{-7}$	4.4721
3	$\frac{1}{68719476736\sqrt{\pi}}\exp\left(-\frac{671}{16}\right)$	8	$\frac{13}{4}$	$5.0248 \cdot 10^{-30}$	8

Table 4.1: The first four values of  $A_r$ ,  $B_r$  and  $C_r$ .

# 4.2 Covers by distinct manifolds

Another natural question related to finite index subgroup is how many *non-isomorphic* subgroups of the same index a given group has. For instance, if  $\Gamma$  is a free group or the fundamental group of a closed orientable surface and  $\Lambda < \Gamma$  is a finite index subgroup, then the index alone is enough to determine isomorphism type of  $\Lambda$ . Indeed, it's again a free group or the fundamental group of a closed surface respectively. Moreover, which free group or surface group it is can be determined through an euler characteristic computation. In other words, even if free groups and fundamental groups of closed surfaces (of genus at least 2) have many finite index subgroups, they don't have many pairwise non-isomorphic subgroups of the same index.

Unlike the case of free groups and surface groups, in general, we are not able to prove counting results. Instead, we will just ask whether groups have the property that all their subgroups of the same index are isomorphic. Even if surfaces and graphs play a very prominent role in this text, this particular property should be the exception rather than the rule. As such, together with Stefan Friedl, JungHwan Park, Jean Raimbault and Arunima Ray [FPP<sup>+</sup>21], we dubbed groups that have this property **exceptional**:

**Definition 4.7.** Let  $\Gamma$  be a group. If for all  $d \ge 1$ , for all pairs of subgroups of  $\Lambda_1, \Lambda_2 < \Gamma$ , both of index d,  $\Lambda_1$  and  $\Lambda_2$  are isomorphic as groups, we call  $\Gamma$  exceptional.

Our main motivation came from compact 3-manifolds, in which case the fundamental group often determines the manifold up to homeomorphism. Indeed, if M and N are closed, orientable, aspherical 3-manifolds and  $\pi_1(M) \simeq \pi_1(N)$ , then M and N are homeomorphic. This follows by combining Perelman's geometrization theorem [Per02, Per03a, Per03b], Mostow's rigidity theorem [Mos68], and work of Waldhausen [Wal68] and Scott [Sco83] (see for instance [AFW15, Theorem 2.1.2])

So, it also makes sense to speak of exceptional manifolds:

**Definition 4.8.** Let M be manifold. If for all  $d \ge 1$ , for all pairs of covers of  $N_1 \to M$ and  $N_2 \to M$ , both of degree d,  $N_1$  and  $N_2$  are homeomorphic as manifolds, we call Mexceptional.

We proved:

**Theorem 4.9.** Let M be a compact 3-manifold with empty or toroidal boundary. Then M is exceptional if and only if it is homeomorphic to one of the following manifolds:

- 1.  $k \cdot (\mathbb{S}^1 \times \mathbb{S}^2)$  for  $k \ge 1$ ,
- 2.  $\mathbb{S}^1 \times \mathbb{S}^2$ ,
- 3.  $\mathbb{S}^1 \times \mathbb{D}^2$ ,
- 4.  $\mathbb{T}^2 \times I$ ,
- 5.  $\mathbb{T}^3$ ,
- 6. all spherical manifolds except those with fundamental group  $P_{48} \times \mathbb{Z}/p\mathbb{Z}$  with gcd(p,3) = 1 and p odd, or  $Q_{8n} \times \mathbb{Z}/q$  with gcd(q,n) = 1, q odd, and  $n \ge 2$ .

In the theorem above,  $\mathbb{S}^n$  denotes the *n*-sphere,  $\mathbb{D}^n$  denotes the *n*-dimensional disk,  $\mathbb{T}^n$  denotes the *n*-torus, *I* denotes the unit interval [0,1] and  $k \cdot M$  denotes the *k*-fold connected sum of the manifold *M*. The finite groups listed in item (6) are defined in [FPP<sup>+</sup>21, Section 6]. Our proof uses a divide and conquer strategy, we use Kneser's prime decomposition theorem and Perelman's geometrization theorem to divide our manifold into various geometric classes that we then treat separately. For the full proof, we refer to the paper, here we'll give a proof for the hyperbolic case:

Proof that closed hyperbolic 3-manifolds are not exceptional. Let M be a closed hyperbolic manifold. By Agol's theorem [Ago13], M admits a cover of finite degree  $N \to M$  such that the first Betti number  $b_1(N) > 0$ .

This means that  $\pi_1(N)$  admits a surjection  $\varphi : \pi_1(N) \to \mathbb{Z}$  and in particular N admits a sequence of cyclic covers  $Y_n^{(1)} \to N$  of degree n. The systoles of this sequence of hyperbolic manifolds are uniformly bounded. Indeed, the subgroup  $K = \ker(\varphi)$  necessarily contains hyperbolic elements. Their translation length is an upper bound for the systole of  $Y_n^{(1)}$  for all n.

On the other hand, N is residually finite by Malcev's theorem [Mal40]. This means that N admits a sequence of finite degree covers  $Y_{n_k}^{(2)} \to N$  whose systoles tend to infinity. Since we have a cyclic cover of any degree, we can find a pair of covers

$$Y_n^{(1)}, Y_n^{(2)} \to N \to M$$

whose systoles are different but whose degrees are the same. By Mostow rigidity [Mos68],  $Y_n^{(1)}$  and  $Y_n^{(2)}$  cannot be homeomorphic as manifolds.

The use of Agol's theorem in the proof above makes it considerably shorter, but it's in fact not necessary. In our paper we present a proof that avoids it. It still uses systoles but instead of Agol's theorem, we use the fact that we can embed a lattice in  $PSL(2, \mathbb{C})$ in  $GL_d(\mathbb{F})$  for some number field  $\mathbb{F}$  (see [MR03, Theorem 3.1.2]). This allows us to apply Nori–Weisfeiler's strong approximation theorem [Wei84a] to find non-isomorphic subgroups of the same finite index. The advantage of that proof is that it generalizes to all semisimple Lie groups that are not locally isomorphic to  $PGL(2, \mathbb{R})$ . Agol's largeness theorem [Ago13] combined with arguments similar to those in [BGLM02, BGLS10] also allow us to prove that, in the case of hyperbolic 3-manifolds, the number of non-isomorphic subgroups of bounded index still grows factorially fast as a function of the index.

# 4.3 Counting curves

The final counting problem that we will briefly discuss is that of counting closed curves of surfaces. There are essentially two (related) types of questions: geometric and topological questions. In this section we will consider the latter. We will discuss some aspects of the former in Section 5.1.

Here, we will ask for the number of closed curves on a closed surface  $\Sigma$ , considered up to homeomorphism and isotopy, with a bounded number of self intersections. One way to describe this is to consider the **mapping class group** 

$$MCG(\Sigma) = Homeo^+(\Sigma)/Homeo^+_0(\Sigma),$$

where  $\operatorname{Homeo}^+(\Sigma)$  denotes the group of orientation preserving homeomorphisms  $\Sigma \to \Sigma$ and  $\operatorname{Homeo}^+_0(\Sigma)$  the normal subgroup consisting of those homeomorphisms that are isotopic to the identity. In this definition, we can replace homeomorphisms by diffeomorphisms and isotopy with homotopy without affecting the resulting mapping class group.

This group acts on the set of free homotopy classes of closed curves. This allows us to define

$$N_k(\Sigma) := \left| \left\{ \begin{array}{c} \text{free homotopy classes of closed curves} \\ \text{on } \Sigma \text{ with } k \text{ self-intersections} \end{array} \right\} \ \Big/ \ \mathrm{MCG}(\Sigma) \right|$$

Here, the number of self-intersections of a free homotopy class of closed curves is the minimum of the self-intersection number of among all curves in the class. One reason to care about  $N_k(\Sigma)$  is that it shows up in the count of the number of closed geodesics on a hyperbolic surface with a fixed number of self intersections and of bounded length [Mir08, Mir16, ES16, EPS20, ES22, DGZZ21, DGZZ22, Liu22, VM22].

There are two immediate questions to ask: how does  $N_k(\Sigma)$  depend on k, and how does it depend on  $\Sigma$ ? We'll start with the former. Aougab–Souto [AS18], improving upon work by Sapir [Sap16], proved that, if  $\Sigma_g$  is a closed orientable surface of genus  $g \ge 2$ , for all  $\delta > 0$ ,

$$\exp\left((\pi\sqrt{2g-2}-\delta)\cdot\sqrt{k}\right) \leqslant N_k(\Sigma_g) \leqslant \exp\left((4\sqrt{2}\sqrt{2g-2}+\delta)\cdot\sqrt{k}\right),$$

for all  $k \gg 0$ . Aougab–Souto conjecture that their lower bound is sharp:

#### Conjecture (Aougab–Souto)

Let  $\Sigma_g$  be a closed orientable surface of genus  $g \ge 2$ , then:

$$\lim_{k \to \infty} \frac{\log(N_k(\Sigma))}{\sqrt{k}} = \pi \sqrt{2g - 2}.$$

With Patricia Cahn and Federica Fanoni [CFP18], we considered the opposite regime. That is, we fix  $k \ge 0$  and ask what happens if we vary the surface. We proved:

Theorem 4.10. We have

$$N_k(\Sigma_g) \sim C_k \cdot g^{k+1}$$
 as  $g \to \infty$ .

where  $C_k > 0$  is a constant that depends on k only.

Our proof makes heavy use of results of Hass and Scott [HS94, HS99] on properties of curves that are not in minimal position. The constant  $C_k$  is a weighted sum over a finite set of ribbon graphs that depends on k. In particular, it can be made explicit for small values of k. For instance, it's well known that  $C_0 = \frac{1}{2}$ .

# 5 Random manifolds

Over the past 20 years, enormous progress has been made on random hyperbolic manifolds. Today there are many different models of random hyperbolic surfaces [BM04, GPY11, Mir13, MP23, BCP21b] and random hyperbolic 3-manifolds [DT06, PR22].

The purpose of this chapter is to discuss these models, their geometric and topological properties, and how the contributions of my co-authors and me fit into this story. We will discuss:

- universal properties of random surfaces obtained by gluing polygons together, which is joint work with Thomas Budzinski and Nicolas Curien [BCP19],
- the distribution of the number of short closed geodesics in random surfaces, based on joint work with Christoph Thäle [PT18] and Maryam Mirzakhani [MP19],
- homological torsion of random Heegaard splittings, based on our project with Hyungryul Baik, David Bauer, Ilya Gekhtman, Ursula Hamenstädt, Sebastian Hensel, Thorben Kastenholz and Daniel Valenzuela [BBG<sup>+</sup>18],
- random covers of torus knot complements, that we worked on with Elizabeth Baker [BP23a],
- and random 3-manifolds with boundary, that we studied with Jean Raimbault [PR22].

# 5.1 Random surfaces

Random surfaces have been around for a long time both in mathematics and in theoretical physics. A model that has attracted a lot of attention is that of random triangulations (or more general cell decompositions, often called **maps**) of a fixed surface  $\Sigma$  of finite type. Often this surface is the 2-sphere, in which case the cell decompositions are called **planar** 

maps. In this model, we pick randomly pick a triangulation  $\Sigma$  with a bounded number of triangles in the set of homeomorphism classes of all such triangulations and ask what the result looks like as the number of triangles tends to infinity. By now, there are many techniques available (see for instance [LGM12, Mie14, Cur23] and references therein) and we have quite a detailed description of the local [AS03] and global [LG13, Mie13, BM22] geometry of these triangulations. Recently these methods have also found applications in hyperbolic geometry [Bud22, BZ23].

### 5.1.1 Random Belyĭ surfaces

The first model of random hyperbolic surfaces, called random Belyĭ surfaces, was developed by Brooks–Makover [BM04]. Their goal, inspired by random regular graphs, was to study random closed hyperbolic surfaces of large genus (so the opposite regime to the one of random planar maps), and find applications of the probabilistic method, in particular for the spectral gap.

A random Belyĭ surface is obtained by randomly partitioning the sides of n (an even number) ideal triangles into pairs and then gluing them into an oriented surface according to the side pairing (this is analogous to the configuration model in graph theory, see Section 2.2.4), using gluings of shear 0 (see Section 3.1.3). The resulting surface  $S_n^O$  is then conformally compactified, thus obtaining a closed random surface  $S_n^C$ , equipped with a conformal structure.

If the genus is at least 2 (which turns out to be generic), this conformal class contains a unique hyperbolic metric by the uniformization theorem. One reason that this is an interesting model of random surfaces is that the non-compact surfaces  $S_n^O$  are covers of  $PSL(2,\mathbb{Z})\backslash\mathbb{H}^2$ , which implies that the compact surfaces  $S_n^C$  are branched covers of the sphere, branched at at most three points. Such surfaces are called Belyĭ surfaces because Belyĭ proved they are exactly the surfaces that can be defined over  $\overline{\mathbb{Q}}$  [Bel79] (see [JS96] for a nice survey). This in particular implies that the set of surfaces we sample from, form a dense set in  $\mathcal{M}_g$  for all  $g \ge 0$ .

**Topology.** The first question is what the topology of these surfaces is like. First of all, we didn't specify that the gluing needs to be into a connected surface. However, it's true that

$$\mathbb{P}\left(S_n^O \text{ and } S_n^C \text{ are connected}\right) \to 1 \text{ as } n \to \infty$$

which follows directly from the analogous result on random regular graphs due to Bollobás and Wormald [Bol81, Wor81a]. Knowing this, the only remaining question is their genus (or equivalently, the number of cusps of  $S_n^O$ ). Very precise results on this are known due to Gamburd (when 4 divides n) [Gam06] and Chmutov–Pittel [CP16]. They for instance prove the following central limit theorem for genus of these surfaces:

$$\frac{\operatorname{genus}(S_n^C) - 1 - \frac{n}{4} - \log(n)}{\sqrt{\log(n)}} \xrightarrow{\operatorname{distribution}} \mathcal{N}(0, 1) \quad \text{as } n \to \infty,$$

where  $\mathcal{N}(0,1)$  denotes the standard normal distribution.

They prove this using the fact that a random surface can be modeled using a uniformly random pair of permutations  $(\sigma_n, \tau_n) \in K(3^n) \times K(2^{3n/2})$ . Here  $K(3^n) \subset \mathfrak{S}_{3n}$  denotes the conjugacy class of fixed point free permutations of order 3 and  $K(2^{3n/2}) \subset \mathfrak{S}_{3n}$  the conjugacy class of fixed point free involutions. The notation " $3^n$ " and " $2^{n/2}$ " is exponential notation for the partitions that describe the cycle types of the permutations in these conjugacy classes. The permutation  $\sigma_n$  encodes the oriented labeling of the sides of the ideal triangles and the permutation  $\tau_n$  describes the side pairing. Their product  $\sigma_n \cdot \tau_n$ describes the cusps of  $S_n^O$ , in the sense that each cusp naturally corresponds to a cycle in the disjoint cycle decomposition of  $\sigma_n \cdot \tau_n$  and moreover the length of that cycle gives us the "size" of the cusp: the number of corners of ideal triangles incident to it.

Using ideas from random walks on finite groups, in particular the Diaconis–Shahshahani lemma [DS81], Gamburd and Chmutov–Pittel proved that, denoting the distribution of  $\sigma_n \cdot \tau_n \in \mathfrak{S}_{3n}$  by  $\mathbb{P}_{\sigma_n \cdot \tau_n}$ , we have that,

$$d_{\mathrm{TV}}(\mathbb{P}_{\sigma_{4k}\cdot\tau_{4k}}, \mathbb{U}_{4k}^{\mathrm{even}}) \to 0 \quad \mathrm{as} \ k \to \infty$$

and

$$d_{\mathrm{TV}}(\mathbb{P}_{\sigma_{4k+2}:\tau_{4k+2}}, \mathbb{U}_{4k+2}^{\mathrm{odd}}) \to 0 \quad \text{as } k \to \infty$$

where  $d_{TV}$  denotes total variational distance,  $\mathbb{U}_{4k}^{\text{even}}$  the uniform measure on the set of even permutations (the alternating group) and  $\mathbb{U}_{4k+2}^{\text{odd}}$  the uniform measure on the set of odd permutations. This for instance implies the central limit theorem above and also that, when properly normalized, the partition describing the cycle type of  $\sigma_n \cdot \tau_n$  converges to a Poisson–Dirichlet distributed random variable as  $n \to \infty$  (see for instance [ABT00]). Chmutov–Pittel also generalized these results to random gluings of polygons with varying perimeters. In [BCP19], with Thomas Budzinski and Nicolas Curien, we proved a weaker version (that we dubbed Poisson–Dirichlet universality) of these results for a slightly larger class of polygon gluings. One of the main points of this is that we did this using different methods, that do in particular not rely on the representation theory of the symmetric group.

Instead, we used dynamical exploration techniques that are well known in the world of planar maps and are often called **peeling techniques** (see [Cur23] for an introduction) and that play an important role at multiple points in this text. The idea is very simple. Formally, our probability space is a finite set of pairings of sides of triangles. In particular, an element in the probability space does not make any reference to the order in which the triangles are glued together. However, we can choose to add an order without influencing the probability of any geometric or topological event. Such an order is called a **peeling algorithm** and is often thought of as the order in which the triangulation is being "discovered".

To show how the method works, we'll present a simple example. This example essentially already appears (without this language) in Brooks and Makover's paper [BM04]:

#### A peeling algorithm

#### Input:

- An even number  $n \in \mathbb{N}$
- A surface  $S^{(0)}$  which is a disconnected union of *n* oriented triangles, with their sides labeled  $1, 2, \ldots, 3n$

#### Iteration:

At step t = 1, ..., 3n/2, create the random surface  $S^{(t)}$  as follows

- 1. Let  $e_1^{(t)}$  and  $e_2^{(t)}$  denote two uniformly random edges on  $\partial S^{(t-1)}$
- 2. Glue  $e_1^{(t)}$  and  $e_2^{(t)}$  together in such a way that the orientations on both sides match. Call the resulting surface  $S^{(t)}$ .

This procedure generates a random sequence of surfaces  $S^{(0)}, \ldots, S^{(3n/2)}$ . The order we have introduced is of no influence on the resulting surface. That is, as a random triangulated surface, the law of  $S^{(3n/2)}$  is the same as that of  $S_n^O$ . Adding this order, however gives us access to the number of vertices in the triangulation. For instance, if we write  $V_n$  for the number of vertices of  $S^{(3n/2)}$ , then we have

$$\mathbb{E}(V_n) = \sum_{t=1}^{3n/2} \mathbb{E}(V_n^{(t)}),$$

where  $V_n^{(t)}$  denotes the number of vertices that in are in the interior of  $S^{(t)}$  but not in the interior of  $S^{(t-1)}$ . The value of  $V_n^{(t)}$  is either 0, 1 or 2. The latter only happens if  $e_1^{(t)}$  and  $e_2^{(t)}$  together form a boundary component of  $S^{(t-1)}$ , which is rare. Creating one vertex in the interior can happen in two ways, either  $e_1^{(t)}$  and  $e_2^{(t)}$  are incident to a common vertex (see Figure 5.1), or both  $e_1^{(t)}$  and  $e_2^{(t)}$  form a full boundary component of  $S^{(t-1)}$ . Of these two, the former is generic. As such, we obtain

$$\mathbb{E}(V_n) = o(\log(n)) + \sum_{t=1}^{3n/2} \mathbb{P}\left(\begin{array}{c} e_1^{(t)} \text{ and } e_2^{(t)} \text{ are on a boundary component containing} \\ \text{at least 3 edges and are incident to a common vertex} \end{array}\right).$$

as  $n \to \infty$  Also the event that  $e_1^{(t)}$  and  $e_2^{(t)}$  are on a boundary component containing at least 3 edges is generic. Using this and the fact that after step t - 1, there are 3n - 2t + 2edges left on the boundary, we obtain

$$\mathbb{E}(V_n) = o(\log(n)) + \sum_{t=1}^{3n/2} \frac{2}{3n - 2t + 1} = \log(n) + o(\log(n))$$

as  $n \to \infty$ , thus recovering the first order term in the central limit theorem above. For finer estimates for more general gluings and proofs of the claims of genericity above, we refer to [BCP19].

In that paper, we also observed that other geometric properties of the graph structure of a random polygon decomposition seem to be universal. To formalize this, let  $\mathcal{P}_n = \{p_1, \ldots, p_k\}$  be a multiset of perimeters of polygons such that  $\sum_i p_i = n$ . Moreover let  $G_{\mathcal{P}_n}$  denote the random graph obtained from a random gluing of polygons of the perimeters prescribed by  $\mathcal{P}_n$ , using the configuration model. The structure of this graph is not entirely random: there is a parity constraint similar to the constraint showing up in the results of Gamburd and Chmutov-Pittel discussed above. Indeed, an Euler characteristic computation shows that the parity of the number of vertices of  $G_{\mathcal{P}_n}$  is the same as the parity of n + k. The question is whether this is essentially the only constraint:



Figure 5.1: Closing off a vertex of  $S^{(t)}$ 

#### Question 9

Let  $\mathbb{G}_n$  be the random graph structure of a uniform random labeled map on n edges, and denote  $\mathbb{G}_n^{\text{odd}}$  (respectively  $\mathbb{G}_n^{\text{even}}$ ) be the random graph  $\mathbb{G}_n$  conditioned respectively on having an odd (respectively even) number of vertices. Suppose moreover that  $(\mathcal{P}_n)_n$  is a sequence of multisets of perimeters such that  $\sum_{p \in \mathcal{P}_n} p = n$  and

$$\lim_{n \to \infty} \frac{|\{p_i \in \mathcal{P}_n; p_i = 1\}|}{\sqrt{n}} = 0 \quad \text{and} \quad \lim_{n \to \infty} \frac{|\{p_i \in \mathcal{P}_n; p_i = 2\}|}{n} = 0$$

Then is it true that

 $d_{\mathrm{TV}}(G_{\mathcal{P}_n}, \mathbb{G}_n^{\epsilon_n}) \to 0 \quad \text{as } n \to \infty \quad ?$ 

Here  $\epsilon_n \in \{\text{even}, \text{odd}\}\ \text{denotes the parity of } |\mathcal{P}_n| + n.$ 

The conditions on the number of monogons and bigons in this question are necessary to make the earlier mentioned peeling based proofs work. They for instance play a role in the proofs of the genericity claims before.

**Geometry.** We also know a lot about the geometry of random Belyĭ surfaces. Brooks and Makover already proved that there exist constants  $c_1, c_2, c_3, c_4 > 0$  such that, with high

probability as  $n \to \infty^{-1}$ ,

$$\operatorname{sys}(S_n^C) \ge c_1, \quad \lambda_1(S_n^C) \ge c_2, \quad h(S_n^C) \ge c_3 \quad \text{and} \quad \operatorname{diam}(S_n^C) \le c_4 \log(\operatorname{genus}(S_n^C)).$$

Here,  $h(S_n^C)$  denotes the Cheeger constant of  $S_n^C$ , defined as

$$h(S_n^C) = \inf \left\{ \frac{\ell(\partial Y)}{\operatorname{area}(Y)}; \begin{array}{l} Y \subset S_n^C \text{ a submanifold with smooth} \\ \text{boundary and } \operatorname{area}(Y) \leqslant \operatorname{area}(S_n^C) \end{array} \right\}$$

which measures how hard it is to cut  $S_n^C$  in two. We will discuss this quantity and its interest at length in Section 6.3. diam $(S_n^C)$  denotes the diameter of  $S_n^C$ . By now, sharper bounds are known on this diameter [BCP21a], which we will briefly discuss in Section 6.2. All these bounds suggest that random Belyĭ surfaces are good expanders. Whether they're near Ramanujan surfaces is currently still open, but it seems quite likely they are, given that the combinatorics and geometry of the model are very close to that of the random covers of Hide–Magee [HM23b]:

Conjecture	
For every $\varepsilon > 0$ ,	$\lim_{n \to \infty} \mathbb{P}\left(\lambda_1(S_n^C) > \frac{1}{4} - \varepsilon\right) = 1.$

The method of proof of the bounds by Brooks and Makover is to first understand the geometry of the non-compact surfaces  $S_n^O$ , using a combination of results on random graphs and hyperbolic geometry and then prove that the given bounds persist (up to manageable errors) in the compactification. This last step is based on the same earlier work due to Brooks [Bro99] that we have mentioned before. This works for the systole, the spectral gap and the Cheeger constant. However, the diameter of  $S_n^O$  is infinite. Instead, the logarithmic upper bound on the diameter of  $S_n^C$  follows from the bounds on the other three invariants, again using earlier work by Brooks [Bro92].

Together with Christoph Thäle [PT18], we studied the length spectrum of random Belyĭ surfaces. Let  $\mathcal{L}(S_n^C)$  denote the **primitive length spectrum** of  $S_n^C$ : the multiset of lengths of primitive closed geodesics on  $S_n^C$ . This is a random countable subset of  $(0, \infty)$ . In order to state our theorem, recall from Section 3.1.3 that the lengths of curves on  $S_n^O$ 

<sup>&</sup>lt;sup>1</sup>The phrase "with high probability as  $n \to \infty$ " means that the probability of the described event tends to 1 as  $n \to \infty$ 

are determined by words in two matrices L and R. To this end, we set

$$W = \{ words in L and R \} / \sim$$

where two words w and w' are equivalent if (as strings in L and R) if w' can be obtained from w through a combination of the following two operations

- cyclic permutation of the letters,
- reading the word backwards and interchanging L and R.

This equivalence captures the ambiguity in the definition of the word associated to a closed curve mentioned in Section 3.1.3. Given  $[w] \in W$ , we will write

$$\ell([w]) = 2\operatorname{arccosh}\left(\frac{\operatorname{tr}(w)}{2}\right).$$

We proved the following:

**Theorem 5.1.** As  $n \to \infty$ ,  $\mathcal{L}(S_n^C)$  converges locally in total variational distance<sup>2</sup> to a Poisson point process  $\mathcal{L}^{BM}$  of intensity  $\lambda^{BM}$  given by

$$\lambda^{\mathrm{BM}}(A) = \sum_{\substack{[w] \in W \\ \ell(w) \in A}} \frac{|[w]|}{2|w|}, \quad A \subset (0, \infty).$$

Because they play an important role in this text, we remind the reader of the definition of a **Poisson point process**.

**Definition 5.2.** Let X be a manifold and  $\mu$  a locally finite Radon measure on X. A Poisson point process on X of intensity  $\mu$  is a random countable subset  $S \subset X$  such that:

(a) For any compact measurable  $A \subset X$ , the random variable

$$N_A = |\mathcal{S} \cap A| \sim \text{Poisson}(\mu(A)).$$

(b) If  $A_1, \ldots, A_k \subset X$  are disjoint compact subsets, then the random variables  $N_{A_1}, \ldots, N_{A_k}$  are an independent family.

<sup>&</sup>lt;sup>2</sup>See [Kal17, Section 4.4] for a definition.
With Christoph Thäle, we obtain explicit bounds on the total variational distance of the random variables that count the numbers of geodesics of bounded length to the counting variables corresponding to  $\mathcal{L}^{BM}$ . This in particular allows us to say something about geodesics whose length grows (slowly) with *n*. Our theorem strengthens work from the author's PhD thesis [Pet17b, Pet17a] in which the method of moments was used to prove local convergence in distribution. In our work with Christoph Thäle, we use the Chen–Stein method for Poisson approximation instead [Che75a, AGG89, BHJ92].

Our result with Christoph Thäle also implies that the systole of random Belyĭ surfaces is not large (for instance, its expectation is uniformly bounded [Pet17b]). In upcoming work with Mingkun Liu [LP23], we will prove that the construction can be modified, using ideas from graph theory due to Linial–Simkin [LS21], in order to produce random surfaces with logarithmic systoles.

### 5.1.2 Random covers of finite degree

Another combinatorial model is that of random covers of a fixed surface. That is, one fixes a hyperbolic surface  $X_1$  with a finitely generated fundamental group  $\Gamma$  and takes a uniformly random index *n* subgroup of  $\Gamma$ , which gives rise to a random cover of degree *n* of  $X_1$ . Due to results by Dixon [Dix69] for free groups and Müller–Schlage-Puchta [MP02, MSP07] and Liebeck–Shalev [LS04a] for surface groups, this model is essentially the same as picking a uniformly random homomorphism

$$\varphi_n \in \operatorname{Hom}(\Gamma, \mathfrak{S}_n)$$

and taking the cover corresponding to the stabilizer  $H_n = \operatorname{Stab}_{\varphi_n}(\{1\})$ .

If the base surface is a non-compact surface of finite area, and hence  $\Gamma$  is a finitely generated free group, this model is somewhat similar to the model of random Belyĭ surfaces discussed above. Counts of the number of short closed geodesics can be derived from results due to Nica [Nic94] and imply in particular that as an IRS (see Section 4.1.1),  $H_n$  converges to the trivial group and the corresponding cover Benjamini–Schramm converges to  $\mathbb{H}^2$ . As we mentioned in Section 3.3, Hide–Magee [HM23b] proved that these surfaces are near Ramanujan with high probability as  $n \to \infty$ . A version of the latter result in the case of Schottky groups in PSL(2,  $\mathbb{R}$ ) is due to Magee–Naud [MN20, MN21].

If the base surface is closed, the fact that its fundamental group is not free makes counting the number of homomorphisms significantly more complicated. Nonetheless, good asymptotic results for this number are known due to the works by Müller–Schlage-Puchta and Liebeck–Shalev mentioned above.

Also the question of counting the number of short closed geodesics that lift to the cover becomes more complicated, but recently techniques for estimating this were developed by Magee–Naud–Puder [MP23, MNP22], which in particular give rise to a Poisson approximation theorem for the length spectrum of these random covers [PZ22] and again imply convergence of the corresponding IRS and Benjamini–Schramm convergence to the hyperbolic plane.

Like in the Brooks–Makover model (and the Weil–Petersson model that we will discuss below), the results above imply that these random constructions do not give rise to surfaces with large systoles. In the earlier mentioned upcoming work with Mingkun Liu [LP23] we will prove that certain random *regular* covers, much like the random Cayley graphs of Section 2.2.4, do yield to surfaces with logarithmic systoles.

Magee–Naud–Puder also prove that for any  $\varepsilon > 0$ , with high probability as the degree of the cover tends to infinity, these covers have no new Laplacian eigenvalues<sup>3</sup> in the interval  $\left[0, \frac{3}{16} - \varepsilon\right]$ . They conjecture that this should hold up to  $\frac{1}{4}$ :

Conjecture (Magee-Naud-Puder)

Let X be a closed orientable hyperbolic surface and let  $X_n \to X$  denote a random cover of degree n of X, as defined above. Moreover, given a closed hyperbolic surface Y, let  $\sigma(Y)$  denote the spectrum of its Laplacian. Then for any  $\varepsilon > 0$ 

$$\lim_{n \to \infty} \mathbb{P}\left(\sigma(X_n) \cap \left[0, \frac{1}{4} - \varepsilon\right] = \sigma(X) \cap \left[0, \frac{1}{4} - \varepsilon\right]\right) = 1.$$

Finally, Naud determined the asymptotic behavior of the variance of smoothened counting functions of eigenvalues of the twisted Laplace operators [Nau22] and the determinant of these surfaces [Nau23].

### 5.1.3 The Weil–Petersson measure

The final model of random hyperbolic surface we will discuss in this chapter (we'll introduce another one in Section 6.2) is the model coming from the Weil–Petersson measure on the

 $<sup>^{3}</sup>$ New Laplacian eigenvalues on a finite degree cover are eigenvalues that do not correspond to eigenvalues obtained by lifting an eigenfunction from the base.

moduli space  $\mathcal{M}_g$  of closed hyperbolic surfaces of genus g. This is the measure coming from the Weil–Petersson volume form, which in turn comes from a Kähler 2-form called the Weil–Petersson form. Random surfaces sampled using this measure were first studied by Mirzakhani [Mir13] and Guth–Parlier–Young [GPY11]: Mirzakhani proved that these random surfaces have similar expansion properties to random Belyĭ surfaces. Guth, Parlier and Young used these surfaces to attack an extremal problem on pants decompositions (see Section 6.1 below for a description of these results).

Together with Maryam Mirzakhani [MP19], we proved a similar Poisson approximation result to Theorem 5.1 for surfaces distributed according to this measure. In the theorem below,  $X_g$  will denote a random hyperbolic surface in  $\mathcal{M}_g$ , distributed according to the Weil–Petersson probability measure.  $\mathcal{L}(X_g)$  will again denote its primitive length spectrum, a random subset of  $(0, \infty)$ . We proved:

**Theorem 5.3.** As  $g \to \infty$ ,  $\mathcal{L}(X_g)$  converges locally in distribution<sup>4</sup> to a Poisson point process  $\mathcal{L}^{WP}$  of intensity  $\lambda^{WP}$  given by

$$\lambda^{\rm WP}(A) = \int_A \frac{\cosh(t) - 1}{t} dt.$$

Our proof is based on the method of moments, combined with Mirzakhani's integration techniques for Weil–Petersson volumes [Mir07a, Mir07b]. Curiously, these same statistics have been observed for certain models of unicellular maps by Janson–Louf [JL22, JL23].

In this section, we will describe some of the ingredients of the proof of the theorem above, starting with an explanation of the Weil–Petersson volume form itself. In the end of this section we'll survey some of the other known results on random surfaces distributed according to the Weil–Petersson measure.

Tecihmüller and moduli spaces. In this text, we won't describe this Kähler structure. Instead, we'll start with Wolpert's results on the associated symplectic form. This will be enough for us, because in the end we're only interested in the volume form. To describe Wolpert's work, we need to first describe Fenchel–Nielsen coordinates. These are global coordinates on the Teichmüller space of the closed oriented surface  $\Sigma_q$  of genus  $g \ge 2$ :

$$\mathcal{T}_{g} = \left\{ (X, f); \begin{array}{c} X \text{ an oriented hyperbolic surface and} \\ f: \Sigma_{g} \to X \text{ an orientation preserving diffeomorphism} \end{array} \right\} \Big/ \sim$$

,

<sup>&</sup>lt;sup>4</sup>See [Kal17, Section 4.1] for a definition.

where  $(X_1, f_1) \sim (X_2, f_2)$  if and only if there exists an isometry  $m : X_1 \to X_2$  such that the map

$$f_2^{-1} \circ m \circ f_1 : \Sigma_g \to \Sigma_g$$

is homotopic to the identity. The mapping class group  $MCG(\Sigma_g)$  (see Section 4.3 for its definition) acts on  $\mathcal{T}_g$  by

$$[\varphi] \cdot [X, f] = [X, f \circ \varphi^{-1}], \quad [X, f] \in \mathcal{T}_g, \varphi \in \mathrm{MCG}(\Sigma_g)$$

this action is properly discontinuous, but not free, and the quotient is the moduli space  $\mathcal{M}_{g}$ .

Given a pants decomposition  $\{\alpha_1, \ldots, \alpha_{3g-3}\}$  of  $\Sigma_g$ , we can define Fenchel–Nielssen coordinates:

$$\left(\ell_{\alpha_1}, \tau_{\alpha_1}, \ldots, \ell_{\alpha_{3g-3}}, \tau_{\alpha_{3g-3}}\right) : \mathcal{T}_g \longrightarrow (\mathbb{R}_{>0} \times \mathbb{R})^{3g-3}.$$

The functions  $\ell_{\alpha}$  measure lengths of curves:  $\ell_{\alpha}([X, f])$  is the length of the unique geodesic on X in the free homotopy class of  $f(\alpha)$ . The coordinates  $\tau_{\alpha_i}$  measure the twists along the pants curves (similar to Section 3.1.4). It turns out that the induced map is a homeomorphism and that moreover the coordinate changes (when changing pants decomposition) are analytic.

Wolpert proved that these coordinates are in fact Darboux coordinates for the symplectic form associated to the Weil–Petersson Kähler form [Wol82]. That is, if we write  $\omega_{WP}$  for the Weil–Petersson symplectic form, then

$$\omega_{\rm WP} = \sum_{i=1}^{3g-3} d\ell_{\alpha_i} \wedge d\tau_{\alpha_i}.$$

In particular, the associated volume form is just the usual Lebesgue volume form on  $(\mathbb{R}_{>0} \times \mathbb{R})^{3g-3}$ . Moreover, it descends to a volume form of finite total volume on  $\mathcal{M}_g$ . This allows us to define a probability measure, simply by normalizing it:

$$\mathbb{P}(A) = \frac{\operatorname{vol}_{WP}(A)}{\operatorname{vol}_{WP}(\mathcal{M}_g)}, \quad \forall A \subset \mathcal{M}_g \text{ measurable},$$

where

$$\operatorname{vol}_{WP}(A) = \int_{A} \frac{\omega_{WP}^{\wedge(3g-3)}}{(3g-3)!}, \quad \forall A \subset \mathcal{M}_g \text{ measurable.}$$

Computing Weil–Petersson volumes. Of course, theoretically, Weil–Petersson volumes can be computed by integrating over a suitable fundamental domain for the  $MCG(\Sigma_g)$ -action on  $\mathcal{T}_g$ . Unfortunately, the topology of  $\mathcal{M}_g$  is very complicated, so no easy to describe sequences of such fundamental domains are known. There are estimates of Weil–Petersson volumes based on approximate fundamental domains [Pen92, Gru01, ST01], but they are generally not sharp enough to answer the type of questions we are interested in. A notable exception is the paper by Guth–Parlier–Young [GPY11], in which these bounds suffice.

In her thesis [Mir07a, Mir07b], Mirzakhani developed an alternative approach to the calculation of Weil–Petersson volumes. In order to describe this, we need to slightly enlarge our set of moduli spaces. First fix an orientable surface  $\Sigma_{g,n}$  of genus g and with n boundary components. Given  $L_1, \ldots, L_n \ge 0$ , we now define the Teichmüller space  $\mathcal{T}_{g,n}(L_1, \ldots, L_n)$  as

$$\mathcal{T}_{g,n}(L_1,\ldots,L_n) = \left\{ \begin{array}{cc} X \text{ an oriented hyperbolic surface with totally} \\ (X,f); & \text{geodeisc boundary and } f: \Sigma_{g,n} \to X \text{ an} \\ & \text{orientation preserving diffeomorphism} \end{array} \right\} \Big/ \sim,$$

By convention, when we say "totally geodesic boundary component of length 0" we mean "cusp". The notion of equivalence is the same as before, except that we now also require the maps  $f_2^{-1} \circ m \circ f_1 : \Sigma_{g,n} \to \Sigma_{g,n}$  to preserve the boundary components of  $\Sigma_{g,n}$  setwise. Likewise, the mapping class group  $MCG(\Sigma_{g,n})$  is the group of isotopy classes of diffeomorphisms  $\Sigma_{g,n} \to \Sigma_{g,n}$  that preserve the boundary components setwise. The corresponding moduli space is

$$\mathcal{M}_{g,n}(L_1,\ldots,L_n) = \mathcal{T}_{g,n} / \mathrm{MCG}(\Sigma_{g,n}).$$

By work of Goldman [Gol84] and Wolpert [Wol82], this moduli space carries also carries a symplectic form that admits the same expression in terms of Fenchel–Nielsen coordinates. In particular, we can speak of its Weil–Petersson volume.

In order to find functions we can integrate, Mirzakhani's idea is to introduce a class of functions  $\mathcal{M}_g \to \mathbb{R}$  that she calls **geometric functions**. These are constructed as follows:

**Definition 5.4.** Let  $\Gamma = (\gamma_1, \ldots, \gamma_k)$  be a sequence of pairwise disjoint essential free homotopy classes of closed curves on  $\Sigma_{g,n}$ . Moreover, let  $F : \mathbb{R}^k_{\geq 0} \to \mathbb{R}$ . Then the geometric function  $F^{\Gamma} : \mathcal{M}_{g,n}(L_1, \ldots, L_n) \to \mathbb{R}$  is defined by

$$F^{\Gamma}(X) = \sum_{(\alpha_1, \dots, \alpha_k) \in \mathrm{MCG}(\Sigma_{g,n}) \cdot (\gamma_1, \dots, \gamma_k)} F(\ell_{\alpha_1}(X), \dots, \ell_{\alpha_k}(X))$$

For example, if  $\gamma$  is a simple non-separating closed curve on  $\Sigma_g$  and  $F = \chi_{[0,L]}$  is the characteristic function of the interval [0, L], then the geometric function  $F^{\gamma} : \mathcal{M}_g \to \mathbb{N}$  is given by

$$F^{\gamma}(X) = \left| \left\{ \begin{array}{c} \text{simple non-separating closed geodesics} \\ \text{of length at most } L \text{ on } X \end{array} \right\} \right|$$

To state Mirzakhani's integration formula, we will, given a sequence of pairwise disjoint essential free homotopy classes of closed curves  $\Gamma = (\gamma_1, \ldots, \gamma_k)$  on  $\Sigma_{g,n}, L \in \mathbb{R}^n_{\geq 0}$  and  $x \in \mathbb{R}^k_{\geq 0}$ , use the notation  $V_{g,n}(\Gamma, L, x)$  for the Weil–Petersson volume of the moduli space of hyperbolic metrics on the surface with boundary  $\overline{\Sigma_{g,n} - (\gamma_1 \cup \cdots \cup \gamma_k)}$ , whose boundary lengths are  $L_1, \ldots, L_n, x_1, x_1, x_2, x_2, \ldots, x_k, x_k$ .

In [Mir07a], Mirzakhani proved the following formula:

**Theorem 5.5** (Mirzakhani integration formula). Let  $\Gamma = (\gamma_1, \ldots, \gamma_k)$  be a sequence of pairwise disjoint essential free homotopy classes of closed curves on  $\Sigma_{g,n}$  and  $F : \mathbb{R}^k_{\geq 0} \to \mathbb{R}$ . Then

$$\int_{\mathcal{M}_{g,n}(L)} F^{\Gamma}(X) \, d\operatorname{vol}_{WP}(X) = C_{\Gamma} \cdot \int_{\mathbb{R}^{k}_{\geq 0}} F(x_{1}, \dots, x_{k}) \cdot V_{g,n}(\Gamma, L, x) \cdot x_{1} \cdots x_{k} \, dx_{1} \cdots dx_{k},$$

where  $C_{\Gamma} > 0$  is a computable constant that depends on  $\Gamma$  only.

The constant  $C_{\Gamma}$  in the theorem above is computable, but it's also very easy to get it wrong. For a formula for it, we refer to [Wri20, Theorem 4.1]. Here, we just note that if  $\Gamma = (\gamma_1, \ldots, \gamma_k)$  is such that  $\Sigma_g - (\gamma_1 \cup \cdots \cup \gamma_k)$  is connected (this determines a unique MCG( $\Sigma_g$ )-orbit), then  $C_{\Gamma} = 2^{-k}$ .

Mirzakhani used this formula, combined with the beautiful idea that the McShane– Mirzakhani identity turns constant functions into geometric functions, to derive a recurrence for Weil–Petersson volumes of moduli spaces of surfaces with boundary. These recurrences are however quite complicated, so it is a lot work to extract good asymptotic bounds on Weil–Petersson volumes of moduli spaces of surfaces of large genus (that are crucial for applications to random surfaces) from it. Nonetheless, Mirzakhani and Zograf [MZ15] proved that, for any  $n, K \ge 0$  fixed

$$\operatorname{vol}_{WP}(\mathcal{M}_{g,n}) = c_{WP} \cdot \frac{(2g - 3 + n)!(4\pi^2)^{2g - 3 + n}}{\sqrt{g}} \left( 1 + \sum_{k=1}^{K} \frac{a_{n,k}}{g^k} + O\left(\frac{1}{g^{K+1}}\right) \right) \quad \text{as } g \to \infty,$$

where  $c_{WP} > 0$  is a universal constant and the constants  $a_{n,k}$  are effectively computable.

### 5.1. RANDOM SURFACES

Based on numerical data, Zograf [Zog20] conjectured:

### **Conjecture** (Zograf)

The multiplicative constant in the asymptotic expansion for Weil–Petersson volumes satisfies:

$$c_{\rm WP} = \frac{1}{\sqrt{\pi}}.$$

The current best known asymptotic expansions on the ratios

$$\operatorname{vol}_{\operatorname{WP}}(\mathcal{M}_{q,n}(L_1,\ldots,L_n))/\operatorname{vol}_{\operatorname{WP}}(\mathcal{M}_{q,n}))$$

are due to Anantharaman–Monk [AM22] and improve on estimates by Mirzakhani [MP19].

The length spectrum. The proof of the Poisson approximation theorem for the length spectrum (Theorem 5.3) uses the method of moments. First of all, we will write  $N_{[a,b]}$ :  $\mathcal{M}_g \to \mathbb{N}$  for the random variable that counts the number of closed geodesics whose length lies in [a, b]. The goal is then to estimate the factorial moments

$$\mathbb{E}\left(\prod_{i=1}^{k} \left(N_{[a_i,b_i]}(X_g)\right)\right)_{m_i}\right) = \frac{1}{\operatorname{vol}_{WP}(\mathcal{M}_g)} \int_{\mathcal{M}_g} \prod_{i=1}^{k} \left(N_{[a_i,b_i]}(X)\right)_{m_i} d\operatorname{vol}_{WP}(X),$$

where

$$(N_{[a,b]}(X))_m := N_{[a,b]}(X) \cdot (N_{[a,b]}(X) - 1) \cdots (N_{[a,b]}(X) - m + 1).$$

In particular, we need to show that, for any collection of disjoint intervals  $[a_1, b_1], \ldots, [a_k, b_k]$ and any  $m_1, \ldots, m_k \in \mathbb{N}$ ,

$$\mathbb{E}\left(\prod_{i=1}^{k} \left(N_{[a_i,b_i]}(X_g)\right)\right)_{m_i} \longrightarrow \prod_{i=1}^{k} \lambda_{[a_i,b_i]}^{m_i},$$

as  $g \to \infty$ .

To do so, we first observe that the random variable  $\prod_{i=1}^{k} (N_{[a_i,b_i]})_{m_i} : \mathcal{M}_g \to \mathbb{N}$  counts ordered tuples of geodesics: the number of tuples of  $m_1 + \ldots + m_k$  geodesics such that the lengths of  $m_1$  of these geodesics lie in  $[a_1, b_1]$ , the lengths of  $m_2$  of these geodesics lie in  $[a_2, b_2]$ , et cetera.

Next, we split this counting function into two counting functions. The first of these counts only tuples of simple geodesics that pairwise don't intersect each other and moreover

leave the surface connected once removed. The second function counts the remaining tuples.

The first function is a geometric function, so we can develop its expectation using Theorem 5.5. The resulting expression can then be estimated using the earlier mentioned asymptotic results on (ratios of) Weil–Petersson volumes. The resulting limit is exactly the expression above.

The second part of the proof is to show that the expectation of the function that counts tuples that do intersect or separate is negligible when  $g \to \infty$ . To do that, we first argue that short tuples with intersections give rise to small subsurfaces with a short boundary. Indeed, if we take a regular neighborhood of such a tuple, then we obtain a subsurface with boundary, the length of which is comparable to the length of the curve. It then remains to prove that it's rare that a surface has such small subsurfaces with a short boundary. This again uses the asymptotic results mentioned above.

Other results on the geometry and spectrum of  $X_g$ . During the last couple of years there has been an explosion of work on random surfaces of large genus distributed according to the Weil–Petersson measure.

The geometry of closed geodesics on these surfaces has been further investigated in [NWX23, WX22a, HSWX23, MT22]. For example, in [WX22a], Wu and Xue showed that the phenomenon that "most short geodesics are simple" that we alluded to in the previous section actually persists for geodesics on  $X_g$  of length up to  $o(\sqrt{g})$ . This confirmed a conjecture by Lipnowski–Wright [LW23]. They also showed that geodesics whose length is significantly more than  $\sqrt{g}$  tend not to be simple. More recently, in [DS23], Dozier–Sapir showed that for this last bit, a uniform spectral gap and injectivity alone are enough and randomness is not necessary.

Wu-Xue [WX22b] and Lipnowski–Wright [LW23] independently showed that for any  $\varepsilon > 0$ , as  $g \to \infty$ ,  $\lambda_1(X_g) > \frac{3}{16} - \varepsilon$ . Hide [Hid22] showed explicit lower bounds for random surfaces with  $o(\sqrt{g})$  cusps. More recently, Anantharaman–Monk [AM23] proved that in fact  $\lambda_1(X_g) > \frac{2}{9} - \varepsilon$  with high probability as  $g \to \infty$ . Moreover, they have announced that in future work, they will be able to push this up to  $\frac{1}{4} - \varepsilon$ . Further spectral statistics were determined in [Mon22, Nau23, Rud23, RW23] and delocalization results on the eigenfunctions of the Laplacian on  $X_g$  were proved in [LMS17, Tho21, GLMST21].

Finally, we note that the geometry and spectrum of random surfaces of bounded genus with many cusps is wildly different. We refer to [SW23, HT22, Bud22, BZ23] for more

information.

# 5.2 Random 3-manifolds

The first two models of random surfaces discussed above – random Belyĭ surfaces and random finite degree covers – have natural analogues in dimension 3. Because of Mostow rigidity, the model based on the Weil–Petersson measure does not.

However, both models we get are currently intractable. The analogue to random Belyĭ surfaces would be 3-manifolds obtained by randomly gluing a finite number of tetrahedra together along their faces. The problem is that the probability that the resulting complex is a manifold tends to zero as the number of tetrahedra grows [DT06, Proposition 2.8]. One could of course condition on the complex being a manifold. In any event, it's hard to avoid having to count the number of triangulated manifolds and the best bounds we have on these numbers are not precise enough for our purposes (see for instance [CP21]). The analogue of the random cover model: taking a random finite degree cover of a fixed hyperbolic 3-manifold of finite volume has the same issue, namely that we currently don't have good bounds on the subgroup growth of the fundamental groups of these manifolds (see Section 4.1).

In this section, we will discuss three alternative models. First we will describe Dunfield and Thurston's model of random Heegaard splittings. We will in particular describe our results on homological torsion with Hyungryul Baik, David Bauer, Ilya Gekhtman, Ursula Hamenstädt, Sebastian Hensel, Thorben Kastenholz and Daniel Valenzuela [BBG<sup>+</sup>18]. After this, we will describe a class of non-hyperbolic 3-manifolds for which we do have sufficiently sharp bounds on subgroup growth to carry out the study of random finite degree covers, namely torus knot complements. This is joint work with Elizabeth Baker [BP23a]. The third and final model we'll discuss is one that we studied with Jean Raimbault [PR22] and gives rise to random hyperbolic manifolds with boundary.

# 5.2.1 Random Heegaard splittings

**Definitions.** Dunfield and Thurston's solution to the problem above was to consider random Heegaard splittings. These are defined as follows. A (3-dimensional) **handle body** is a 3-manifold with boundary obtained by gluing a finite number of 1-handles (copies of  $[0,1] \times \mathbb{D}$ , where  $\mathbb{D}$  denotes the 1-dimensional disk) to a 3-ball  $\mathbb{B}$  (along  $\{0\} \times \mathbb{D}$  and



Figure 5.2: A Heegaard splitting

 $\{1\} \times \mathbb{D}$ ). Alternatively, these can be thought of as the closed regular neighborhood of a finite graph in  $\mathbb{R}^3$ . Given a handle body H, its boundary  $\partial H$  is a closed surface, the **genus** of H is the genus of  $\partial H$ .

If  $H_1$  and  $H_2$  are handle bodies and  $\varphi : \partial H_1 \to \partial H_2$  is an orientation reversing diffeomorphism, then

$$M_{\varphi} = (H_1 \sqcup H_2) / (x \in \partial H_1 \sim \varphi(x) \in \partial H_2)$$

is a closed orientable 3-manifold (see Figure 5.2 for a schematic picture). The decomposition of  $M_{\varphi}$  is what is called a **Heegaard splitting**. The genus of the Heegaard splitting is the genus of the handle bodies involved.

Moise proved that every closed 3-manifold can be triangulated [Moi52]. It follows from this that every closed 3-manifold admits a Heegaard splitting. In fact, it admits infinitely many non-isomorphic Heegaard splittings. The minimal genus among these splitting is called the **Heegaard genus** of the manifold.

An important observation for us is that if  $\varphi_1 : \partial H_1 \to \partial H_2$  and  $\varphi_2 : \partial H_1 \to \partial H_2$  are isotopic, then  $M_{\varphi_1}$  and  $M_{\varphi_2}$  are diffeomorphic. In other words,  $M_{\varphi}$  is determined by the image of  $\varphi$  in the mapping class group<sup>5</sup> MCG( $\partial H$ ). The mapping class group MCG( $\Sigma_g$ ) is a finitely generated group, so we obtain a notion of random manifolds by performing a random walk on it. That is, we fix some probability measure  $\mu$  on MCG( $\Sigma_g$ ) of finite support<sup>6</sup> and let  $\varphi_n$  denote the random mapping class distributed according to the *n*fold convolution  $\mu^{*n}$ . We will assume that the support of  $\mu$  generates a non-elementary

<sup>&</sup>lt;sup>5</sup>Technically we've identified two sets of mapping classes here:  $\varphi$  was supposed to be orientation reversing rather than orientation preserving and go between two distinct (but diffeomorphic) surfaces.

<sup>&</sup>lt;sup>6</sup>This condition is stricter than typically necessary, usually a finite first moment type condition suffices. However not to get into unnecessary technical difficulties, we'll stick to finite support.

subgroup of  $MCG(\Sigma_g)$ . The associated **random Heegaard splitting** is the manifold  $M_{\varphi_n}$ .

On a side note, we mention there is a second way to construct a manifold out of  $\varphi_n$ , namely we can consider the associated mapping torus. This model has also been studied and it turns out that geometrically the resulting manifolds are very similar to random Heegaard splittings (see for instance [Via21]).

**Geometry and topology.** Dunfield and Thurston introduced random Heegaard splittings in [DT06] to study the virtual Haken conjecture (which at that point had not yet been proved). They ended up proving a negative result: among abelian covers of a random Heegaard splitting  $M_{\varphi_n}$  of genus 2, Haken covers are rare. They also conjectured that  $M_{\varphi_n}$ should be hyperbolic with high probability as  $n \to \infty$ , which was later proved by Maher [Mah10]. By Mostow rigidity, the resulting metric is determined by the topology (so  $\varphi_n$ ) alone, which allows one to ask what its geometry is like.

Viaggi proved that the volume of  $M_{\varphi_n}$  satisfies a law of large numbers: it grows linearly in n, at a rate determined uniquely by the measure  $\mu$  driving the random walk. Feller– Sisto–Viaggi [FSV22] proved bounds on their injectivity radius and diameter and found a new proof of the fact that they're hyperbolic that avoids the use of the Perelman's hyperbolization theorem. Finally, Hamenstädt–Viaggi [HV22] proved that the first nonzero eigenvalue of their Laplacian decays quadratically in their volume (see [BGH20] for the case of random mapping tori). In particular, random Heegaard splittings have very different spectral properties from random graphs and random surfaces.

The drawback of both random Heegaard splittings and random mapping tori is that in the end, we're not sampling from all manifolds after all. We once and for all fix the genus of the handle body and the randomness only comes from the mapping class. In particular, we fix an a priori upper bound on the Heegaard genus of the manifolds we consider. This in turn also implies a global upper bound on the injectivity radius of our manifolds [Whi02], i.e. even at uniformly bounded scales, the geometry of  $M_{\varphi_n}$  is different from  $\mathbb{H}^3$  at every point. This for instance means that, unlike random regular graphs and all the models of random surfaces we have discussed above, the manifolds  $M_{\varphi_n}$  do not Benjamini–Schramm converge to  $\mathbb{H}^3$  (see [ABB<sup>+</sup>17] for definitions).

**Homological torsion.** Nonetheless, random Heegaard splittings can be used for different purposes. They for instance form a good testing ground for conjectures that are still out of reach for all manifolds.

Together with Hyungryul Baik, David Bauer, Ilya Gekhtman, Ursula Hamenstädt, Sebastian Hensel, Thorben Kastenholz and Daniel Valenzuela [BBG<sup>+</sup>18], we studied the growth of torsion in homology in towers of covers of random 3-manifolds. Before we state our results, we start with some context.

Given a closed 3-manifold M, let  $H_1(M;\mathbb{Z})_{\text{tors}}$  denote the torsion subgroup of the integral first homology group  $H_1(M;\mathbb{Z})$ . Lück conjectured the following<sup>7</sup> (see for instance [Lüc13, Conjecture 1.12(2)] and also [BV13, Conjecture 1.3] for a restricted version):

### Conjecture (Lück)

If  $M = \Gamma \setminus \mathbb{H}^3$  is a closed hyperbolic 3-manifold and  $(\Gamma_i \lhd \Gamma)_{i=1}^{\infty}$  is a sequence of finite index normal subgroups such that

$$\bigcap_{i=1}^{\infty} \Gamma_i = \{e\},$$

then

$$\lim_{i \to \infty} \frac{\log \left( |H_1(\Gamma_i \setminus \mathbb{H}^3; \mathbb{Z})_{\text{tors}}| \right)}{[\Gamma : \Gamma_i]} = \frac{\operatorname{vol}(M)}{6\pi}.$$

The reason that the constant  $\frac{1}{6\pi}$  shows up in this conjecture relates to  $\ell^2$ -torsion of hyperbolic manifolds. The idea is that the normalized homological torsion of the covers should approximate the  $\ell^2$ -torsion of the manifold, in a similar way to how normalized Betti numbers  $b_k(\Gamma_i \setminus \mathbb{H}^3; \mathbb{Z}) / [\Gamma : \Gamma_i]$  approximate the  $\ell^2$ -Betti numbers of M, by Lück's approximation theorem [Lüc94]. It's also known that [Kam18], if true, the conjecture above implies the conjecture we briefly discussed in Section 4.1: the volume of a hyperbolic 3manifold is a profinite invariant.

Lê [Lê18] proved that the limit in the conjecture is always bounded from above by  $\operatorname{vol}(M)/6\pi$ . On the other hand, currently not a single example of a hyperbolic 3-manifold and a sequence of normal subgroups with trivial intersection that exhibits exponential torsion growth (let alone at the predicted rate) is known. The best evidence available is a result by Sun [Sun15] proving that, given a closed hyperbolic manifold M, we can make any finite abelian group appear as a summand in  $H_1(\widehat{M};\mathbb{Z})$  for some regular cover  $\widehat{M} \to M$  of finite degree.

On the other hand, if we relax the conditions, much more is known. First of all,

 $<sup>^7\</sup>mathrm{L\""uck}$  's conjecture is stated in a much more general setting, but we will stick to hyperbolic 3-manifolds here.

Liu [Liu19] proved that every closed hyperbolic 3-manifold  $M = \Gamma \setminus \mathbb{H}^3$  admits a nested sequence of subgroups  $(\Gamma_i < \Gamma)_{i=1}^{\infty}$  with trivial intersection such that the torsion in their first homology grows exponentially. However, these subgroups are not normal and the corresponding covers do not Benjamini–Schramm converge to  $\mathbb{H}^3$ . In particular, there is no reason to expect that the exponential growth rate equals the one in the conjecture.

The first examples of exponential torsion growth in general were given by sequences of cyclic (or more generally, abelian) covers. So, this comes down to letting go of the condition that the intersection of the subgroups is trivial. In this case, the exponential growth rate of the torsion in the homology of the covers is determined by the Mahler measure of the Alexander polynomial, namely the torsion in the sequence grows exponentially if and only if the multiplicative Mahler measure of the Alexander polynomial is different from 1 [Ril90, GAnS91, SW02b, SW02a, Rai12, Le14]. Note that in order for a manifold to have an infinite sequence of abelian covers, its first Betti number needs to be positive.

Our paper [BBG<sup>+</sup>18] studies exponential torsion growth for random Heegaard splittings. We study two problems: the growth of torsion in homology as a function of the step length of the random walk and the existence of a sequence of cyclic covers with exponential torsion growth. We prove:

**Theorem 5.6.** (a) Suppose the support of the measure  $\mu$  generates  $MCG(\Sigma_g)$ , then there exists a constant  $\alpha_{\mu} > 0$  such that

$$\frac{\log\left(|H_1(M_{\varphi_n};\mathbb{Z})_{\text{tors}}|\right)}{n} \longrightarrow \alpha_{\mu}$$

in probability as  $n \to \infty$ .

(b) Suppose the support of the measure  $\mu$  generates  $\mathcal{I}(\Sigma_g)$  then  $M_{\varphi_n}$  admits a sequence of cyclic covers with exponential torsion growth with high probability as  $n \to \infty$ .

In item (b) above, the group  $\mathcal{I}(\Sigma_g)$  is the **Torelli subgroup** of MCG( $\Sigma_g$ ), defined by

$$\mathcal{I}(\Sigma_q) \coloneqq \{\varphi \in \mathrm{MCG}(\Sigma_q); \, \varphi_* = \mathrm{Id} : H_1(\Sigma_q; \mathbb{Z}) \to H_1(\Sigma_q; \mathbb{Z})\},\$$

the subgroup of mapping classes that act trivially on homology. Randomly walking on this group instead of the full mapping class group ensures that the first Betti number of  $M_{\varphi_n}$  is positive, in fact, a Mayer–Vietoris argument shows that it's at least g. On the other hand, if we randomly walk on the full mapping class group, the probability that the first Betti number is positive is exponentially small [Kow08, Proposition 7.19]. Item (a) answers a question by Kowalski, who had already proved that homological torsion grows super polynomially fast [Kow08, Section 7.7].

The main idea behind our proofs is to translate these questions into questions about random walks on linear groups. For item (a), we first observe that, by a Mayer–Vietoris argument

$$H_1(M_{\varphi};\mathbb{Z}) = H_1(\Sigma_g;\mathbb{Z}) / \langle L, \varphi_*L \rangle, \qquad (5.2.1)$$

where  $\varphi_* : H_1(\Sigma_g; \mathbb{Z}) \to H_1(\Sigma_g; \mathbb{Z})$  is the map  $\varphi$  induces on homology and L is a Lagrangian in  $H_1(\Sigma_g; \mathbb{Z})$  given by

$$L = \ker \left( i_* : H_1(\Sigma_g; \mathbb{Z}) \to H_1(H; \mathbb{Z}) \right),$$

where  $i: \Sigma_g \to H$  is identification of  $\Sigma_g$  with the boundary  $\partial H$  of our handle body H.

The action homology gives rise to a linear representation to a symplectic group

$$MCG(\Sigma_g) \longrightarrow Sp(2g, \mathbb{Z}).$$

So our random walk descends to a random walk  $((\varphi_n)_*)_n$  on  $\operatorname{Sp}(2g, \mathbb{Z})$ . Looking at (5.2.1), asking for a positive first Betti number is asking that the vector spaces  $(\varphi_n)_*(L \otimes \mathbb{R})$  and  $L \otimes \mathbb{R}$  intersect non-trivially, which turns out to be non-generic. The torsion in  $H_1(M_{\varphi_n}; \mathbb{Z})$  can be computed as the determinant of a block in the matrix  $(\varphi_n)_*$ , which grows exponentially. Both of these are results from random walks on linear groups [GM89, BQ14, BQ16].

To prove item (b), in which case the symplectic representation doesn't "see" anything, we use work of Looijenga [Loo97]. Looijenga shows that a linear representations of the mapping class group on the homology of abelian covers do not annihilate the Torelli group. To control torsion growth, we use the earlier mentioned results that relate this to the Mahler measure of the Alexander polynomial.

# 5.2.2 Random covers of torus knot complements

We have already mentioned before that very little is known about the subgroup growth of lattices in  $PSL(2, \mathbb{C})$ . However, there are (non-hyperbolic) 3-manifolds for which something can be said, namely Seifert fibered manifolds. The subgroup growth of orientable circle bundles over surfaces was determined by Liskovets and Mednykh [LM00] and the subgroup growth of Euclidean manifolds can be derived from general results on the subgroup growth of virtually abelian groups [dSMS99, Sul16].

Together with Elizabeth Baker [BP23a], We studied groups of the form

$$\Gamma_{p_1,\dots,p_m} = \langle x_1,\dots,x_m | x_1^{p_1} = x_2^{p_2} = \dots = x_m^{p_m} \rangle.$$

When gcd(p,q) = 1 and  $p,q \ge 2$  then  $\Gamma_{p,q}$  is the fundamental group of a (p,q)-torus knot complement. More generally,  $\Gamma_{p_1,\dots,p_m}$  is a central extension of the form

$$1 \longrightarrow \mathbb{Z} \longrightarrow \Gamma_{p_1,\dots,p_m} \xrightarrow{\Phi_{p_1,\dots,p_m}} C_{p_1} * \dots * C_{p_m} \longrightarrow 1, \qquad (5.2.2)$$

where  $C_p$  denotes the finite cyclic group of order p and  $\Phi_{p_1,...,p_m}$  is the map that sends the generator  $x_j$  to a generator of  $C_{p_j}$ .

We will consider the case in which

$$-m + 1 + \sum_{i=1}^{m} \frac{1}{p_i} < 0.$$

This includes all torus knot groups. In this setting  $\Gamma_{p_1,\ldots,p_m}$  appears as a non-uniform lattice in  $PSL(2,\mathbb{R}) \times \mathbb{R}$  (see for instance [Eck04, Proposition 7.2]).  $\Phi_{p_1,\ldots,p_m}(\Gamma_{p_1,\ldots,p_m})$  is then the projection onto  $PSL(2,\mathbb{R})$  of  $\Gamma_{p_1,\ldots,p_m}$  and is a Fuchsian group that acts on the hyperbolic plane  $\mathbb{H}^2$ . The displayed expression above is the orbifold Euler characteristic of  $\Phi_{p_1,\ldots,p_m}(\Gamma_{p_1,\ldots,p_m}) \setminus \mathbb{H}^2$ .

**Subgroup growth.** Recall that  $s_n(\Gamma_{p_1,\ldots,p_m})$  denotes the number of subgroups of index n in  $\Gamma_{p_1,\ldots,p_m}$ . Because  $\Gamma_{p_1,\ldots,p_m}$  is a central extension of a free product of finite groups by  $\mathbb{Z}$ , a standard argument, combined with results due to Müller [Mül96], Volynets [Vol86] and Wilf [Wil86] on the subgroup growth on free products of finite groups proves the following:

**Theorem 5.7.** Let  $p_1, \ldots, p_m \in \mathbb{N}_{>1}$  such that  $\sum_{j=1}^m \frac{1}{p_j} < m-1$ . Then it holds that

$$s_n(\Gamma_{p_1,...,p_m}) \sim A_{p_1,...,p_m} \cdot n^{-1/2} \cdot \exp\left(\sum_{\substack{i=1 \ 0 < j < p_i \\ s.t. \ j|p_i}}^m \frac{n^{j/p_i}}{j}\right) \cdot \left(\frac{n}{e}\right)^{n \cdot \left(m - 1 - \sum_{i=1}^m \frac{1}{p_i}\right)}$$

as  $n \to \infty$ , where

$$A_{p_1...,p_m} = \sqrt{2\pi} \exp\left(-\sum_{i: \ p_i \ even} \frac{1}{2p_i}\right) \prod_{i=1}^m p_i^{-1/2}$$

**Random subgroups and covers.** The asymptotic equivalent in Theorem 5.7 allows us to study a uniformly random subgroup

$$H_n < \Gamma_{p_1,\dots,p_m}.$$

This is an example of an Invariant Random Subgroup (IRS) – i.e. a conjugation invariant Borel measure on the Chabauty space of subgroups of  $\Gamma_{p_1,\dots,p_m}$  (see Section 4.1.1).

In order to make topological statements, we will also fix a classifying space  $X_{p_1,\ldots,p_m}$ for  $\Gamma_{p_1,\ldots,p_m}$ . For instance, if  $p, q \ge 2$  and gcd(p,q) = 1 we can take the corresponding torus knot complement. More generally, since  $\Gamma_{p_1,\ldots,p_m}$  appears as a torsion-free lattice in PSL $(2, \mathbb{R}) \times \mathbb{R}$ , we may take the manifold  $\Gamma \setminus (\mathbb{H}^2 \times \mathbb{R})$ .  $H_n$  gives rise to a random degree n cover of  $X_{p_1,\ldots,p_m}$ .

First of all, we determined the limit of the random subgroup  $H_n$ :

**Theorem 5.8.** As  $n \to \infty$ ,  $H_n$  converges to

$$L_{p_1,\dots,p_m} := \ker(\Phi_{p_1,\dots,p_m}) \simeq \mathbb{Z}.$$

as an IRS.

For a (p,q) torus knot,  $L_{p,q}$  is the subgroup generated by the longitude. Using results by Elek [Ele10, Lemma 6.1] and Lück [Lüc94], this implies the following corollary on the growth of the Betti numbers of these random subgroups:

Corollary 5.9. We have that

$$\lim_{n \to \infty} \frac{b_k(H_n; \mathbb{R})}{n} = \begin{cases} m - 1 - \sum_{i=1}^m \frac{1}{p_i} & \text{if } k = 1, 2\\ 0 & \text{otherwise.} \end{cases}$$

in probability.

Finally, we studied the problem of curve counting. To this end, given a conjugacy class  $K \subset \Gamma_{p_1,\dots,p_m}$ , we will write

$$Z_K(H_n) = |\{\gamma H_n \in \Gamma_{p_1,\dots,p_m}/H_n; g \cdot \gamma H_n = \gamma H_n\}|$$

where  $g \in K$  is any element. In topological terms, K corresponds to a free homotopy class of loops in  $X_{p_1,\ldots,p_m}$  and  $Z_K(H_n)$  is the number of closed lifts of that loop to the cover of  $X_{p_1,\ldots,p_m}$  corresponding to  $H_n$ . We obtained the following Poisson approximation for these random variables:

**Theorem 5.10.** Let  $p_1, \ldots, p_m \in \mathbb{N}_{>1}$  be such that  $\sum_{j=1}^m \frac{1}{p_j} < m-1$ . If for all  $g \in K_i$ , for all  $i = 1, \ldots, r$ , the image  $\Phi_{p_1, \ldots, p_m}(g)$  is either trivial or of infinite order, then, as  $n \to \infty$ , the random variables  $Z_{K_i}(H_n)$ ,  $i = 1, \ldots, r$  are asymptotically independent. Moreover,

• if  $K_i \subset L_{p_1,\dots,p_m}$  then

$$\lim_{n \to \infty} \mathbb{P}\left( Z_{K_i}(H_n) = n \right) = 1$$

- and if K<sub>i</sub> ∉ L<sub>p1,...,pm</sub> is the conjugacy class of the k<sup>th</sup> power of a primitive element g<sub>0</sub> then Z<sub>Ki</sub>(H<sub>n</sub>) converges in distribution to a random variable Z<sup>∞</sup><sub>Ki</sub>.
  - If  $\Phi_{p_1,\dots,p_m}(g_0)$  is not a product of two elements of order two,

$$Z_{K_i}^{\infty} \sim \sum_{d|k} d \cdot X_{1/d}$$

where  $X_{1/d} \sim \text{Poisson}(1/d)$  and  $X_1, \ldots, X_{1/k}$  are independent.

- If  $\Phi_{p_1,\dots,p_m}(g_0)$  is a product of two elements of order two,

$$Z_{K_i}^{\infty} \sim \sum_{d|k} 2d \cdot X_{1/2d}^{d,1} + \sum_{d|k, \ d \ even} d \cdot X_{1/2}^{d,2} + d \cdot X_{1/2}^{d,3} + \sum_{d|k, \ d \ odd} d \cdot X_1^{d,4}$$

where  $X_{1/2d}^{d,1} \sim \text{Poisson}(1/2d)$ ,  $X_{1/2}^{d,2}$ ,  $X_{1/2}^{d,3} \sim \text{Poisson}(1/2)$ ,  $X_1^{d,4} \sim \text{Poisson}(1)$  and all these variables are independent.

The fact that products of two involutions play a special role in this theorem was first observed by Doron Puder and Tomer Zimhoni, who noticed an error in a previous version of our paper. They also generalized some of the Poisson approximations we proved on the way to the one stated above [PZ22].

For the conjugacy classes that were left out of the previous statement –those that project to finite order elements in the free product of cyclic groups– we proved:

**Theorem 5.11.** Let  $p_1, \ldots, p_m \in \mathbb{N}_{>1}$  be such that  $\sum_{j=1}^m \frac{1}{p_j} < m-1$ . If the images of the elements of  $K_i$  under  $\Phi_{p_1,\ldots,p_m}$  have order  $k_i \in \mathbb{N}$  for  $i = 1, \ldots, r$ , then the vector of random variables

$$\left(\frac{Z_{K_1}(H_n) - n^{1/k_1} - \varepsilon_1 \cdot n^{1/2k_1}}{\sqrt{p_{j_1}/k_1} \cdot n^{1/2k_1}}, \dots, \frac{Z_{K_r}(H_n) - n^{1/k_r} - \varepsilon_r \cdot n^{1/2k_r}}{\sqrt{p_{j_r}/k_r} \cdot n^{1/2k_r}}\right)$$

converges in distribution to a  $\mathcal{N}(0,1)^{\otimes r}$ -distributed random variable as  $n \to \infty$ . Here  $p_{j_i} \in \mathbb{N}$  is such that  $\Phi_{p_1,\ldots,p_m}(K_i)$  is the conjugacy class of  $x_{j_i}^{l_i}$ , for  $i = 1,\ldots,r$ . Finally  $\varepsilon_i$  equals 1 if  $p_{j_i}/k_i$  is even and 0 otherwise.

The idea behind the proofs of our results on random subgroups is to first prove the analogous results for random index n subgroups of  $C_{p_1} * \cdots * C_{p_m}$  and then use the fact that most index n subgroups of  $\Gamma_{p_1,\ldots,p_m}$  come from index n subgroups of  $C_{p_1} * \cdots * C_{p_m}$  to upgrade these into results about  $\Gamma_{p_1,\ldots,p_m}$ .

First, we consider the problem of counting the number of fixed points of an element  $g \in C_{p_1} * \cdots * C_{p_m}$  under a random homomorphism  $C_{p_1} * \cdots * C_{p_m} \to \mathfrak{S}_n$ . There are two cases to consider:

- Müller–Schlage-Puchta [MSP04, MSP10] proved the central limit theorem we need for finite order elements.
- If g is of infinite order, its number of fixed points can be approximated by a sum of multiples of Poisson-distributed random variables. Like in Section 5.1.3, we prove this by estimating the factorial moments of the random variables that count the fixed points of g.

To turn these results into statements about  $\Gamma_{p_1,\ldots,p_m}$  instead of  $C_{p_1} * \cdots * C_{p_m}$ , we use the fact that asymptotically most actions factor through the projection  $\Phi_{p_1,\ldots,p_m}$  (this is essentially Theorem 5.7). To promote them to statements on subgroups, we use Lemma 4.1.

Finally, the fact that a conjugacy class  $K \subset \Gamma_{p_1,\dots,p_m}$  typically has very few lifts to  $H_n$ if it does not lie in  $L_{p_1,\dots,p_m}$  and typically has n lifts if it does (this is essentially the content of Theorems 5.10 and 5.11), implies that  $H_n$  converges to  $L_{p_1,\dots,p_m}$  (Theorem 5.8).

### 5.2.3 Random 3-manifolds with boundary

The final model of random 3-manifolds we will discuss is a modification of the model of random gluings of tetrahedra. We noted above that the probability that a gluing of n tetrahedra along their faces is a manifold tends to 0 as  $n \to \infty$ . However, the only points in the resulting complex that do not have neighborhoods that are homeomophic to an open set in  $\mathbb{R}^3$  are the vertices in the complex.

As such, one obtains a random manifold with boundary by truncating the tetrahedra at their vertices and gluing the resulting polytopes (see Figure 5.3) at random along their

hexagonal faces.



Figure 5.3: A truncated tetrahedron.  $M_n$  is built by randomly gluing *n* copies of this polyhedron together along their hexagonal faces.

It can be derived from Moise's theorem [Moi52] that all compact 3-manifolds with boundary admit a decomposition into a finite number of copies of this polytope. The question of studying this model has been evoked before (see for instance [DHM15, Question 6.2]) but we aren't aware of any prior work on it.

Together with Jean Raimbault [PR22], we investigated the geometry and topology of these random manifolds. Formally, we let  $M_n$  denote a random gluing of n truncated tetrahedra along their hexagonal faces, obtained by using the configuration model to pair the faces (conditioned on not having loops and multiple edges in the dual graph) and then by gluing each pair of matched faces with one of the three possible orientation reversing simplicial gluings.

**Results.** The topological properties of the random manifolds we obtain are as follows:

**Theorem 5.12** (Topology). (a) We have

$$\lim_{n \to \infty} \mathbb{P}\left(M_n \text{ is connected and has a single boundary component}\right) = 1$$

(b) The genus  $g(\partial M_n)$  of the boundary of  $M_n$  satisfies

$$\lim_{n \to +\infty} \mathbb{P}\left(n - \theta(n) \leq g(\partial M_n) \leq n + 1\right) = 1,$$

for any function  $\theta : \mathbb{N} \to \mathbb{R}$  that grows super-logarithmically<sup>8</sup>.

<sup>8</sup>By this we mean that  $\lim_{n\to\infty} \frac{\theta(n)}{\log(n)} = +\infty$ .

(c) Let  $\mathcal{D} M_n$  denotes the double of  $M_n$  along its boundary and  $g(\mathcal{D} M_n)$  its Heegaard genus. Then

$$\lim_{n \to +\infty} \mathbb{P}\left(n - \theta(n) \leq g(\mathcal{D} M_n) \leq n + \theta(n)\right) = 1,$$

for any function  $\theta : \mathbb{N} \to \mathbb{R}$  that grows super-logarithmically.

(d) There exists C such that the Betti numbers  $b_1(M_n)$  and  $b_1(M_n, \partial M_n)$  satisfy

$$\lim_{n \to +\infty} \mathbb{P}\left(b_1(M_n, \partial M_n) \le \theta(n)\right) = 1, \lim_{n \to +\infty} \mathbb{P}\left(|b_1(M_n) - n| \le \theta(n)\right) = 1$$

for any function  $\theta : \mathbb{N} \to \mathbb{R}$  that grows super-logarithmically.

Our main reason for studying this model is that we hoped that it would give rise to hyperbolic manifolds, which turns out to be the case. Note that it follows from Mostow rigidity that if  $M_n$  carries a hyperbolic metric with totally geodesic boundary, then this metric is unique up to isometry. As such, one can also ask for the geometric properties of this metric. We prove:

Theorem 5.13 (Geometry). We have

 $\lim_{n \to +\infty} \mathbb{P}\left(M_n \text{ carries a hyperbolic metric with totally geodesic boundary}\right) = 1.$ 

This metric has the following properties:

(a) The hyperbolic volume  $vol(M_n)$  of  $M_n$  satisfies:

$$\operatorname{vol}(M_n) \sim n \cdot v_O \quad as \ n \to \infty$$

in probability. Here  $v_O$  denotes the volume of the regular right-angled ideal hyperbolic octahedron.

(b) There exists a constant  $c_{\lambda} > 0$  so that the first discrete Laplacian eigenvalue  $\lambda_1(M_n)$ of  $M_n$  satisfies

$$\lim_{n \to +\infty} \mathbb{P}\left(\lambda_1(M_n) > c_\lambda\right) = 1.$$

(c) There exists a constant  $c_d > 0$  such that the diameter diam $(M_n)$  of  $M_n$  satisfies:

$$\lim_{n \to +\infty} \mathbb{P}\left(\operatorname{diam}(M_n) < c_d \log(\operatorname{vol}(M_n))\right) = 1$$

(d) There exists a constant  $c_s > 0$  such that the systole sys $(M_n)$  of  $M_n$  satisfies:

$$\lim_{n \to +\infty} \mathbb{P}\left(\operatorname{sys}(M_n) > c_s\right) = 1$$

(e) For every  $\varepsilon > 0$ ,

$$\lim_{n \to +\infty} \mathbb{P}\left(\frac{1-\varepsilon}{4n} < \operatorname{sys}(\mathcal{D}\,M_n) < \frac{1}{n^{1-\varepsilon}}\right) = 1.$$

The same holds for the minimal length among arcs in  $M_n$  that are homotopically non-trivial relative to  $\partial M_n$ .

Finally, it turns out that the Benjamini–Schramm limit of  $M_n$  can be identified with a tree of regular right-angled ideal octahedra (the same polytope we discussed in Section 4.1.4) pointed at a uniform random point (this makes sense since this manifold has a cofinite group of isometries).

Methods. In big lines, our proofs consist of two steps. The goal of the first of these is to understand the combinatorics of the cell decomposition of  $M_n$ . Most of this part of the proof is spent on the combinatorics of the interior edges. There are two basic questions: how many interior edges are there, and how many truncated tetrahedra are incident to them. To this end, we let  $E(M_n)$  denote the number of interior edges and  $E_k(M_n)$  the number of interior edges that have k tetrahedra incident to them. In the latter variable, tetrahedra are counted with multiplicity. That is, if an edge appears multiple times in the boundary of a given tetrahedron, this tetrahedron adds to its "length" each time. As such, we have

$$\sum_{k=1}^{6n} k \cdot E_k(M_n) = 6n.$$

To understand these numbers, we use peeling techniques similar to those in Section 5.1.1. We for instance prove that, uniformly for all  $k = o(\sqrt{n})$ ,

$$\mathbb{E}\left(E(M_n)\right) = \frac{1}{2}\log(n) + O(1) \quad \text{and} \quad \mathbb{E}\left(E_k(M_n)\right) = \frac{1}{2k}(1+o(1)) \quad \text{as } n \to \infty.$$

These and the other similar bounds we find are sufficient to prove Theorem 5.12.

The bounds we obtain are less sharp than the analogous bounds for random Belyĭ surfaces (see Section 5.1.1). For instance the following question is open:



Figure 5.4: Three polytopes: a tetrahedron, a truncated tetrahedron and an octahedron

# Question 10

Does the random normalized partition

$$1^{E_{6n}(M_n)} \left(\frac{6n-1}{6n}\right)^{E_{6n-1}(M_n)} \cdots \left(\frac{1}{6n}\right)^{E_1(M_n)}$$

converge to a Poisson–Dirichlet distributed random variable as  $n \to \infty$ ?

The hyperbolization of  $M_n$  is based on the observation that if we contract the interior edges of the cell decomposition, and remove the resulting points, we obtain a non-compact manifold  $X_n$  that is decomposed into octahedra (see Figure 5.4).

The other way around,  $M_n$  can be obtained from  $X_n$  by a Dehn filling (with cylinders). The reason that this is useful is that  $X_n$  admits a complete hyperbolic metric of finite volume with totally geodesic boundary. Indeed, we can equip the octahedra in the decomposition with the metric of a regular right-angled ideal octahedron and perform the gluings with isometries.

There is a lot of work, starting with Thurston's Dehn filling theorem (see for instance [BP92, Chapter E] or [Mar22, Chapter 15]), on the question of when a Dehn filling of a hyperbolic manifold admits a hyperbolic metric and how close this metric is to the original metric. We use Andreev's theorem [RHD07] and recent work by Futer–Purcell–Schleimer [FPS22] on Dehn fillings. This, combined with our combinatorial bounds, implies the manifolds  $M_n$  are hyperbolizable and moreover allows us to estimate their geometric properties.

Further questions and results. More recently, the author's graduate student Anna Roig Sanchis [RS23] proved a similar Poisson approximation theorem to Theorems 5.1 and 5.3 for the primitive length spectrum  $\mathcal{L}(M_n)$  of the random 3-manifolds  $M_n$ :

**Theorem 5.14** (Roig Sanchis). As  $n \to \infty$ ,  $\mathcal{L}(M_n)$  converges locally in distribution to a

Poisson point process  $\mathcal{L}^{3D}$  of computable intensity  $\lambda^{3D}$ .

There are still plenty of elementary geometric and topological questions that remain unanswered by the results above. For example:

### Question 11

(a) Is  $H_1(M_n; \mathbb{Z})_{\text{tors}}$  trivial with high probability as  $n \to \infty$ ? Or a weaker variant: does

$$\frac{\log\left(|H_1(M_n;\mathbb{Z})_{\text{tors}}|\right)}{n} \longrightarrow 0$$

in probability?

- (b) Is  $b_1(M_n; \partial M_n) = 0$  with high probability as  $n \to \infty$ ?
- (c) Do the spectral gaps of  $M_n$  and  $\mathcal{D} M_n$  converge in probability as  $n \to \infty$ ? And if so what are these limits?

# 6 Extremal problems II

This penultimate chapter will be dedicated to application of the probabilistic method – i.e. using probability theory to attack extremal problems – in hyperbolic geometry.

This is especially effective in the study of connectivity properties of hyperbolic manifolds. For example, during the last three years, a lot of progress has been made on questions on the connectivity of closed hyperbolic surfaces of large area. The three most common ways to measure how connected a hyperbolic surface is, are through its diameter, its Cheeger constant and its spectral gap. Some of what we have learned in the last three years is:

- With Thomas Budzinski and Nicolas Curien, we used a random construction to prove that the minimal possible diameter of a hyperbolic surface of genus g is asymptotic to  $\log(g)$  [BCP21b]. Due to the volume entropy of the hyperbolic plane, this is the smallest one can hope for.
- As we mentioned in Section 3.3, using compactifications of random covers of the thrice punctured sphere, Hide and Magee [HM23b] showed that the maximal possible spectral gap of a closed hyperbolic surface of genus g tends to that of the hyperbolic plane (which is <sup>1</sup>/<sub>4</sub>) as g → ∞, thus resolving a longstanding conjecture (see for instance [BBD88]).
- Contrary to the previous two, it turns out that the maximal possible Cheeger constant of a hyperbolic surface of large area is strictly smaller than that of the hyperbolic plane. With Thomas Budzinski and Nicolas Curien [BCP22], we recently proved that the Cheeger constant of a hyperbolic surface of large area can't be much larger than  $\frac{2}{\pi} \approx 0.63...$  (the Cheeger constant of the hyperbolic plane is 1), confirming a conjecture by Lipnowski–Wright [Wri20].

The goal of this chapter is to describe our joint work with Thomas Budzinski and

Nicolas Curien on diameters and Cheeger constants.

# 6.1 The Bers constant and related problems

But before that, we describe what to the author's best knowledge is the first application of probabilistic methods to extremal problems in hyperbolic geometry: the paper on pants decompositions by Guth–Parlier–Young [GPY11].

Here the **Bers length** of a hyperbolic surface is the minimum over all pants decompositions of that surface of the maximal length among the curves in the pants decomposition. The **Bers constant**  $B_g$  in genus g is the maximum over  $\mathcal{M}_g$  of the Bers length and plays an important role in hyperbolic geometry of surfaces and 3-manifolds. The value of the Bers constant is known only in genus 2 [Gen11]. Buser [Bus10, Chapter 5] provided examples of surfaces whose Bers length is  $\gg \sqrt{g}$  and the current best known upper bound on  $B_g$  is due to Parlier [Par23] and is:

$$B_q \leqslant 4\pi(g-1).$$

Earlier linear bounds were proved in [Bus10, Chapter 5] and [Par14]. Buser conjectured his construction is essentially optimal:

Conjecture (Buser)

The Bers constant satisfies

$$B_g = O(\sqrt{g})$$
 as  $g \to \infty$ 

Guth, Parlier and Young studied the related problem of **total pants length**. From the earlier mentioned bounds on the Bers constant, it follows that every hyperbolic surface admits a pants decomposition of total length  $\leq 12\pi(g-1)^2$ . It's however not at all clear that this is sharp. Using the probabilistic method, they proved a lower bound. Concretely, they proved that for any  $\varepsilon > 0$ ,

$$\mathbb{P}\left(\begin{array}{c} X \in \mathcal{M}_g \text{ admits a pants decomposition} \\ \text{of total length } \leqslant g^{7/6-\varepsilon} \end{array}\right) \longrightarrow 0 \text{ as } g \to \infty$$

where the probability is taken with respect to the Weil–Petersson measure (see Section

#### 6.2. THE DIAMETER

#### 5.1.3). It in particular follows that

$$\liminf_{g \to \infty} \frac{\log \left( \sup_{X \in \mathcal{M}_g} \left\{ \mathrm{MTPL}(X) \right\} \right)}{\log(g)} \in \left[ \frac{7}{6}, 2 \right],$$

where

$$MTPL(X) := \min\left\{\sum_{i} \ell_{\alpha_i}(X); \ (\alpha_i)_i \text{ a pants decomposition of } X\right\}$$

denotes the minimal total pants length of X. The bounds above currently still are the best bounds on the maximal minimal total pants length. Moreover, no explicit constructions of sequences of surfaces whose minimal total pants length grows faster than linearly in their genus is known at the moment.

# 6.2 The diameter

Together with Thomas Budzinski and Nicolas Curien, we studied at hyperbolic surfaces of small diameter. Let us first prove an elementary lower bound (similar to Lemma 3.1) that works in any dimension:

**Lemma 6.1.** Let  $d \ge 2$ . There exists a constant  $c_d > 0$  such that

diam
$$(M) \ge \frac{1}{d-1}\log(\operatorname{vol}(M)) - c_d$$

for all closed hyperbolic d-manifolds M.

*Proof.* Again denote the *R*-ball around a point  $p \in M$  by B(p, R). By definition of the diameter, we have

$$B(p, \operatorname{diam}(M)) = M$$

for any  $p \in M$ . Now the volume of  $B(p, \operatorname{diam}(M))$  is smaller than the volume of a ball of the same radius in  $\mathbb{H}^n$ . So, using the formula of the volume of a ball in  $\mathbb{H}^d$  again, we get

$$\operatorname{vol}(M) = \operatorname{vol}(B(p,\operatorname{diam}(M))) \leq \operatorname{vol}(\mathbb{S}^{n-1}) \cdot \int_0^{\operatorname{diam}(M)} \sinh^{n-1}(t) dt$$

which implies the lemma.

The only dimension in which a better lower bound is known is dimension 2. Bavard [Bav96] proved that the diameter of a closed hyperbolic surface M of genus g satisfies

$$\operatorname{diam}(M) \geqslant \operatorname{arccosh}\left(\frac{1}{\sqrt{3}\tan\left(\pi/(12g-6)\right)}\right).$$

This improves on the lemma above by an additive constant.

Given the bounds above, we say that manifolds with small diameter are manifolds with logarithmic diameter. All the models of random manifolds from the previous chapter, except random Heegaard splittings, yield manifolds with small diameter. However, except in the case of random Belyĭ surfaces, as we will discuss below, it's not clear how close their diameters come to the known lower bounds.

In [BCP21b] we proved that, in dimension 2, the bound from Lemma 6.1 is asymptotically sharp:

**Theorem 6.2.** The minimal diameter among closed hyperbolic surfaces of genus g satisfies

$$\lim_{g \to \infty} \frac{\min \{ \operatorname{diam}(X); X \in \mathcal{M}_g \}}{\log(g)} = 1.$$

# 6.2.1 A model for random surfaces

The proof of this theorem is based on a model of random surfaces that we haven't mentioned yet. First of all, we will write  $P_a$  for the hyperbolic pair of pants all of whose boundary components have length a > 0. The random surface  $S_{g,a}$  will be the the surface obtained by randomly gluing 2g - 2 copies of  $P_a$  together along their boundary components with gluings of twist 0, again using the configuration model (Figure 6.1 shows an example).



Figure 6.1: A random gluing of pairs of pants.

What we really prove with Thomas Budzinski and Nicolas Curien is that for every  $\varepsilon > 0$ there exists a boundary length a > 0 such that

$$\lim_{g \to \infty} \mathbb{P}\left(\operatorname{diam}(S_{g,a}) \leqslant (1+\varepsilon)\log(g)\right) = 1.$$
(6.2.1)

# 6.2.2 Ingredients of the proof

There are two basic inputs, a geometric one and a probabilistic one.

**Geometric input.** The main geometric input comes from counting problems in the hyperbolic plane. Given a discrete group  $\Gamma < \text{Isom}(\mathbb{H}^2)$  and  $x \in \mathbb{H}^2$ , one can ask for the growth, as a function of R, of the function

$$N_R(\Gamma, x) = |\Gamma \cdot x \cap B(x, R)|,$$

where B(x, R) denotes the disk of radius R around x. There exists a lot of literature on this problem (see for instance [GN12, Section 1.3] for an overview). We will be interested be interested in the orbit growth of the group  $\Gamma_a$ : the group generated by the reflections in the three non-consecutive sides of length a/2 of  $H_a$  – a right-angled hexagon in  $\mathbb{H}^2$  with three non-consecutive sides of length a/2 (Figure 6.2). Combining results by Patterson [Pat88]



Figure 6.2: The orbit of  $H_a$  under  $\Gamma_a$ .

and McMullen [McM98], we obtain that for every a > 0 there exist constants  $C_a > 0$  and  $\delta_a \in (0, 1)$  such that

$$N_R(\Gamma_a, x) \sim C_a \cdot e^{\delta_a R}, \quad \text{as } R \to \infty.$$

Moreover,  $\delta_a \to 1$  as  $a \to \infty$ .

The pants tree. We will apply these counting results to a hyperbolic surface of infinite area we will call the pants tree  $T_a$ . This surface is formed by gluing countable many copies of  $P_a$  according to the pattern of a trivalent tree. Figure 6.3 shows an example. We use



Figure 6.3: A pants tree

the orbit counting results from above as follows. Fix a point  $m \in P_a$ . For convenience, we fix this to be a midpoint of one of the copies of  $H_a$  that  $P_a$  is made out of. We will now fix one of the copies of  $P_a$  in  $T_a$  and denote its midpoint by  $m_0$ . Moreover  $N_a(R)$  will denote the number of midpoints in  $T_a$  at distance at most R from  $m_0$ .. Essentially because  $T_a$ consists of two copies of the tiling in Figure 6.2, it follows from the orbit counting results above that

$$N_a(R) \sim C_a e^{\delta_a R}, \quad \text{as } R \to \infty.$$
 (6.2.2)

**Probabilistic input.** First of all observe that there exists a constant  $D_a$ , depending on a only, such that

$$\operatorname{diam}(S_{g,a}) \leq \max \left\{ \operatorname{d}(m,m'); \begin{array}{c} m \text{ and } m' \text{ midpoints of} \\ \text{copies of } P_a \text{ in } S_{g,a} \end{array} \right\} + D_a$$

so it will be sufficient to control the maximal distance between midpoints on  $S_{g,a}$ . For argument's sake, we will imagine  $S_{g,a}$  has the geometry of the pants tree  $T_a$  around every midpoint – this is not quite true, but using peeling techniques we prove that it's close enough to true for the rest of the proof to work. This part of the argument is similar to the proof by Bollobás–Fernandez-de-la-Vega [BFdlV82] of the fact that for a random k-regular  $G_n$  on n vertices diam $(G_n)/\log_{k-1}(n) \to 1$  in probability. Let us sketch the rest of the argument. Pick any pair of midpoints m, m' of  $S_{g,a}$ . Because the geometry is like that of  $T_a$ , the number of midpoints at distance at most Rfrom m (resp. m') is (by (6.2.2))  $N_a(R) \sim C_a e^{\delta_a R}$ . Furthermore let R > 0 be such that

$$N_a(R) > g^{1/2+\varepsilon},$$

This happens when

$$R \approx \frac{1/2 + \varepsilon}{\delta_a} \log\left(\frac{g}{C_a}\right).$$

Now, given a pair of boundary components of copies of  $P_a$ , the probability that they are not glued together is roughly  $\frac{6g-8}{6g-7} = 1 - \frac{1}{6g-7}$ . Falsely imagining these probabilities for different pairs are all independent we get that the probability that none of the pants at distance  $\leq R$  from m are glued to a pair of pants at distance  $\leq R$  from m' is

$$\leq \left( \left( 1 - \frac{1}{6g - 7} \right)^{g^{1/2 + \varepsilon}} \right)^{g^{1/2 + \varepsilon}} = o(g^{-2}) \quad \text{as } g \to \infty$$

summing this over the  $\approx g^2$  pairs of midpoints, we see that

$$\mathbb{P}_g \left( \begin{array}{c} \text{There is a path of length } \leqslant 2R \\ \text{between every pair of midpoints} \end{array} \right) \xrightarrow{g \to \infty} 1$$

So we get for all  $\varepsilon > 0$ :

$$\mathbb{P}_g\left(\operatorname{diam}(S_{g,a}) \leqslant \frac{1+2\varepsilon}{\delta_a} \log\left(\frac{g}{C_a}\right) + D_a\right) \xrightarrow{g \to \infty} 1$$

Since  $\delta_a \to 1$  as  $a \to \infty$ , which proves (6.2.1) and hence Theorem 6.2.

### 6.2.3 Further remarks and questions

The random surfaces  $S_{g,a}$  are somewhat of a funny model for random surfaces. They for instance all come with an order two symmetry, so they aren't very suitable for studying the geometry of a "typical" hyperbolic surface of large genus.

In a companion paper together with Thomas Budzinski and Nicolas Curien [BCP21a], we proved that random Belyĭ surfaces miss the optimal diameter by a factor 2. The fact that their diameters are at least  $2\log(\text{genus})$  follows from the fact that they have multiple large embedded disks, which in turn follows from Gamburd and Chmutov-Pittel's Poisson– Dirichlet approximation theorems (see Section 5.1.1). To prove that the diameter is at most  $(2 + o(1)) \log(\text{genus})$ , we use peeling techniques again. We furthermore conjectured that if one conditions on surfaces whose triangulation has only one vertex, the diameter should be asymptotic to  $\log(\text{genus})$ .

It would be interesting two know the behavior of the diameter for the other models of random manifolds we described in Sections 5.1 and 5.2.3:

### Question 12

(a) Let  $X_g$  and  $Y_n$  denote a Weil–Petersson random surface of genus g and a random cover of degree n of a closed hyperbolic surface  $Y_1$  respectively. Do

$$\frac{\operatorname{diam}(X_g)}{\log(g)} \quad \text{and} \quad \frac{\operatorname{diam}(Y_n)}{\log(n)}$$

converge in probability (as g and n tend to infinity respectively) and if so, what are these limits?

(b) Let  $\mathcal{D} M_n$  denote the double of a random 3-manifold with boundary built out of *n* truncated tetrahedra. Does

$$\frac{\operatorname{diam}(M_n)}{\log(\operatorname{vol}(\mathcal{D}\,M_n))}$$

converge in probability as  $n \to \infty$ ?

We also note that a uniform lower bound on the spectral gap (like those of Magee–Naud–Puder [MNP22] and Anantharaman–Monk [AM23]) and the systole imply a logarithmic upper bound on the diameter [Mag20]. However, this is not enough to get the optimal multiplicative constant, even if the surfaces are Ramanujan. In particular, the fact that random Belyĭ surfaces have diameter  $\sim 2 \log(\text{genus})$  does not imply they are not Ramanujan.

# 6.3 The Cheeger constant

The last connectivity problem we will discuss in this chapter is that of the **Cheeger** constant, that we already briefly discussed in Section 5.1.1. Given a Riemannian *d*-

manifold M, its Cheeger constant is

$$h(M) = \inf \left\{ \frac{\operatorname{vol}_{d-1}(\partial A)}{\operatorname{vol}_d(A)}; \begin{array}{c} A \subset M \text{ compact with smooth} \\ \operatorname{boundary and } \operatorname{vol}_d(A) \leqslant \operatorname{vol}_d(M)/2 \end{array} \right\}$$

It measures how hard it is to cut off a piece of the manifold. It's called the Cheeger constant, because Cheeger [Che70] proved that

$$\lambda_1(M) \ge \frac{h(M)^2}{4}$$

for any complete Riemannian manifold M. In fact, Cheeger stated this inequality only for compact manifolds. His proof however works for manifolds of infinite volume as well. In that case constant functions are not in  $L^2(M)$  and  $\lambda_1(M)$  should be interpreted as the infimum of the spectrum of the Laplacian

$$\lambda_1(M) = \inf \left\{ \frac{\int_M ||\nabla f||^2 \, d \operatorname{vol}_M}{||f||_2^2}; \, f \in C^\infty(M) \cap L^2(M) \right\}.$$

Buser [Bus82] proved a converse to Cheeger's inequality, assuming bounds on the Ricci curvature. Moreover, Brooks proved that if a sequence of closed hyperbolic manifolds has uniformly bounded Cheeger constant and systole, their diameter is automatically logarithmic as a function of their volume [Bro92]. However, just like for the spectral gap, the resulting bound does not yield the optimal multiplicative constant.

The Cheeger constant of  $\mathbb{H}^d$  equals

$$h(\mathbb{H}^d) = d - 1$$

It is not attained by a smooth submanifold but can be approximated by balls of growing radius. In particular, Cheeger's inequality is sharp for  $\mathbb{H}^d$ :  $\lambda_1(\mathbb{H}^d) = h(\mathbb{H}^d)^2/4$ .

The natural extremal question that arises is what the maximal possible Cheeger constant of a closed hyperbolic manifold of large volume is. It already follows from Theorem 3.8 and similar bounds in higher dimension [Che75b] combined with Cheeger's inequality that, if the volume is large, the Cheeger constant cannot be significantly larger than that of the hyperbolic space of the same dimension. There is also a more direct proof of this:

**Lemma 6.3.** For any  $d \ge 2$ , there exists a constant  $c_d > 0$  such that

$$h(M) \leq d - 1 + \frac{c_d}{\log(\operatorname{vol}(M))}$$

for any closed hyperbolic d-manifold M.

*Proof.* It follows from the Kazhdan–Margulis theorem [KM68] that there exists a constant  $r_d > 0$  such that every closed hyperbolic *d*-manifold contains an embedded ball of radius  $r_d$  (see [Mar89, Yam82, Fan15] for effective versions).

Suppose  $x \in M$  is the center of such a ball  $B_{r_d}(x)$ . On the other hand, for any t > 0 such that  $\operatorname{vol}(B_t(x)) \leq \operatorname{vol}(M)/2$   $(t \leq \frac{1}{d-1}\log(\operatorname{vol}(M))$  suffices),

$$\frac{d}{dt}\operatorname{vol}(B_t(x)) \ge h(M) \cdot \operatorname{vol}(B_t(x)).$$

Combining these two, we obtain

$$\operatorname{vol}(B_{r_d}(x)) \cdot \exp\left(h(M) \cdot (t - r_d)\right) \leq \operatorname{vol}(B_t(x))$$
$$\leq \operatorname{vol}(B_t(\widetilde{x})) = \operatorname{vol}(\mathbb{S}^{d-1}) \cdot \int_0^t \sinh^{d-1}(s) ds,$$

where  $\widetilde{x} \in \mathbb{H}^d$  is any point. This implies the lemma.

# 6.3.1 Results

Again, the question is how sharp this bound is. As opposed to the case of the spectral gap and the diameter, the natural conjecture is that this is not sharp. This is the case for the analogous problem for regular graphs [Bol88, BZ02, Alo97] and has been conjectured for surfaces for instance by Lipnowski–Wright [Wri20]. Together with Thomas Budzinski and Nicolas Curien [BCP22], we proved this conjecture for surfaces:

Theorem 6.4. We have

$$\limsup_{g \to \infty} \sup \{h(X); X \in \mathcal{M}_g\} \leqslant \frac{2}{\pi} \approx 0.63 \dots$$

The main point of this result is that there is a gap between the Cheeger constant of  $\mathbb{H}^2$  and that of a closed orientable hyperbolic surface of large genus. There is no reason

to expect that the value we get is optimal. It is however optimal for our method of proof. The theorem also implies that Cheeger's inequality cannot be used to prove that hyperbolic surfaces of large genus are Ramanujan.

Similar gap results were already known for principal congruence covers of  $PSL(2, \mathbb{Z}) \setminus \mathbb{H}^2$ , random covers of the Bolza surface and random Belyĭ surfaces [BZ02, SW21, SW22].

# 6.3.2 Method of proof

Our proof is inspired by Bollobás's proof in the case of regular graphs, so we will briefly discuss this first. If G = (V, E) is a graph, then its Cheeger constant<sup>1</sup> is

$$h(G) = \inf\left\{\frac{|E(A, V - A)|}{|A|}; A \subset V \text{ finite and } |A| \leq |V|/2\right\},\$$

where E(A, V - A) denotes the set of edges connecting A to V - E. There is an analogous theory for the Cheeger constant in graph theory, we refer to [HLW06] for an introduction.

We will denote the infinite k-regular tree by  $\mathbb{T}_k$ . Its Cheeger constant equals  $h(\mathbb{T}_k) = k-2$ . There is an analogous elementary bound to Lemma 6.3 that states that the Cheeger constant of a finite k-regular graph on a large number of vertices cannot be significantly larger than k-2.

Bollobás's [Bol88] idea is that we can do a lot better with a elementary probabilistic argument, that we will paraphrase next. Indeed, let G = (V, E) be a finite k-regular graph. We will write n = |V|. Now randomly color each vertex black or white, each with probability  $\frac{1}{2}$  independently from each other. Let B and W denote the sets of black and white vertices respectively. We observe that |B| and |W| are binomial variables and apply Chebyshev's inequality to them to obtain:

$$\mathbb{P}\left(\left||B| - \frac{n}{2}\right| \ge n^{\frac{1}{2} + \delta}\right) = \mathbb{P}\left(\left||W| - \frac{n}{2}\right| \ge n^{\frac{1}{2} + \delta}\right) \le \frac{1}{4} n^{-2\delta}.$$

Moreover, since an edge is part of E(B, W) if and only if it runs between vertices of different colors,  $\mathbb{E}(|E(B, W)|) = k \cdot n/4$ . So, by Markov's inequality

$$\mathbb{P}\left(\left|E(B,W)\right| \ge (1+\delta) \cdot \frac{k \cdot n}{4}\right) \le \frac{1}{1+\delta}.$$

<sup>&</sup>lt;sup>1</sup>There are multiple options for the isoperimetric constant of a graph. For more information, we refer to [HLW06]

As a result

$$\mathbb{P}\left(\frac{|E(B,W)|}{\min\{|B|,|W|\}} \le \frac{k}{2} \cdot \frac{n}{n-2 n^{\frac{1}{2}+\delta}}\right) \ge 1 - \frac{1}{4} n^{-2\delta} - \frac{1}{1+\delta},$$

thus yielding an upper bound of  $\frac{k}{2} + o(1)$  (as  $|V| \to \infty$ ) on the Cheeger constant, which for large degree is better by a factor of almost 2. Alon [Alo97] later sharpened this and obtained an improvement of the trivial bound in all degrees.

Looking at this proof with hyperbolic surfaces in mind, what we need to be able to do is randomly selecting half the hyperbolic surface X. With Thomas Budzinski and Nicolas Curien, we do this by using a **Poisson–Voronoi tessellation**. This is a random cell decomposition of the surface that is obtained as follows. First of all, we consider a Poisson point process S (see Definition 5.2) whose intensity is of the form  $\lambda \cdot \mu_{\text{area}}$ , where  $\mu_{\text{area}}$ denotes the hyperbolic area measure on X and  $\lambda > 0$ . The associated cell decomposition of X is the **Voronoi decomposition** corresponding to S. The 2-dimensional cells of this decomposition are given by

$$C(s) = \{x \in X; d(x, s) \leq d(x, s') \text{ for all } s' \in \mathcal{S} \setminus \{s\}\}, \quad s \in \mathcal{S}$$

The 1-skeleton of the decomposition consists of points on X that are equidistant to multiple elements of  $\mathcal{S}$ .

So, to randomly split our closed hyperbolic surface X into two sets, we equip it a Poisson–Voronoi tessellation and color each top dimensional cell black or white, both with probability  $\frac{1}{2}$ , independently of all other cells. We call the resulting random subsets  $B_{\lambda}$ and  $W_{\lambda}$  respectively. We then prove a concentration result for min{area $(B_{\lambda})$ , area $(W_{\lambda})$ } that is similar to the one for graphs we proved above. Most of the work goes into proving that

$$\limsup_{\lambda \to 0} \sup_{g \to \infty} \sup_{X \in \mathcal{M}_g} \frac{1}{\operatorname{area}(X)} \mathbb{E}\left(\ell(\partial B_{\lambda})\right) \leq \frac{1}{\pi}.$$

In fact, using results by Isokawa [Iso00] together with the existence of sequences of surfaces whose genus and systole tend to infinity, one can show that the above is an equality (which is why one can't do better than  $\frac{2}{\pi}$  with our proof). Once we have these two bounds, the proof proceeds in the same way as in the graph case.

Finally, there is a particular cell decomposition of the hyperbolic plane that naturally arises from our proof. Indeed, in our proof we let the parameter  $\lambda$  tend to 0, which comes down to decreasing the number of points per unit of area. The quantity  $\frac{1}{\pi}$  that shows
up can be thought of as  $\frac{1}{2}$  times the ratio between the boundary length and the area of a typical cell of a limiting tessellation that we dubbed the **pointless Poisson–Voronoi tessellation** of  $\mathbb{H}^2$ . In this tessellation, the points defining the centers of the cells of this tessellation have converged out of the boundary. The cells themselves however remain visible from any base point in  $\mathbb{H}^2$ . The properties of this tessellation have recently been further studied for hyperbolic spaces in [DCE<sup>+</sup>23] and for symmetric spaces of higher rank in [FMW23, Mel23].

### 6.3.3 Further remarks and questions

Poisson–Voronoi tessellations are a classical object in stochastic geometry, see for instance [Iso00, BS01, HM22, HM23a, OT23, HOOT23] for versions in hyperbolic geometry and [CCE21] in a more general Riemannian geometric setting.

Given what's known for hyperbolic surfaces and graphs, it's natural to conjecture:

### Conjecture

For every  $d \ge 3$ , there exists a number  $\varepsilon_d > 0$  such that

$$h(M) < d - 1 - \varepsilon_d$$

for all closed hyperbolic *d*-manifolds of sufficiently large volume.

In fact, there is no reason that random decompositions based on Poisson–Voronoi tessellations don't work in higher dimension. What's less clear is whether the bound they yield gets better in higher dimension. For instance, is it asymptotic to d/2 as  $d \to \infty$ ?

The distribution of Cheeger constants of random surfaces is also an interesting question:

### Question 13

Is there some h > 0 such that  $h(X) \longrightarrow h$  in probability as  $\operatorname{area}(X) \to \infty$ , for any of the models of random surfaces in Section 5.1 ?

If such a number exists, it would be a natural candidate for the maximal possible Cheeger constant of a hyperbolic surface of large genus.

# 7 Future directions

In the chapters above, we have tried to convince the reader that simple extremal questions connect to many different aspects of hyperbolic geometry. Moreover, in hyperbolic geometry, just like in Euclidean geometry and graph theory, many of the simplest questions – of which we have listed some throughout the text – are still open. In this last chapter, we discuss some future directions of research, based on some of the questions above and other questions like them.

**Systoles.** In Section 3.1, we discussed the maximal possible systole of a hyperbolic manifold of bounded volume. By now, we have plenty of constructions of sequences of closed hyperbolic surfaces whose systoles grow logarithmically as a function of their genus – which is the fastest possible rate. There are arithmetic constructions and more combinatorial constructions, both of which we discussed above. Moreover, in upcoming work with Mingkun Liu [LP23], we will present two random constructions, both inspired by ideas from graph theory [LS21, GHS<sup>+</sup>09].

In higher dimensions, currently only arithmetic examples are known. It would be interesting to have other constructions. An example of a question is where a variant of the construction from Section 3.1.3 can be made to work for 3-manifolds built out of regular right-angled octahedra.

A related, but probably much harder, question is what the systole, and more generally the length spectrum of a random finite degree subgroup of the corresponding reflection group behaves like. This would require deriving much sharper bounds on the subgroup growth of that reflection group than those we found with Hyungryul Baik and Jean Raimbault (see Section 4.1.4). Already proving that a random finite degree cover Benjamini– Schramm converges to  $\mathbb{H}^3$  would be very interesting. If these random orbifolds behave like their 2-dimensional counterparts, then one would expect that their systoles do not tend to infinity. In order to obtain manifolds of large systoles, it would also be interesting study random normal subgroups.

Linear programming. A more manageable project is to generalize our linear programming bounds with Maxime Fortier Bourque, that we discussed in Chapter 3, to (potentially non-compact) hyperbolic orbifolds of finite volume. The only thing that needs to be done is deal with the extra terms that show up in the trace formula. It would for instance be interesting to see how the bound one obtains compares with the growth in the sequences of non-compact manifolds with super-linear kissing number found by Dória–Murillo and Dória–Freire–Murillo [DM21, DFM23] and the bound by Fanoni–Parlier [FP15] in dimension 2.

A much more ambitious (and speculative) question on our linear programming bounds is whether, like in Euclidean geometry, there exist magical functions in some special cases. The bounds we have presented in this text don't make this seem very hopeful: our bounds for hyperbolic surfaces are much further away from the best examples than the Cohn– Elkies bounds on sphere packings were. However, based on recent numerics by Émile Gruda-Médiavilla and Mathieu Pineauilt, it seems like in genus 10 and 17, the bounds might be closer.

**Expansion.** Above, we have discussed three measures of connectivity of hyperbolic manifolds: their diameter, their Cheeger constant and their spectral gap. We have recently learned a lot about these invariants in dimension 2. Indeed, there exist sequences of closed hyperbolic surfaces whose diameter [BCP21b] and spectral gap [HM23b] saturate the classical bounds on these quantities. Moreover, the Cheeger constant behaves differently: a closed hyperbolic surface of large genus has a smaller Cheeger constant than the hyperbolic plane [BCP22] (see Chapter 6).

It would be nice to know what the actual maximum of the Cheeger constant is in large genus. Moreover, we have already mentioned in Section 6.3 that the Cheeger constants of random hyperbolic surfaces of large genus are not yet fully understood. Also for graphs, the analogous questions are still open.

In higher dimensions we know much less. It doesn't seem unlikely that our method with Thomas Budzinski and Nicolas Curien will allow us to prove a similar gap result on the Cheeger constant to the 2-dimensional case. The fact that higher dimensional hyperbolic manifolds are less flexible than surfaces (i.e. they satisfy the Mostow–Prasad rigidity theorem) and that as a result it is much harder to come up with good models of random manifolds, also makes the questions on the diameter and spectral gap harder to approach. One model that does show expansion properties is the model of random 3manifolds with boundary and their (closed) doubles that we studied with Jean Raimbault [PR22]. It would be very interesting get better estimates on their diameter and spectral gap. In fact, there are reasons to believe that their diameter misses the optimal rate by a factor 4 and that their spectral gap is related to that of the Apollonian group (for which very good numerical approximations exist).

## Bibliography

- [AB20] Karim Adiprasito and Bruno Benedetti. A Cheeger-type exponential bound for the number of triangulated manifolds. Ann. Inst. Henri Poincaré D, 7(2):233–247, 2020.
- [ABB<sup>+</sup>17] Miklos Abert, Nicolas Bergeron, Ian Biringer, Tsachik Gelander, Nikolay Nikolov, Jean Raimbault, and Iddo Samet. On the growth of L<sup>2</sup>-invariants for sequences of lattices in Lie groups. Ann. of Math. (2), 185(3):711–790, 2017.
- [ABT00] R. Arratia, A. D. Barbour, and S. Tavaré. Limits of logarithmic combinatorial structures. Ann. Probab., 28(4):1620–1644, 2000.
- [AFW15] Matthias Aschenbrenner, Stefan Friedl, and Henry Wilton. 3-manifold groups. EMS Series of Lectures in Mathematics. European Mathematical Society (EMS), Zürich, 2015.
- [AGG89] R. Arratia, L. Goldstein, and L. Gordon. Two moments suffice for Poisson approximations: the Chen-Stein method. Ann. Probab., 17(1):9–25, 1989.
- [Ago06] Ian Agol. Systoles of hyperbolic 4-manifolds. Preprint, arXiv:math/0612290, 2006.
- [Ago13] Ian Agol. The virtual Haken conjecture. *Doc. Math.*, 18:1045–1087, 2013. With an appendix by Agol, Daniel Groves, and Jason Manning.
- [AGV14] Miklós Abért, Yair Glasner, and Bálint Virág. Kesten's theorem for invariant random subgroups. *Duke Math. J.*, 163(3):465–488, 2014.
- [AK15] J. Azarija and S. Klavžar. Moore graphs and cycles are extremal graphs for convex cycles. J. Graph Theory, 80(1):34–42, 2015.

- [Akr03] H. Akrout. Singularités topologiques des systoles généralisées. *Topology*, 42(2):291–308, 2003.
- [Alo86] N. Alon. Eigenvalues and expanders. volume 6, pages 83–96. 1986. Theory of computing (Singer Island, Fla., 1984).
- [Alo97] Noga Alon. On the edge-expansion of graphs. *Combinatorics, Probability and Computing*, 6(2):145–152, 1997.
- [AM22] Nalini Anantharaman and Laura Monk. A high-genus asymptotic expansion of Weil-Petersson volume polynomials. J. Math. Phys., 63(4):Paper No. 043502, 26, 2022.
- [AM23] Nalini Anantharaman and Laura Monk. Friedman-Ramanujan functions in random hyperbolic geometry and application to spectral gaps. Preprint, arXiv:2304.02678, 2023.
- [AR93] Noga Alon and Yuval Roichman. Random Cayley graphs and expanders (abstract). In Expanding graphs (Princeton, NJ, 1992), volume 10 of DIMACS Ser. Discrete Math. Theoret. Comput. Sci., pages 1–3. Amer. Math. Soc., Providence, RI, 1993.
- [AS03] Omer Angel and Oded Schramm. Uniform infinite planar triangulations. Comm. Math. Phys., 241(2-3):191–213, 2003.
- [AS18] Tarik Aougab and Juan Souto. Counting curve types. Amer. J. Math., 140(6):1423–1441, 2018.
- [Bam54] R. P. Bambah. On lattice coverings by spheres. *Proc. Nat. Inst. Sci. India*, 20:25–52, 1954.
- [Bav96] C. Bavard. Disques extrémaux et surfaces modulaires. Ann. Fac. Sci. Toulouse Math. (6), 5(2):191–202, 1996.
- [BB90] N.L. Biggs and A.G. Boshier. Note on the girth of Ramanujan graphs. J. Combin. Theory Ser. B, 49(2):190–194, 1990.
- [BBD88] Peter Buser, Marc Burger, and Jozef Dodziuk. Riemann surfaces of large genus and large λ<sub>1</sub>. In Geometry and analysis on manifolds (Katata/Kyoto, 1987), volume 1339 of Lecture Notes in Math., pages 54–63. Springer, Berlin, 1988.

- [BBG<sup>+</sup>18] Hyungryul Baik, David Bauer, Ilya Gekhtman, Ursula Hamenstädt, Sebastian Hensel, Thorben Kastenholz, Bram Petri, and Daniel Valenzuela. Exponential torsion growth for random 3-manifolds. Int. Math. Res. Not. IMRN, (21):6497–6534, 2018.
- [BC13] Nicolas Bergeron and Laurent Clozel. Quelques conséquences des travaux d'Arthur pour le spectre et la topologie des variétés hyperboliques. Invent. Math., 192(3):505–532, 2013.
- [BC19] Charles Bordenave and Benoît Collins. Eigenvalues of random lifts and polynomials of random permutation matrices. Ann. of Math. (2), 190(3):811–875, 2019.
- [BC23] Charles Bordenave and Benoit Collins. Norm of matrix-valued polynomials in random unitaries and permutations. Preprint, arXiv:2304.05714, 2023.
- [BCP19] Thomas Budzinski, Nicolas Curien, and Bram Petri. Universality for random surfaces in unconstrained genus. *Electron. J. Combin.*, 26(4):Paper No. 4.2, 34, 2019.
- [BCP21a] Thomas Budzinski, Nicolas Curien, and Bram Petri. The diameter of random Belyĭ surfaces. *Algebr. Geom. Topol.*, 21(6):2929–2957, 2021.
- [BCP21b] Thomas Budzinski, Nicolas Curien, and Bram Petri. On the minimal diameter of closed hyperbolic surfaces. *Duke Math. J.*, 170(2):365–377, 2021.
- [BCP22] Thomas Budzinski, Nicolas Curien, and Bram Petri. On cheeger constants of hyperbolic surfaces. Preprint, arXiv:2207.00469, 2022.
- [Bel79] G. V. Belyĭ. Galois extensions of a maximal cyclotomic field. Izv. Akad. Nauk SSSR Ser. Mat., 43(2):267–276, 479, 1979.
- [Ber11] Nicolas Bergeron. Le spectre des surfaces hyperboliques. Savoirs Actuels (Les Ulis). [Current Scholarship (Les Ulis)]. EDP Sciences, Les Ulis; CNRS Éditions, Paris, 2011.
- [Bes80] G. Besson. Sur la multiplicité de la première valeur propre des surfaces riemanniennes. Ann. Inst. Fourier (Grenoble), 30(1):x, 109–128, 1980.

- [BF20] Michel Boileau and Stefan Friedl. The profinite completion of 3-manifold groups, fiberedness and the Thurston norm. In What's next?—the mathematical legacy of William P. Thurston, volume 205 of Ann. of Math. Stud., pages 21–44. Princeton Univ. Press, Princeton, NJ, 2020.
- [BFdIV82] B. Bollobás and W. Fernandez de la Vega. The diameter of random regular graphs. *Combinatorica*, 2(2):125–134, 1982.
- [BG08] Jean Bourgain and Alex Gamburd. Uniform expansion bounds for Cayley graphs of  $SL_2(\mathbb{F}_p)$ . Ann. of Math. (2), 167(2):625–642, 2008.
- [BGGT15] Emmanuel Breuillard, Ben Green, Robert Guralnick, and Terence Tao. Expansion in finite simple groups of Lie type. J. Eur. Math. Soc. (JEMS), 17(6):1367–1434, 2015.
- [BGH20] Hyungryul Baik, Ilya Gekhtman, and Ursula Hamenstädt. The smallest positive eigenvalue of fibered hyperbolic 3-manifolds. Proc. Lond. Math. Soc. (3), 120(5):704–741, 2020.
- [BGLM02] M. Burger, T. Gelander, A. Lubotzky, and S. Mozes. Counting hyperbolic manifolds. *Geom. Funct. Anal.*, 12(6):1161–1173, 2002.
- [BGLS10] M. Belolipetsky, T. Gelander, A. Lubotzky, and A. Shalev. Counting arithmetic lattices and surfaces. Ann. of Math. (2), 172(3):2197–2221, 2010.
- [BHJ92] A. D. Barbour, Lars Holst, and Svante Janson. Poisson approximation, volume 2 of Oxford Studies in Probability. The Clarendon Press, Oxford University Press, New York, 1992. Oxford Science Publications.
- [BI73] Eiichi Bannai and Tatsuro Ito. On finite Moore graphs. J. Fac. Sci. Univ. Tokyo Sect. IA Math., 20:191–208, 1973.
- [Big98] N. Biggs. Constructions for cubic graphs with large girth. Electron. J. Combin., 5:Article 1, 25, 1998.
- [BLS92] M. Burger, J.-S. Li, and P. Sarnak. Ramanujan duals and automorphic spectrum. Bull. Amer. Math. Soc. (N.S.), 26(2):253–257, 1992.
- [BM04] Robert Brooks and Eran Makover. Random construction of Riemann surfaces. J. Differential Geom., 68(1):121–157, 2004.

#### BIBLIOGRAPHY

- [BM22] Jérémie Bettinelli and Grégory Miermont. Compact brownian surfaces ii. orientable surfaces. Preprint, arXiv:2212.12511, 2022.
- [BMRS20] M. R. Bridson, D. B. McReynolds, A. W. Reid, and R. Spitler. Absolute profinite rigidity and hyperbolic geometry. Ann. of Math. (2), 192(3):679– 719, 2020.
- [Bol80] Béla Bollobás. A probabilistic proof of an asymptotic formula for the number of labelled regular graphs. *European J. Combin.*, 1(4):311–316, 1980.
- [Bol81] Béla Bollobás. Random graphs. In Combinatorics (Swansea, 1981), volume 52 of London Math. Soc. Lecture Note Ser., pages 80–102. Cambridge Univ. Press, Cambridge-New York, 1981.
- [Bol88] Béla Bollobás. The isoperimetric number of random regular graphs. *European Journal of combinatorics*, 9(3):241–244, 1988.
- [Bol01] Béla Bollobás. Random graphs, volume 73 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, second edition, 2001.
- [Bon22] J. Bonifacio. Bootstrapping closed hyperbolic surfaces. J. High Energy Phys.,
  (3):Paper No. 093, 18, 2022.
- [Bor20] Charles Bordenave. A new proof of Friedman's second eigenvalue theorem and its extension to random lifts. Ann. Sci. Éc. Norm. Supér. (4), 53(6):1393– 1439, 2020.
- [Bow14] Lewis Bowen. Random walks on random coset spaces with applications to Furstenberg entropy. *Invent. Math.*, 196(2):485–510, 2014.
- [BP92] Riccardo Benedetti and Carlo Petronio. Lectures on hyperbolic geometry. Universitext. Springer-Verlag, Berlin, 1992.
- [BP22] Maxime Fortier Bourque and Bram Petri. Kissing numbers of closed hyperbolic manifolds. *Amer. J. Math.*, 144(4):1067–1085, 2022.
- [BP23a] Elizabeth Baker and Bram Petri. Statistics of finite degree covers of torus knot complements. Ann. H. Lebesgue, 2023+. to appear.
- [BP23b] Maxime Fortier Bourque and Bram Petri. The Klein quartic maximizes the multiplicity of the first positive eigenvalue of the laplacian. J. Differential Geom., 2023+. available at: arXiv:2111.14699.

- [BP23c] Maxime Fortier Bourque and Bram Petri. Linear programming bounds for hyperbolic surfaces. Preprint, arXiv:2302.02540, 2023.
- [BPR19] Hyungryul Baik, Bram Petri, and Jean Raimbault. Subgroup growth of virtually cyclic right-angled Coxeter groups and their free products. *Combinatorica*, 39(4):779–811, 2019.
- [BPR20] Hyungryul Baik, Bram Petri, and Jean Raimbault. Subgroup growth of rightangled Artin and Coxeter groups. J. Lond. Math. Soc. (2), 101(2):556–588, 2020.
- [BQ14] Yves Benoist and Jean-François Quint. Random walks on projective spaces. Compos. Math., 150(9):1579–1606, 2014.
- [BQ16] Yves Benoist and Jean-François Quint. Random walks on reductive groups, volume 62 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer, Cham, 2016.
- [Bro88] Robert Brooks. Injectivity radius and low eigenvalues of hyperbolic manifolds. J. Reine Angew. Math., 390:117–129, 1988.
- [Bro92] Robert Brooks. Some relations between spectral geometry and number theory.
  In Topology '90 (Columbus, OH, 1990), volume 1 of Ohio State Univ. Math. Res. Inst. Publ., pages 61–75. de Gruyter, Berlin, 1992.
- [Bro99] Robert Brooks. Platonic surfaces. Comment. Math. Helv., 74(1):156–170, 1999.
- [BS94] P. Buser and P. Sarnak. On the period matrix of a Riemann surface of large genus. *Invent. Math.*, 117(1):27–56, 1994. With an appendix by J. H. Conway and N. J. A. Sloane.
- [BS01] Itai Benjamini and Oded Schramm. Percolation in the hyperbolic plane. J. Amer. Math. Soc., 14(2):487–507, 2001.
- [BT11] Mikhail V. Belolipetsky and Scott A. Thomson. Systoles of hyperbolic manifolds. *Algebr. Geom. Topol.*, 11(3):1455–1469, 2011.

- [BU83] S. Bando and H. Urakawa. Generic properties of the eigenvalue of the Laplacian for compact Riemannian manifolds. *Tohoku Math. J. (2)*, 35(2):155–172, 1983.
- [Bud22] Timothy Budd. Irreducible metric maps and Weil-Petersson volumes. Comm. Math. Phys., 394(2):887–917, 2022.
- [Bus77] Peter Buser. Riemannsche Flächen mit Eigenwerten in (0,1/4). Comment. Math. Helv., 52(1):25–34, 1977.
- [Bus78] Peter Buser. Riemannsche Flächen mit grosser Kragenweite. Comment. Math. Helv., 53(3):395–407, 1978.
- [Bus82] Peter Buser. A note on the isoperimetric constant. Ann. Sci. École Norm. Sup. (4), 15(2):213–230, 1982.
- [Bus10] Peter Buser. Geometry and spectra of compact Riemann surfaces. Modern Birkhäuser Classics. Birkhäuser Boston, Ltd., Boston, MA, 2010. Reprint of the 1992 edition.
- [BV13] Nicolas Bergeron and Akshay Venkatesh. The asymptotic growth of torsion homology for arithmetic groups. J. Inst. Math. Jussieu, 12(2):391–447, 2013.
- [BW12] Nicolas Bergeron and Daniel T. Wise. A boundary criterion for cubulation. Amer. J. Math., 134(3):843–859, 2012.
- [BZ02] Robert Brooks and Andrzej Zuk. On the asymptotic isoperimetric constants for Riemann surfaces and graphs. J. Differential Geom., 62(1):49–78, 2002.
- [BZ11] Bruno Benedetti and Günter M. Ziegler. On locally constructible spheres and balls. *Acta Math.*, 206(2):205–243, 2011.
- [BZ23] Timothy Budd and Bart Zonneveld. Topological recursion of the weilpetersson volumes of hyperbolic surfaces with tight boundaries. Preprint, arXiv:2307.04708, 2023.
- [CCdV88] B. Colbois and Y. Colin de Verdière. Sur la multiplicité de la première valeur propre d'une surface de Riemann à courbure constante. Comment. Math. Helv., 63(2):194–208, 1988.

- [CCE21] Pierre Calka, Aurélie Chapron, and Nathanaël Enriquez. Poisson-Voronoi tessellation on a Riemannian manifold. Int. Math. Res. Not. IMRN, (7):5413– 5459, 2021.
- [CD94] B. Colbois and J. Dodziuk. Riemannian metrics with large  $\lambda_1$ . Proc. Amer. Math. Soc., 122(3):905–906, 1994.
- [CdV86] Y. Colin de Verdière. Sur la multiplicité de la première valeur propre non nulle du laplacien. Comment. Math. Helv., 61(2):254–270, 1986.
- [CdV87] Y. Colin de Verdière. Construction de laplaciens dont une partie finie du spectre est donnée. Ann. Sci. École Norm. Sup. (4), 20(4):599–615, 1987.
- [CE03] H. Cohn and N. Elkies. New upper bounds on sphere packings. I. Ann. of Math. (2), 157(2):689–714, 2003.
- [CFP18] Patricia Cahn, Federica Fanoni, and Bram Petri. Mapping class group orbits of curves with self-intersections. *Israel J. Math.*, 223(1):53–74, 2018.
- [Che70] Jeff Cheeger. A lower bound for the smallest eigenvalue of the Laplacian. pages 195–199, 1970.
- [Che75a] Louis H. Y. Chen. Poisson approximation for dependent trials. Ann. Probability, 3(3):534–545, 1975.
- [Che75b] Shiu Yuen Cheng. Eigenvalue comparison theorems and its geometric applications. *Math. Z.*, 143(3):289–297, 1975.
- [Che76] S. Y. Cheng. Eigenfunctions and nodal sets. *Comment. Math. Helv.*, 51(1):43–55, 1976.
- [CHM51] S. Chowla, I. N. Herstein, and W. K. Moore. On recursions connected with symmetric groups. I. *Canadian J. Math.*, 3:328–334, 1951.
- [CK09] H. Cohn and A. Kumar. Optimality and uniqueness of the Leech lattice among lattices. Ann. of Math. (2), 170(3):1003–1050, 2009.
- [CKM<sup>+</sup>17] H. Cohn, A. Kumar, S. D. Miller, D. Radchenko, and M. Viazovska. The sphere packing problem in dimension 24. Ann. of Math. (2), 185(3):1017– 1033, 2017.

- [Coo18] J. Cook. Properties of eigenvalues on Riemann surfaces with large symmetry groups. PhD thesis, Loughborough University, arXiv:2108.11825, 2018.
- [Cou23] R. Courant. Ein allgemeiner satzt zur theorie der eigenfunktionen selbsadjungierter differentialausdrücke. Nachrichten von der Gesellschaft der Wissenschaften zu Göttingen, Mathematisch-Physikalische Klasse, 1923:81–84, 1923.
- [CP16] Sergei Chmutov and Boris Pittel. On a surface formed by randomly gluing together polygonal discs. *Adv. in Appl. Math.*, 73:23–42, 2016.
- [CP21] Guillaume Chapuy and Guillem Perarnau. On the number of coloured triangulations of *d*-manifolds. *Discrete Comput. Geom.*, 65(3):601–617, 2021.
- [CS93] J. H. Conway and N. J. A. Sloane. Sphere packings, lattices and groups, volume 290 of Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, New York, second edition, 1993. With additional contributions by E. Bannai, R. E. Borcherds, J. Leech, S. P. Norton, A. M. Odlyzko, R. A. Parker, L. Queen and B. B. Venkov.
- [Cur23] Nicolas Curien. Peeling random planar maps. available at: https://www.imo.universite-paris-saclay.fr/ nicolas.curien, 2023.
- [CZ14] H. Cohn and Y. Zhao. Sphere packing bounds via spherical codes. Duke Math. J., 163(10):1965–2002, 2014.
- [Dah14] X. Dahan. Regular graphs of large girth and arbitrary degree. *Combinatorica*, 34(4):407–426, 2014.
- [Dam73] R. M. Damerell. On Moore graphs. Proc. Cambridge Philos. Soc., 74:227–236, 1973.
- [DCE<sup>+</sup>23] Matteo D'Achille, Nicolas Curien, Nathanaël Enriquez, Russell Lyons, and Meltem Ünel. Ideal Poisson-Voronoi tessellations on hyperbolic spaces, I. Preprint, arXiv:2303.16831, 2023.
- [DFM23] Cayo Dória, Emanoel M. S. Freire, and Plinio G. P. Murillo. Hyperbolic manifolds with a large number of systoles. *Trans. Amer. Math. Soc.*, to appear, 2023+.

- [DGS77] P. Delsarte, J. M. Goethals, and J. J. Seidel. Spherical codes and designs. *Geometriae Dedicata*, 6(3):363–388, 1977.
- [DGZZ21] Vincent Delecroix, Élise Goujard, Peter Zograf, and Anton Zorich. Masur-Veech volumes, frequencies of simple closed geodesics, and intersection numbers of moduli spaces of curves. *Duke Math. J.*, 170(12):2633–2718, 2021.
- [DGZZ22] Vincent Delecroix, Élise Goujard, Peter Zograf, and Anton Zorich. Large genus asymptotic geometry of random square-tiled surfaces and of random multicurves. *Invent. Math.*, 230(1):123–224, 2022.
- [DHM15] Kelly Delp, Diane Hoffoss, and Jason Fox Manning. Problems in groups, geometry, and three-manifolds. Preprint, arXiv:1512.04620, 2015.
- [Dix69] John D. Dixon. The probability of generating the symmetric group. *Math.* Z., 110:199–205, 1969.
- [DJ95] Bergfinnur Durhuus and Thórdur Jónsson. Remarks on the entropy of 3manifolds. *Nuclear Phys. B*, 445(1):182–192, 1995.
- [DM21] Cayo Dória and Plinio G. P. Murillo. Hyperbolic 3-manifolds with large kissing number. *Proc. Amer. Math. Soc.*, 149(11):4595–4607, 2021.
- [DR63] B. N. Delone and S. S. Ryškov. Solution of the problem on the least dense lattice covering of a 4-dimensional space by equal spheres. *Dokl. Akad. Nauk* SSSR, 152:523–524, 1963.
- [DS81] Persi Diaconis and Mehrdad Shahshahani. Generating a random permutation with random transpositions. Z. Wahrsch. Verw. Gebiete, 57(2):159–179, 1981.
- [DS23] Benjamin Dozier and Jenya Sapir. Counting geodesics on expander surfaces. Preprint, arXiv:2304.07938, 2023.
- [dSMS99] M. P. F. du Sautoy, J. J. McDermott, and G. C. Smith. Zeta functions of crystallographic groups and analytic continuation. *Proc. London Math. Soc.* (3), 79(3):511–534, 1999.
- [DT06] Nathan M. Dunfield and William P. Thurston. Finite covers of random 3manifolds. *Invent. Math.*, 166(3):457–521, 2006.
- [Ebe17] Sean Eberhard. The trivial lower bound for the girth of  $S_n$ . Preprint, arXiv:1706.09972, 2017.

### BIBLIOGRAPHY

[Eck04]	Beno Eckmann. Lattices, $l_2$ -Betti numbers, deficiency, and knot groups. <i>Enseign. Math.</i> (2), 50(1-2):123–137, 2004.
[EJ08]	Geoffrey Exoo and Robert Jajcay. Dynamic cage survey. <i>Electron. J. Combin.</i> , DS16(Dynamic Surveys):48, 2008.
[EJ22]	Sean Eberhard and Urban Jezernik. Babai's conjecture for high-rank classical groups with random generators. <i>Invent. Math.</i> , 227(1):149–210, 2022.
[Ele10]	Gábor Elek. Betti numbers are testable. In <i>Fete of combinatorics and computer science</i> , volume 20 of <i>Bolyai Soc. Math. Stud.</i> , pages 139–149. János Bolyai Math. Soc., Budapest, 2010.
[EPS20]	Viveka Erlandsson, Hugo Parlier, and Juan Souto. Counting curves, and the stable length of currents. J. Eur. Math. Soc. (JEMS), 22(6):1675–1702, 2020.
[ES63]	P. Erdős and H. Sachs. Reguläre graphen gegebene taillenweite mit minimaler knotenzahl. Wiss. Z. Univ. Hall Martin Luther Univ. Halle-Wittenberg MathNatur.Reine, 12:251–257, 1963.
[ES16]	Viveka Erlandsson and Juan Souto. Counting curves in hyperbolic surfaces. Geom. Funct. Anal., 26(3):729–777, 2016.
[ES22]	Viveka Erlandsson and Juan Souto. <i>Mirzakhani's curve counting and geodesic currents</i> , volume 345 of <i>Progress in Mathematics</i> . Birkhäuser/Springer, Cham, [2022] ©2022.
[Fan15]	Federica Fanoni. The maximum injectivity radius of hyperbolic orbifolds. <i>Geom. Dedicata</i> , 175:281–307, 2015.
[FBP22]	Maxime Fortier Bourque and Bram Petri. Kissing numbers of regular graphs. Combinatorica, 42(4):529–551, 2022.
[FBR22]	Maxime Fortier Bourque and Kasra Rafi. Local maxima of the systole func- tion. J. Eur. Math. Soc. (JEMS), 24(2):623–668, 2022.
[Fej43]	L. Fejes. Über die dichteste Kugellagerung. Math. Z., 48:676–684, 1943.
[FMW23]	Mikolaj Fraczyk, Sam Mellick, and Amanda Wilkens. Poisson-voronoi tessel- lations and fixed price in higher rank. Preprint, arXiv:2307.01194, 2023.

- [FP15] Federica Fanoni and Hugo Parlier. Systoles and kissing numbers of finite area hyperbolic surfaces. *Algebr. Geom. Topol.*, 15(6):3409–3433, 2015.
- [FPP+21] Stefan Friedl, JungHwan Park, Bram Petri, Jean Raimbault, and Arunima Ray. On distinct finite covers of 3-manifolds. *Indiana Univ. Math. J.*, 70(2):809–846, 2021.
- [FPS22] David Futer, Jessica S. Purcell, and Saul Schleimer. Effective bilipschitz bounds on drilling and filling. *Geom. Topol.*, 26(3):1077–1188, 2022.
- [Fri08] Joel Friedman. A proof of Alon's second eigenvalue conjecture and related problems. *Mem. Amer. Math. Soc.*, 195(910):viii+100, 2008.
- [FS09] Philippe Flajolet and Robert Sedgewick. Analytic combinatorics. Cambridge University Press, Cambridge, 2009.
- [FSV22] Peter Feller, Alessandro Sisto, and Gabriele Viaggi. Uniform models and short curves for random 3-manifolds. Preprint, arXiv:1910.09486, 2022.
- [Gag80] Michael E. Gage. Upper bounds for the first eigenvalue of the Laplace-Beltrami operator. *Indiana Univ. Math. J.*, 29(6):897–912, 1980.
- [Gam06] Alex Gamburd. Poisson-Dirichlet distribution for random Belyi surfaces. Ann. Probab., 34(5):1827–1848, 2006.
- [GAnS91] Francisco González-Acuña and Hamish Short. Cyclic branched coverings of knots and homology spheres. Rev. Mat. Univ. Complut. Madrid, 4(1):97–120, 1991.
- [Gel63] I. M. Gel'fand. Automorphic functions and the theory of representations. In *Proc. Internat. Congr. Mathematicians (Stockholm, 1962)*, pages 74–85. Inst. Mittag-Leffler, Djursholm, 1963.
- [Gel18] T Gelander. A lecture on invariant random subgroups. In *New Directions in Locally Compact Groups*, pages 186–204. Cambridge Univ. Press, 2018.
- [Gen11] Matthieu Gendulphe. Constante de Bers en genre 2. *Math. Ann.*, 350(4):919– 951, 2011.
- [GHS<sup>+</sup>09] A. Gamburd, S. Hoory, M. Shahshahani, A. Shalev, and B. Virág. On the girth of random Cayley graphs. *Random Structures Algorithms*, 35(1):100– 117, 2009.

- [GLMST21] Clifford Gilmore, Etienne Le Masson, Tuomas Sahlsten, and Joe Thomas. Short geodesic loops and L<sup>p</sup> norms of eigenfunctions on large genus random surfaces. Geom. Funct. Anal., 31(1):62–110, 2021.
- [GLP04] Dorian Goldfeld, Alexander Lubotzky, and László Pyber. Counting congruence subgroups. *Acta Math.*, 193(1):73–104, 2004.
- [GM89] I. Ya. Gol'dsheĭd and G. A. Margulis. Lyapunov exponents of a product of random matrices. *Uspekhi Mat. Nauk*, 44(5(269)):13–60, 1989.
- [GN12] Alexander Gorodnik and Amos Nevo. Counting lattice points. J. Reine Angew. Math., 663:127–176, 2012.
- [Gol84] William M. Goldman. The symplectic nature of fundamental groups of surfaces. Adv. in Math., 54(2):200–225, 1984.
- [Gor00] D. V. Gorbachev. An extremal problem for entire functions of exponential spherical type, which is connected with the Levenshtein bound for the density of a packing of  $\mathbb{R}^n$  by balls. *Izv. Tul. Gos. Univ. Ser. Mat. Mekh. Inform.*, 6(1):71–78, 2000.
- [GPS88] M. Gromov and I. Piatetski-Shapiro. Nonarithmetic groups in Lobachevsky spaces. Inst. Hautes Études Sci. Publ. Math., (66):93–103, 1988.
- [GPY11] Larry Guth, Hugo Parlier, and Robert Young. Pants decompositions of random surfaces. Geom. Funct. Anal., 21(5):1069–1090, 2011.
- [Gru01] Samuel Grushevsky. An explicit upper bound for Weil-Petersson volumes of the moduli spaces of punctured Riemann surfaces. Math. Ann., 321(1):1–13, 2001.
- [Hal05] Thomas C. Hales. A proof of the Kepler conjecture. Ann. of Math. (2), 162(3):1065–1185, 2005.
- [Hay56] W. K. Hayman. A generalisation of Stirling's formula. J. Reine Angew. Math., 196:67–95, 1956.
- [Hid22] Will Hide. Spectral gap for weil-petersson random surfaces with cusps. Preprint, arXiv:2107.14555, 2022.
- [Hid23] Will Hide. Effective lower bounds for spectra of random covers and random unitary bundles. Preprint, arXiv:2305.04584, 2023.

- [HLW06] Shlomo Hoory, Nathan Linial, and Avi Wigderson. Expander graphs and their applications. *Bull. Amer. Math. Soc.* (N.S.), 43(4):439–561, 2006.
- [HM22] Benjamin T. Hansen and Tobias Müller. The critical probability for Voronoi percolation in the hyperbolic plane tends to 1/2. Random Structures Algorithms, 60(1):54–67, 2022.
- [HM23a] Benjamin T. Hansen and Tobias Müller. Poisson-voronoi percolation in the hyperbolic plane with small intensities. Preprint, arXiv:2111.04299, 2023.
- [HM23b] Will Hide and Michael Magee. Near optimal spectral gaps for hyperbolic surfaces. Ann. of Math. (2), 198(2):791–824, 2023.
- [HOOT23] Christian Hirsch, Moritz Otto, Takashi Owada, and Christoph Thäle. Large deviations for hyperbolic k-nearest neighbor balls. Preprint, arXiv:2304.08744, 2023.
- [HS94] Joel Hass and Peter Scott. Shortening curves on surfaces. *Topology*, 33(1):25–43, 1994.
- [HS99] Joel Hass and Peter Scott. Configurations of curves and geodesics on surfaces.
  In Proceedings of the Kirbyfest (Berkeley, CA, 1998), volume 2 of Geom. Topol. Monogr., pages 201–213. Geom. Topol. Publ., Coventry, 1999.
- [HSWX23] Yuxin He, Yang Shen, Yunhui Wu, and Yuhao Xue. Non-simple systoles on random hyperbolic surfaces for large genus. Preprint, arXiv:2308.16447, 2023.
- [HT22] Will Hide and Joe Thomas. Short geodesics and small eigenvalues on random hyperbolic punctured spheres. Preprint, arXiv:2209.15568, 2022.
- [Hub59] Heinz Huber. Zur analytischen Theorie hyperbolischen Raumformen und Bewegungsgruppen. *Math. Ann.*, 138:1–26, 1959.
- [Hub60] Heinz Huber. Zur analytischen Theorie hyperbolischer Raumformen und Bewegungsgruppen. II. Math. Ann., 142:385–398, 1960.
- [Hub61] Heinz Huber. Nachtrag zu: Zur analytischen Theorie hyperbolischer Raumformen und Bewegungsgruppen. II. *Math. Ann.*, 143:463—464, 1961.
- [Hub74] Heinz Huber. Über den ersten Eigenwert des Laplace-Operators auf kompakten Riemannschen Flächen. *Comment. Math. Helv.*, 49:251–259, 1974.

- [HV22] Ursula Hamenstädt and Gabriele Viaggi. Small eigenvalues of random 3manifolds. Trans. Amer. Math. Soc., 375(6):3795–3840, 2022.
- [Imr84] W. Imrich. Explicit construction of regular graphs without small cycles. *Combinatorica*, 4(1):53–59, 1984.
- [Iso00] Yukinao Isokawa. Some mean characteristics of Poisson-Voronoi and Poisson-Delaunay tessellations in hyperbolic planes. Bull. Fac. Ed. Kagoshima Univ. Natur. Sci., 52:11–25, 2000.
- [Iwa02] Henryk Iwaniec. Spectral methods of automorphic forms, volume 53 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI; Revista Matemática Iberoamericana, Madrid, second edition, 2002.
- [Jam21] P. Jammes. Systole and small eigenvalues of hyperbolic surfaces. Preprint, arXiv:2111.08985, 2021.
- [Jen81] F. Jenni. Ueber das spektrum des Laplace-operators auf einer schar konmpakter Riemannscher flächen. PhD thesis, University of Basel, 1981.
- [JL22] Svante Janson and Baptiste Louf. Short cycles in high genus unicellular maps. Ann. Inst. Henri Poincaré Probab. Stat., 58(3):1547–1564, 2022.
- [JL23] Svante Janson and Baptiste Louf. Unicellular maps vs. hyperbolic surfaces in large genus: simple closed curves. Ann. Probab., 51(3):899–929, 2023.
- [JŁR00] Svante Janson, Tomasz Łuczak, and Andrzej Rucinski. Random graphs. Wiley-Interscience Series in Discrete Mathematics and Optimization. Wiley-Interscience, New York, 2000.
- [Joh15] Tobias Johnson. Exchangeable pairs, switchings, and random regular graphs. Electron. J. Combin., 22(1):Paper 1.33, 28, 2015.
- [JS96] Gareth Jones and David Singerman. Belyĭ functions, hypermaps and Galois groups. Bull. London Math. Soc., 28(6):561–590, 1996.
- [JTY<sup>+</sup>21] Zilin Jiang, Jonathan Tidor, Yuan Yao, Shengtong Zhang, and Yufei Zhao. Equiangular lines with a fixed angle. *Ann. of Math.* (2), 194(3):729–743, 2021.

- [Kal17] Olav Kallenberg. Random measures, theory and applications, volume 77 of Probability Theory and Stochastic Modelling. Springer, Cham, 2017.
- [Kam18] Holger Kammeyer. A remark on torsion growth in homology and volume of 3-manifolds. Preprint, arXiv:1802.09244, 2018.
- [Kat07] Mikhail G. Katz. Systolic geometry and topology, volume 137 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2007. With an appendix by Jake P. Solomon.
- [Kee74] Linda Keen. Collars on Riemann surfaces. In Discontinuous groups and Riemann surfaces (Proc. Conf., Univ. Maryland, College Park, Md., 1973), volume No. 79 of Ann. of Math. Stud., pages 263–268. Princeton Univ. Press, Princeton, NJ, 1974.
- [Ker39] Richard Kershner. The number of circles covering a set. Amer. J. Math., 61:665–671, 1939.
- [Kim03] H. H. Kim. Functoriality for the exterior square of  $GL_4$  and the symmetric fourth of  $GL_2$ . J. Amer. Math. Soc., 16(1):139–183, 2003. With appendix 1 by Dinakar Ramakrishnan and appendix 2 by Kim and Peter Sarnak.
- [KL78] G. A. Kabatjanskiĭ and V. I. Levenšteĭn. Bounds for packings on the sphere and in space. *Problemy Peredači Informacii*, 14(1):3–25, 1978.
- [KM68] D. A. Každan and G. A. Margulis. A proof of Selberg's hypothesis. *Mat. Sb.* (N.S.), 75(117):163–168, 1968.
- [KM12] Jeremy Kahn and Vladimir Markovic. Immersing almost geodesic surfaces in a closed hyperbolic three manifold. Ann. of Math. (2), 175(3):1127–1190, 2012.
- [KMP21] P. Kravchuk, D. Mazac, and S. Pal. Automorphic spectra and the conformal bootstrap. Preprint, arXiv:2111.12716, 2021.
- [Kow08] E. Kowalski. The large sieve and its applications, volume 175 of Cambridge Tracts in Mathematics. Cambridge University Press, Cambridge, 2008. Arithmetic geometry, random walks and discrete groups.

#### BIBLIOGRAPHY

- [KSV07] Mikhail G. Katz, Mary Schaps, and Uzi Vishne. Logarithmic growth of systole of arithmetic Riemann surfaces along congruence subgroups. J. Differential Geom., 76(3):399–422, 2007.
- [Le14] Thang Le. Homology torsion growth and Mahler measure. Comment. Math. Helv., 89(3):719–757, 2014.
- [Lê18] Thang T. Q. Lê. Growth of homology torsion in finite coverings and hyperbolic volume. Ann. Inst. Fourier (Grenoble), 68(2):611–645, 2018.
- [Lev79] V. I. Levenšteĭn. On bounds for packings in n-dimensional Euclidean space. Dokl. Akad. Nauk SSSR, 245(6):1299–1303, 1979.
- [Lev99] Silvio Levy, editor. The eightfold way, volume 35 of Mathematical Sciences Research Institute Publications. Cambridge University Press, Cambridge, 1999. The beauty of Klein's quartic curve.
- [LG13] Jean-François Le Gall. Uniqueness and universality of the Brownian map. Ann. Probab., 41(4):2880–2960, 2013.
- [LGM12] Jean-François Le Gall and Grégory Miermont. Scaling limits of random trees and planar maps. In Probability and statistical physics in two and more dimensions, volume 15 of Clay Math. Proc., pages 155–211. Amer. Math. Soc., Providence, RI, 2012.
- [Liu19] Yi Liu. Immersing quasi-Fuchsian surfaces of odd Euler characteristic in closed hyperbolic 3-manifolds. J. Differential Geom., 111(3):457–493, 2019.
- [Liu22] Mingkun Liu. Length statistics of random multicurves on closed hyperbolic surfaces. *Groups Geom. Dyn.*, 16(2):437–459, 2022.
- [Liu23] Yi Liu. Finite quotients, arithmetic invariants, and hyperbolic volume. Peking Math. J., to appear, 2023+.
- [LM00] Valery Liskovets and Alexander Mednykh. Enumeration of subgroups in the fundamental groups of orientable circle bundles over surfaces. Comm. Algebra, 28(4):1717–1738, 2000.
- [LM23] Cyril Letrouit and Simon Machado. Maximal multiplicity of laplacian eigenvalues in negatively curved surfaces. Preprint, arXiv:2307.06646, 2023.

[LMS17]	Etienne Le Masson and Tuomas Sahlsten. Quantum ergodicity and Benjamini-Schramm convergence of hyperbolic surfaces. <i>Duke Math. J.</i> , 166(18):3425–3460, 2017.
[LN04]	Alexander Lubotzky and Nikolay Nikolov. Subgroup growth of lattices in semisimple Lie groups. <i>Acta Math.</i> , 193(1):105–139, 2004.
[Loo97]	Eduard Looijenga. Prym representations of mapping class groups. <i>Geom. Dedicata</i> , 64(1):69–83, 1997.
[LP23]	Mingkun Liu and Bram Petri. Random surfaces with large systoles. In preparation, 2023.
[LPS88]	A. Lubotzky, R. Phillips, and P. Sarnak. Ramanujan graphs. <i>Combinatorica</i> , 8(3):261–277, 1988.
[LR04]	Zeph Landau and Alexander Russell. Random Cayley graphs are expanders: a simple proof of the Alon-Roichman theorem. <i>Electron. J. Combin.</i> , 11(1):Research Paper 62, 6, 2004.
[LS03]	Alexander Lubotzky and Dan Segal. <i>Subgroup growth</i> , volume 212 of <i>Progress in Mathematics</i> . Birkhäuser Verlag, Basel, 2003.
[LS04a]	Martin W. Liebeck and Aner Shalev. Fuchsian groups, coverings of Riemann surfaces, subgroup growth, random quotients and random walks. <i>J. Algebra</i> , 276(2):552–601, 2004.
[LS04b]	Po-Shen Loh and Leonard J. Schulman. Improved expansion of random Cayley graphs. <i>Discrete Math. Theor. Comput. Sci.</i> , 6(2):523–528, 2004.
[LS19]	Martin W. Liebeck and Aner Shalev. Girth, words and diameter. <i>Bull. Lond.</i> <i>Math. Soc.</i> , 51(3):539–546, 2019.
[LS21]	Nati Linial and Michael Simkin. A randomized construction of high girth regular graphs. <i>Random Structures Algorithms</i> , 58(2):345–369, 2021.
[Lub95]	Alexander Lubotzky. Subgroup growth and congruence subgroups. <i>Invent.</i> Math., 119(2):267–295, 1995.
[Lüc94]	W. Lück. Approximating $L^2$ -invariants by their finite-dimensional analogues. Geom. Funct. Anal., 4(4):455–481, 1994.

- [Lüc13] W. Lück. Approximating  $L^2$ -invariants and homology growth. *Geom. Funct.* Anal., 23(2):622–663, 2013.
- [LW23] Michael Lipnowski and Alex Wright. Towards optimal spectral gaps in large genus. Preprint, arXiv:2103.07496, 2023.
- [Mag20] Michael Magee. Letter to Bram Petri. Available at: http://maths.dur.ac. uk/~lcxt26/diameter.pdf, 2020.
- [Mah46] K. Mahler. On lattice points in *n*-dimensional star bodies. I. Existence theorems. *Proc. Roy. Soc. London Ser. A*, 187:151–187, 1946.
- [Mah10] Joseph Maher. Random Heegaard splittings. J. Topol., 3(4):997–1025, 2010.
- [Mak13] Shotaro Makisumi. A note on Riemann surfaces of large systole. J. Ramanujan Math. Soc., 28(3):359–377, 2013.
- [Mal40] A. Malcev. On isomorphic matrix representations of infinite groups. Rec. Math. [Mat. Sbornik] N.S., 8/50:405–422, 1940.
- [Mar82] G.A. Margulis. Explicit constructions of graphs without short cycles and low density codes. *Combinatorica*, 2(1):71–78, 1982.
- [Mar88] G.A. Margulis. Explicit group-theoretic constructions of combinatorial schemes and their applications in the construction of expanders and concentrators. *Problemy Peredachi Informatsii*, 24(1):51–60, 1988.
- [Mar89] Gaven J. Martin. Balls in hyperbolic manifolds. J. London Math. Soc. (2), 40(2):257-264, 1989.
- [Mar22] Bruno Martelli. An Introduction to Geometric Topology. Independently published, 2nd edition, 2022. Available at: http://people.dm.unipi.it/martelli/geometric\_topology.html.
- [Mat13] Carlos Matheus. Some quantitative versions of Ratner's mixing estimates. Bull. Braz. Math. Soc. (N.S.), 44(3):469–488, 2013.
- [McM98] Curtis T. McMullen. Hausdorff dimension and conformal dynamics. III. Computation of dimension. *Amer. J. Math.*, 120(4):691–721, 1998.
- [Mel23] Sam Mellick. Gaboriau's criterion and fixed price one for locally compact groups. Preprint, arXiv:2307.11728, 2023.

[Mie13]	Grégory Miermont. The Brownian map is the scaling limit of uniform random plane quadrangulations. <i>Acta Math.</i> , 210(2):319–401, 2013.
[Mie14]	Grégory Miermont. Aspects of random maps. Available at: http://perso. ens-lyon.fr/gregory.miermont/coursSaint-Flour.pdf, 2014.
[Mir07a]	Maryam Mirzakhani. Simple geodesics and Weil-Petersson volumes of moduli spaces of bordered Riemann surfaces. <i>Invent. Math.</i> , 167(1):179–222, 2007.
[Mir07b]	Maryam Mirzakhani. Weil-Petersson volumes and intersection theory on the moduli space of curves. J. Amer. Math. Soc., 20(1):1–23, 2007.
[Mir08]	Maryam Mirzakhani. Growth of the number of simple closed geodesics on hyperbolic surfaces. Ann. of Math. (2), 168(1):97–125, 2008.
[Mir13]	Maryam Mirzakhani. Growth of Weil-Petersson volumes and random hyper- bolic surfaces of large genus. J. Differential Geom., 94(2):267–300, 2013.
[Mir16]	Maryam Mirzakhani. Counting mapping class group orbits on hyperbolic surfaces. Preprint, arXiv:1601.03342, 2016.
[MN20]	Michael Magee and Frédéric Naud. Explicit spectral gaps for random covers of Riemann surfaces. <i>Publ. Math. Inst. Hautes Études Sci.</i> , 132:137–179, 2020.
[MN21]	Michael Magee and Frédéric Naud. Extension of alon's and friedman's conjectures to schottky surfaces. Preprint, arXiv:2106.02555, 2021.
[MNP22]	Michael Magee, Frédéric Naud, and Doron Puder. A random cover of a compact hyperbolic surface has relative spectral gap $\frac{3}{16} - \varepsilon$ . <i>Geom. Funct. Anal.</i> , 32(3):595–661, 2022.
[Moi52]	Edwin E. Moise. Affine structures in 3-manifolds. V. The triangulation theo- rem and Hauptvermutung. Ann. of Math. (2), 56:96–114, 1952.
[Mon22]	Laura Monk. Benjamini-Schramm convergence and spectra of random hyper- bolic surfaces of high genus. <i>Anal. PDE</i> , 15(3):727–752, 2022.
[Mor94]	M. Morgenstern. Existence and explicit constructions of $q+1$ regular Ramanu- jan graphs for every prime power $q$ . J. Combin. Theory Ser. B, 62(1):44–62, 1994.

- [Mos68] G. D. Mostow. Quasi-conformal mappings in n-space and the rigidity of hyperbolic space forms. Inst. Hautes Études Sci. Publ. Math., (34):53–104, 1968.
- [MP02] Thomas W. Müller and Jan-Christoph Puchta. Character theory of symmetric groups and subgroup growth of surface groups. J. London Math. Soc. (2), 66(3):623–640, 2002.
- [MP19] Maryam Mirzakhani and Bram Petri. Lengths of closed geodesics on random surfaces of large genus. *Comment. Math. Helv.*, 94(4):869–889, 2019.
- [MP23] Michael Magee and Doron Puder. The asymptotic statistics of random covering surfaces. *Forum Math. Pi*, 11:Paper No. e15, 51, 2023.
- [MR03] Colin Maclachlan and Alan W. Reid. *The arithmetic of hyperbolic 3-manifolds*, volume 219 of *Graduate Texts in Mathematics*. Springer-Verlag, New York, 2003.
- [MSP04] Thomas W. Müller and Jan-Christoph Schlage-Puchta. Classification and statistics of finite index subgroups in free products. *Adv. Math.*, 188(1):1–50, 2004.
- [MSP07] Thomas W. Müller and Jan-Christoph Schlage-Puchta. Character theory of symmetric groups, subgroup growth of Fuchsian groups, and random walks. Adv. Math., 213(2):919–982, 2007.
- [MSP10] Thomas W. Müller and Jan-Christoph Schlage-Puchta. Statistics of isomorphism types in free products. *Adv. Math.*, 224(2):707–730, 2010.
- [MSS15] Adam W. Marcus, Daniel A. Spielman, and Nikhil Srivastava. Interlacing families I: Bipartite Ramanujan graphs of all degrees. Ann. of Math. (2), 182(1):307–325, 2015.
- [MT22] Laura Monk and Joe Thomas. The tangle-free hypothesis on random hyperbolic surfaces. *Int. Math. Res. Not. IMRN*, (22):18154–18185, 2022.
- [Mül96] Thomas Müller. Subgroup growth of free products. *Invent. Math.*, 126(1):111–131, 1996.
- [Mül97] Thomas Müller. Finite group actions and asymptotic expansion of  $e^{P(z)}$ . Combinatorica, 17(4):523–554, 1997.

- [Mum71] David Mumford. A remark on Mahler's compactness theorem. *Proc. Amer.* Math. Soc., 28:289–294, 1971.
- [Mur19] Plinio G. P. Murillo. Systole of congruence coverings of arithmetic hyperbolic manifolds. *Groups Geom. Dyn.*, 13(3):1083–1102, 2019. With an appendix by Cayo Dória and Murillo.
- [Mus08] Oleg R. Musin. The kissing number in four dimensions. Ann. of Math. (2), 168(1):1–32, 2008.
- [MW56] Leo Moser and Max Wyman. Asymptotic expansions. *Canad. J. Math.*, 8:225–233, 1956.
- [MW57] Leo Moser and Max Wyman. Asymptotic expansions. II. Canad. J. Math., 9:194–209, 1957.
- [MWW04] B.D. McKay, N.C. Wormald, and B. Wysocka. Short cycles in random regular graphs. *Electron. J. Combin.*, 11(1):Research Paper 66, 12, 2004.
- [MZ15] Maryam Mirzakhani and Peter Zograf. Towards large genus asymptotics of intersection numbers on moduli spaces of curves. *Geom. Funct. Anal.*, 25(4):1258–1289, 2015.
- [Nad87] N. S. Nadirashvili. Multiple eigenvalues of the Laplace operator. *Mat. Sb.* (N.S.), 133(175)(2):223–237, 272, 1987.
- [Nad96] N. Nadirashvili. Berger's isoperimetric problem and minimal immersions of surfaces. Geom. Funct. Anal., 6(5):877–897, 1996.
- [Nau22] Frédéric Naud. Random covers of compact surfaces and smooth linear spectral statistics. Preprint, arXiv:2209.07941, 2022.
- [Nau23] Frédéric Naud. Determinants of laplacians on random hyperbolic surfaces. Preprint, arXiv:2301.09965, 2023.
- [Nic94] Alexandru Nica. On the number of cycles of given length of a free word in several random permutations. *Random Structures Algorithms*, 5(5):703–730, 1994.
- [NWX23] Xin Nie, Yunhui Wu, and Yuhao Xue. Large genus asymptotics for lengths of separating closed geodesics on random surfaces. J. Topol., 16(1):106–175, 2023.

- $\left[ \text{Oes19} \right]$ Joseph Oesterlé. Densité maximale des empilements de sphères en dimensions 8 et 24 [d'après Maryna S. Viazovska et al.]. Number 407, pages Exp. No. 1133, 455–478. 2019. Séminaire Bourbaki. Vol. 2016/2017. Exposés 1120–1135. [OR09] J.-P. Otal and E. Rosas. Pour toute surface hyperbolique de genre g,  $\lambda_{2q-2} >$ 1/4. Duke Math. J., 150(1):101–115, 2009. [OS79] A. M. Odlyzko and N. J. A. Sloane. New bounds on the number of unit spheres that can touch a unit sphere in n dimensions. J. Combin. Theory Ser. A, 26(2):210-214, 1979. [OT23] Moritz Otto and Christoph Thäle. Large nearest neighbour balls in hyperbolic stochastic geometry. *Extremes*, 26(3):413–431, 2023. [Ota08] J.-P. Otal. Three topological properties of small eigenfunctions on hyperbolic surfaces. In Geometry and dynamics of groups and spaces, volume 265 of Progr. Math., pages 685–695. Birkhäuser, Basel, 2008. [Pak99] Igor Pak. Random Cayley graphs with  $O(\log |G|)$  generators are expanders. In Algorithms—ESA '99 (Prague), volume 1643 of Lecture Notes in Comput. Sci., pages 521–526. Springer, Berlin, 1999. [Par13] Hugo Parlier. Kissing numbers for surfaces. J. Topol., 6(3):777–791, 2013. [Par14] Hugo Parlier. A short note on short pants. Canad. Math. Bull., 57(4):870–876, 2014.[Par23] Hugo Parlier. A shorter note on shorter pants. Preprint, arXiv:2304.06973, 2023.[Pat88] S. J. Patterson. On a lattice-point problem in hyperbolic space and related questions in spectral theory. Ark. Mat., 26(1):167–172, 1988. R. C. Penner. Weil-Petersson volumes. J. Differential Geom., 35(3):559–608, [Pen92] 1992. [Per02] Grisha Perelman. The entropy formula for the ricci flow and its geometric applications. Preprint, arXiv:math/0211159, 2002.
- [Per03a] Grisha Perelman. Finite extinction time for the solutions to the ricci flow on certain three-manifolds. Preprint, arXiv:math/0307245, 2003.

- [Per03b] Grisha Perelman. Ricci flow with surgery on three-manifolds. Preprint, arXiv:math/0303109, 2003.
- [Pet17a] Bram Petri. Finite length spectra of random surfaces and their dependence on genus. J. Topol. Anal., 9(4):649–688, 2017.
- [Pet18] Bram Petri. Hyperbolic surfaces with long systoles that form a pants decomposition. *Proc. Amer. Math. Soc.*, 146(3):1069–1081, 2018.
- [PR22] Bram Petri and Jean Raimbault. A model for random three-manifolds. *Comment. Math. Helv.*, 97(4):729–768, 2022.
- [Pra73] Gopal Prasad. Strong rigidity of Q-rank 1 lattices. Invent. Math., 21:255–286, 1973.
- [PT18] Bram Petri and Christoph Thäle. Poisson approximation of the length spectrum of random surfaces. *Indiana Univ. Math. J.*, 67(3):1115–1141, 2018.
- [PV05] Leonid Potyagailo and Ernest Vinberg. On right-angled reflection groups in hyperbolic spaces. *Comment. Math. Helv.*, 80(1):63–73, 2005.
- [PW18] Bram Petri and Alexander Walker. Graphs of large girth and surfaces of large systole. *Math. Res. Lett.*, 25(6):1937–1956, 2018.
- [PZ04] Florian Pfender and Günter M. Ziegler. Kissing numbers, sphere packings, and some unexpected proofs. Notices Amer. Math. Soc., 51(8):873–883, 2004.
- [PZ22] Doron Puder and Tomer Zimhoni. Local statistics of random permutations from free products. Preprint, arXiv:2203.12250, 2022.
- [Rai12] Jean Raimbault. Exponential growth of torsion in abelian coverings. *Algebr. Geom. Topol.*, 12(3):1331–1372, 2012.
- [RB75] S. S. Ryškov and E. P. Baranovskiĭ. Solution of the problem of the least dense lattice covering of five-dimensional space by equal spheres. Dokl. Akad. Nauk SSSR, 222(1):39–42, 1975.

- [RB78] S. S. Ryškov and E. P. Baranovskii. C-types of n-dimensional lattices and 5-dimensional primitive parallelohedra (with application to the theory of coverings). Proc. Steklov Inst. Math., (4):140, 1978. Cover to cover translation of Trudy Mat. Inst. Steklov 137 (1976), Translated by R. M. Erdahl.
- [Rei15] Alan W. Reid. Profinite properties of discrete groups. In Groups St Andrews 2013, volume 422 of London Math. Soc. Lecture Note Ser., pages 73–104.
  Cambridge Univ. Press, Cambridge, 2015.
- [Rei18] Alan W. Reid. Profinite rigidity. In Proceedings of the International Congress of Mathematicians—Rio de Janeiro 2018. Vol. II. Invited lectures, pages 1193–1216. World Sci. Publ., Hackensack, NJ, 2018.
- [RHD07] Roland K. W. Roeder, John H. Hubbard, and William D. Dunbar. Andreev's theorem on hyperbolic polyhedra. Ann. Inst. Fourier (Grenoble), 57(3):825– 882, 2007.
- [Ril90] Robert Riley. Growth of order of homology of cyclic branched covers of knots. Bull. London Math. Soc., 22(3):287–297, 1990.
- [Rog59] C. A. Rogers. Lattice coverings of space. *Mathematika*, 6:33–39, 1959.
- [RS23] Anna Roig Sanchis. The length spectrum of random hyperbolic 3-manifolds. Preprint, 2023.
- [Rud23] Zeév Rudnick. Goe statistics on the moduli space of surfaces of large genus. Preprint, arXiv:2202.06379, 2023.
- [RW23] Zeév Rudnick and Igor Wigman. On the central limit theorem for linear eigenvalue statistics on random surfaces of large genus. Preprint, arXiv:2301.00685, 2023.
- [RY68] G. Ringel and J. W. T. Youngs. Solution of the Heawood map-coloring problem. Proc. Nat. Acad. Sci. U.S.A., 60:438–445, 1968.
- [RZ10] Luis Ribes and Pavel Zalesskii. Profinite groups, volume 40 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer-Verlag, Berlin, second edition, 2010.

- [Sap16] Jenya Sapir. Orbits of non-simple closed curves on a surface. Preprint, arXiv:1602.09093, 2016.
- [Sar19] N.T. Sardari. Diameter of Ramanujan graphs and random Cayley graphs. Combinatorica, 39(2):427–446, 2019.
- [Sch94] P. Schmutz. Systoles on Riemann surfaces. Manuscripta Math., 85(3-4):429– 447, 1994.
- [Sch96a] Paul Schmutz. Arithmetic Fuchsian groups and the number of systoles. *Math.* Z., 223(1):13–25, 1996.
- [Sch96b] Paul Schmutz. Compact Riemann surfaces with many systoles. *Duke Math.* J., 84(1):191–198, 1996.
- [Sco83] P. Scott. The geometries of 3-manifolds. Bull. London Math. Soc., 15(5):401– 487, 1983.
- [Sel56] A. Selberg. Harmonic analysis and discontinuous groups in weakly symmetric Riemannian spaces with applications to Dirichlet series. J. Indian Math. Soc. (N.S.), 20:47–87, 1956.
- [Ser77] Jean-Pierre Serre. Arbres, amalgames, SL<sub>2</sub>. Astérisque, No. 46. Société Mathématique de France, Paris, 1977. Avec un sommaire anglais, Rédigé avec la collaboration de Hyman Bass.
- [Sév02] B. Sévennec. Multiplicity of the second Schrödinger eigenvalue on closed surfaces. *Math. Ann.*, 324(1):195–211, 2002.
- [SS97] Paul Schmutz Schaller. Extremal Riemann surfaces with a large number of systoles. In *Extremal Riemann surfaces (San Francisco, CA, 1995)*, volume 201 of *Contemp. Math.*, pages 9–19. Amer. Math. Soc., Providence, RI, 1997.
- [SS98] Paul Schmutz Schaller. Geometry of Riemann surfaces based on closed geodesics. Bull. Amer. Math. Soc. (N.S.), 35(3):193–214, 1998.
- [ST01] Georg Schumacher and Stefano Trapani. Estimates of Weil-Petersson volumes via effective divisors. *Comm. Math. Phys.*, 222(1):1–7, 2001.
- [Sul16] Diego Sulca. Zeta functions of virtually nilpotent groups. Israel J. Math., 213(1):371–398, 2016.

- [Sun15] Hongbin Sun. Virtual homological torsion of closed hyperbolic 3-manifolds. J. Differential Geom., 100(3):547–583, 2015.
- [SvdW53] K. Schütte and B. L. van der Waerden. Das Problem der dreizehn Kugeln. Math. Ann., 125:325–334, 1953.
- [SW02a] Daniel S. Silver and Susan G. Williams. Mahler measure, links and homology growth. *Topology*, 41(5):979–991, 2002.
- [SW02b] Daniel S. Silver and Susan G. Williams. Torsion numbers of augmented groups with applications to knots and links. *Enseign. Math. (2)*, 48(3-4):317–343, 2002.
- [SW21] Yang Shen and Yunhui Wu. Large degree asymptotics for Cheeger constants of covering hyperbolic surfaces. Preprint, 2021.
- [SW22] Yang Shen and Yunhui Wu. The cheeger constants of random belyi surfaces. International Mathematics Research Notices, 2023(17):15266–15298, nov 2022.
- [SW23] Yang Shen and Yunhui Wu. Arbitrarily small spectral gaps for random hyperbolic surfaces with many cusps. Preprint, arXiv:2203.15681, 2023.
- [Tho21] Joe Thomas. Delocalisation of eigenfunctions on large genus random surfaces. Preprint, arXiv:2002.01403, 2021.
- [Thu10] Axel Thue. Über die dichteste zusammenstellung von kongruenten kreisen in einer ebene. Norske Vid. Selsk. Skr., 1:1–9, 1910.
- [Thu02] W. P. Thurston. The geometry and topology of three-manifolds. Princeton lecture notes, preprint, available at http://library.msri.org/books/gt3m/., 2002.
- [TK92] C.P. Teo and K.M. Koh. The number of shortest cycles and the chromatic uniqueness of a graph. J. Graph Theory, 16(1):7–15, 1992.
- [Tut63] W. T. Tutte. A census of planar maps. Canadian J. Math., 15:249–271, 1963.
- [Ver12] A. M. Vershik. Totally nonfree actions and the infinite symmetric group. Mosc. Math. J., 12(1):193–212, 216, 2012.

[Via17]	M. S. Viazovska. The sphere packing problem in dimension 8. Ann. of Math. (2), 185(3):991–1015, 2017.
[Via21]	Gabriele Viaggi. Volumes of random 3-manifolds. J. Topol., 14(2):504–537, 2021.
[VM22]	Delecroix Vincent and Liu Mingkun. Length partition of random multicurves on large genus hyperbolic surfaces. Preprint, 2202.10255, 2022.
[Vog15]	Karen Vogtmann. $GL(n,\mathbb{Z})$ , $Out(F_n)$ and everything in between: automorphism groups of RAAGs. In <i>Groups St Andrews 2013</i> , volume 422 of <i>London Math. Soc. Lecture Note Ser.</i> , pages 105–127. Cambridge Univ. Press, Cambridge, 2015.
[Vol86]	L. M. Volynets. The number of solutions of the equation $x^s = e$ in a symmetric group. <i>Mat. Zametki</i> , 40(2):155–160, 286, 1986.
[Wal68]	Friedhelm Waldhausen. On irreducible 3-manifolds which are sufficiently large. Ann. of Math. (2), 87:56–88, 1968.
[Wan72]	Hsien Chung Wang. Topics on totally discontinuous groups. In Symmetric spaces (Short Courses, Washington Univ., St. Louis, Mo., 1969–1970), Pure and Appl. Math., Vol. 8, pages 459–487. Dekker, New York, 1972.
[Wei84a]	Boris Weisfeiler. Strong approximation for Zariski-dense subgroups of semisimple algebraic groups. Ann. of Math. (2), 120(2):271–315, 1984.
[Wei84b]	A. Weiss. Girths of bipartite sextet graphs. <i>Combinatorica</i> , 4(2-3):241–245, 1984.
[Whi02]	Matthew E. White. Injectivity radius and fundamental groups of hyperbolic 3-manifolds. <i>Comm. Anal. Geom.</i> , 10(2):377–395, 2002.
[Wil86]	Herbert S. Wilf. The asymptotics of $e^{P(z)}$ and the number of elements of each order in $S_n$ . Bull. Amer. Math. Soc. (N.S.), 15(2):228–232, 1986.
[Wis21]	Daniel T. Wise. The structure of groups with a quasiconvex hierarchy, volume 209 of Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, [2021] ©2021.
[Wol82]	Scott Wolpert. The Fenchel-Nielsen deformation. Ann. of Math. (2), 115(3):501–528, 1982.

- [Wor81a] Nicholas C. Wormald. The asymptotic connectivity of labelled regular graphs. J. Combin. Theory Ser. B, 31(2):156–167, 1981.
- [Wor81b] Nicholas C. Wormald. The asymptotic distribution of short cycles in random regular graphs. J. Combin. Theory Ser. B, 31(2):168–182, 1981.
- [Wor99] N. C. Wormald. Models of random regular graphs. In Surveys in combinatorics, 1999 (Canterbury), volume 267 of London Math. Soc. Lecture Note Ser., pages 239–298. Cambridge Univ. Press, Cambridge, 1999.
- [Wri20] Alex Wright. A tour through Mirzakhani's work on moduli spaces of Riemann surfaces. *Bull. Amer. Math. Soc.* (N.S.), 57(3):359–408, 2020.
- [WX22a] Yunhui Wu and Yuhao Xue. Prime geodesic theorem and closed geodesics for large genus. Preprint, arXiv:2209.10415, 2022.
- [WX22b] Yunhui Wu and Yuhao Xue. Random hyperbolic surfaces of large genus have first eigenvalues greater than  $\frac{3}{16} \epsilon$ . Geom. Funct. Anal., 32(2):340–410, 2022.
- [Yam82] Akira Yamada. On Marden's universal constant of Fuchsian groups. II. J. Analyse Math., 41:234–248, 1982.
- [YY80] P. C. Yang and S. T. Yau. Eigenvalues of the Laplacian of compact Riemann surfaces and minimal submanifolds. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 7(1):55–63, 1980.
- [Zog20] Peter Zograf. On the large genus asymptotics of Weil-Petersson volumes. Preprint, arXiv:0812.0544, 2020.